UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 8-K

CURRENT REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported): July 16, 2021

Evolv Technologies Holdings, Inc.

(Exact name of registrant as specified in its charter)

Delaware	001-39417	84-4473840
(State or other jurisdiction of incorporation)	(Commission File Number)	(IRS Employer Identification No.)
500 Totten Pond Road, 4 th Floor Waltham, Massachusetts		02451
(Address of principal executive offices)		(Zip Code)
Registra	(781) 374-8100 ant's telephone number, including area code	
	NewHold Investment Corp. 12141 Wickchester Ln., Suite 325 Houston, Texas 77079 ne or former address, if changed since last repo	ort.)
Check the appropriate box below if the Form 8-K filing is intended to	simultaneously satisfy the filing obligation of	the registrant under any of the following provisions:
☐ Written communications pursuant to Rule 425 under the Securities	es Act (17 CFR 230.425)	
☐ Soliciting material pursuant to Rule 14a-12 under the Exchange	Act (17 CFR 240.14a-12)	
☐ Pre-commencement communications pursuant to Rule 14d-2(b) to	under the Exchange Act (17 CFR 240.14d-2(b)	
☐ Pre-commencement communications pursuant to Rule 13e-4(c) u	under the Exchange Act (17 CFR 240.13e-4(c))
Securities	registered pursuant to Section 12(b) of the Ac	t:
Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Class A common stock, par value \$0.0001 per share Warrants to purchase one share of Class A common stock Units, each consisting of one share of common stock, \$0.0001 par value, and one-half of one redeemable warrant	EVLV EVLVW EVLVU	The Nasdaq Stock Market The Nasdaq Stock Market The Nasdaq Stock Market
Indicate by check mark whether the registrant is an emerging growth the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).	company as defined in Rule 405 of the Securi	ties Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of
Emerging growth company ⊠		
If an emerging growth company, indicate by check mark if the registr accounting standards provided pursuant to Section 13(a) of the Exchar		tion period for complying with any new or revised financial

INTRODUCTORY NOTE

Unless the context otherwise requires, "we," "us," "our," "Evolv" and the "Company" refer to Evolv Technologies Holdings, Inc., a Delaware corporation, and its consolidated subsidiaries following the Closing (as defined below). Unless the context otherwise requires, references to "NHIC" refer to NewHold Investment Corp., a Delaware corporation, prior to the Closing. All references herein to the "Board" refer to the board of directors of the Company.

Terms used in this Current Report on Form 8-K (this "Report") but not defined herein, or for which definitions are not otherwise incorporated by reference herein, shall have the meaning given to such terms in the Proxy Statement /Prospectus (as defined below) in the section entitled "Frequently Used Terms" beginning on page 1 thereof, and such definitions are incorporated herein by reference.

As disclosed under the sections entitled "Proposal No. 1—The Business Combination Proposal" beginning on page 58 of the proxy statement/prospectus (the "Proxy Statement/Prospectus") filed with the Securities and Exchange Commission (the "SEC") on June 28, 2021 by NHIC, NHIC entered into an Agreement and Plan of Merger, dated March 5, 2021, with NHIC Sub Inc., a wholly-owned subsidiary of NHIC ("Merger Sub"), and Evolv Technologies, Inc. dba Evolv Technology, Inc. ("Legacy Evolv"), as amended by that certain First Amendment to Agreement and Plan of Merger dated June 5, 2021 by and among NHIC, Merger Sub and Legacy Evolv (the "Amendment" and as amended, the "Merger Agreement"). Pursuant to the Merger Agreement, Merger Sub was merged with and into Legacy Evolv, with Legacy Evolv surviving the merger as a wholly owned subsidiary of NHIC (the "Business Combination" and, together with the other transactions contemplated by the Merger Agreement, the "Transactions"). The foregoing description of the Merger Agreement and the Amendment is not complete and is qualified in its entirety by reference to the full text of the Merger Agreement and the Amendment, which are attached as Exhibits 2.1 and 2.2, respectively, to this Report.

On July 15, 2021, NHIC held a special meeting of stockholders (the "Special Meeting"), at which the NHIC stockholders considered and adopted, among other matters, a proposal to approve the Business Combination, including (a) adopting the Merger Agreement and (b) approving the other transactions and matters contemplated by the Merger Agreement and related agreements as described in the Proxy Statement/Prospectus.

Pursuant to the terms and subject to the conditions set forth in the Merger Agreement, following the Special Meeting, on July 16, 2021 (the "Closing Date"), the Transactions were consummated (the "Closing").

Item 2.01 of this Report discusses the consummation of the Transactions and the entry into agreements relating thereto and is incorporated herein by reference.

Item 2.01. Completion of Acquisition or Disposition of Assets.

As described above, on July 15, 2021, NHIC held the Special Meeting, at which the NHIC stockholders considered and adopted, among other matters, a proposal to approve the Merger Agreement and the Transactions. On July 16, 2021, the parties consummated the Business Combination. In connection with the Closing, the Company changed its name from NewHold Investment Corp. to Evolv Technologies Holdings, Inc.

Holders of 8,755,987 shares of NHIC's Class A common stock sold in its initial public offering (the "public shares") properly exercised their right to have such shares redeemed for a full pro rata portion of the trust account holding the proceeds from NHIC's initial public offering, calculated as of two business days prior to the consummation of the Business Combination, which was approximately \$10.00 per share, or \$87,564,196.77 in the aggregate.

As a result of the Business Combination, each share of Legacy Evolv preferred stock and common stock was converted into the right to receive approximately 0.378 shares of the Company's Class A common stock.

2

Additionally, the 4,312,500 shares of NHIC Class B common stock held by NewHold Industrial Technology Holdings LLC (the "Sponsor"), automatically converted to 4,312,500 shares of the Company's Class A common stock.

Pursuant to subscription agreements entered into in connection with the Merger Agreement (collectively, the "Subscription Agreements"), certain investors agreed to subscribe for an aggregate of 30,000,000 newly-issued shares of Class A common stock at a purchase price of \$10.00 per share for an aggregate purchase price of \$300,000,000 (the "PIPE Investment"). At the Closing, the Company consummated the PIPE Investment.

After giving effect to the Transactions, the redemption of public shares as described above, and the consummation of the PIPE Investment there are currently 142,260,102 shares of the Company's Class A common stock issued and outstanding.

The Company's Class A common stock and warrants commenced trading on the Nasdaq Stock Market ("Nasdaq") under the symbols "EVLV" and "EVLVW," respectively, on July 19, 2021, subject to ongoing review of the Company's satisfaction of all listing criteria following the Business Combination.

As noted above, an aggregate of \$87,564,196.77 was paid from the Company's trust account to holders that properly exercised their right to have public shares redeemed, and the remaining balance immediately prior to the Closing of approximately \$85,033,803.23 million remained in the trust account. The remaining amount in the trust account was used to fund the Business Combination.

FORM 10 INFORMATION

Item 2.01(f) of Form 8-K states that if the registrant was a shell company, as NHIC was immediately before the Business Combination, then the registrant must disclose the information that would be required if the registrant were filing a general form for registration of securities on Form 10. Accordingly, the Company is providing below the information that would be included in a Form 10 if it were to file a Form 10. Please note that the information provided below relates to the Combined Company after the consummation of the Business Combination, unless otherwise specifically indicated or the context otherwise requires.

Cautionary Note Regarding Forward-Looking Statements

This Report includes statements that express the Company's opinions, expectations, beliefs, plans, objectives, assumptions or projections regarding future events or future results and therefore are, or may be deemed to be, "forward-looking statements." These forward-looking statements can generally be identified by the use of forward-looking terminology, including the terms "believes," "estimates," "anticipates," "expects," "seeks," "projects," "intends," "plans," "may," "will" or "should" or, in each case, their negative or other variations or comparable terminology. These forward-looking statements include all matters that are not historical facts. They appear in a number of places throughout this Report (including in information that is incorporated by reference into this Report) and include statements regarding our intentions, beliefs or current expectations concerning, among other things, the Transactions and the benefits of the Transactions, including results of operations, financial condition, liquidity, prospects, growth, strategies and the markets in which the Company operates. Such forward-looking statements are based on available current market material and management's expectations, beliefs and forecasts concerning future events impacting the Company. Factors that may impact such forward-looking statements include:

- expectations regarding Evolv's strategies and future financial performance, including its future business plans or objectives, prospective performance and opportunities and competitors, revenues, products and services, pricing, operating expenses, market trends, liquidity, cash flows and uses of cash, capital expenditures, and Evolv's ability to invest in growth initiatives and pursue acquisition opportunities;
- · geopolitical risk and changes in applicable laws or regulations;
- the possibility that Evolv may be adversely affected by other economic, business, and/or competitive factors;
- operational risk;

- risk that the COVID-19 pandemic, and local, state, and federal responses to addressing the pandemic may have an adverse effect on our business operations, as well as our financial condition and results of operations; and
- · litigation and regulatory enforcement risks, including the diversion of management time and attention and the additional costs and demands on Evolv's resources.

The forward-looking statements contained in this Report are based on the Company's current expectations and beliefs concerning future developments and their potential effects on the Transactions and the Company. There can be no assurance that future developments affecting the Company will be those that the Company has anticipated. These forward-looking statements involve a number of risks, uncertainties (some of which are beyond the Company's control) or other assumptions that may cause actual results or performance to be materially different from those expressed or implied by these forward-looking statements. These risks and uncertainties include, but are not limited to, those factors described or incorporated by reference under the heading "Risk Factors" below. Should one or more of these risks or uncertainties materialize, or should any of the assumptions prove incorrect, actual results may vary in material respects from those projected in these forward-looking statements. The Company will not and does not undertake any obligation to update or revise any forward-looking statements, whether as a result of new information, future events or otherwise, except as may be required under applicable securities laws.

Business

The business of the Company is described in the Proxy Statement/Prospectus in the section entitled "Information About Evolv Technology" beginning on page 103 thereof and that information is incorporated herein by reference.

Risk Factors

The risks associated with the Company's business are described in the Proxy Statement/Prospectus in the section entitled "Risk Factors" beginning on page 26 thereof and are incorporated herein by reference. A summary of the risks associated with the Company's business are also described on page 20 of the Proxy Statement/Prospectus under the heading "Summary Risk Factors" and are incorporated herein by reference.

Financial Information

The financial information of the Company is described in the Proxy Statement/Prospectus in the sections entitled "Selected Historical Consolidated Financial and Operating Data of Evolv" and "Management's Discussion and Analysis of Financial Condition and Results of Operations of Evolv Technology" beginning on pages 23 and 122 thereof, respectively, and are incorporated herein by reference.

The financial information of NHIC is described in the Proxy Statement/Prospectus in the sections entitled "Selected Historical Financial Information of NHIC" and "Management's Discussion and Analysis of Financial Condition and Results of Operations of NHIC" beginning on pages 22 and 98 thereof, respectively, and are incorporated herein by reference.

Reference is made to the disclosure set forth in Item 9.01 of this Report relating to the financial information of the Company and NHIC, and to Exhibit 99.2, which is incorporated herein by reference.

Security Ownership of Certain Beneficial Owners and Management

The following table sets forth information known to us regarding the beneficial ownership of our Common Stock immediately following consummation of the Transactions by:

- each person who is the beneficial owner of more than 5% of the outstanding shares of our Common Stock;
- · each of our named executive officers and directors; and
- · all of our executive officers and directors as a group.

4

Beneficial ownership is determined according to the rules of the SEC, which generally provide that a person has beneficial ownership of a security if he, she or it possesses sole or shared voting or investment power over that security, including options and warrants that are currently exercisable or exercisable within 60 days. Except as described in the footnotes below and subject to applicable community property laws and similar laws, we believe that each person listed below has sole voting and investment power with respect to such shares. Unless otherwise noted, the address of each beneficial owner is c/o Evolv Technologies Holdings, Inc., 500 Totten Pond Road, 4th Floor, Waltham, MA 02451.

The beneficial ownership of our Class A common stock is based on 142,260,102 shares of Class A common stock issued and outstanding immediately following consummation of the Transactions, including the redemption of public shares as described above and the consummation of the PIPE Investment.

Beneficial Ownership Table

Name of Beneficial Owner ⁽¹⁾	Number of Shares of Class A Common Stock Beneficially Owned	Percentage of Outstanding Class A Common Stock
5% Stockholders:		
Gates Frontier, LLC ⁽²⁾	16,839,983	11.7%
Data Collective IV, L.P. ⁽³⁾	11,871,242	8.3%
General Catalyst Group V, L.P. (4)	14,896,657	11.7%
Finback Evolv, LLC ⁽⁵⁾	17,243,342	11.8%
Lux Ventures III, L.P. ⁽⁶⁾	14,442,033	10.4%
SineWave Ventures, LLC ⁽⁷⁾	7,419,132	5.8%
Directors and Executive Officers:		
Peter George ⁽⁸⁾	9,349,534	6.2%
Peter Faubert ⁽⁹⁾	1,808,285	1.3%
Anil Chitkara ⁽¹⁰⁾	5,661,317	3.9%

Michael Ellenbogen ⁽¹¹⁾	15,819,627	10.4%
Anthony DeRosa ⁽¹²⁾	1,249,595	*
Alan Cohen ⁽¹³⁾	1,173,687	*
Kevin Charlton	_	_
Neil Glat	_	_
David Orfao	_	_
Mahesh Saptharishi	_	_
Merline Saintil ⁽¹⁴⁾	313,452	*
Kimberly Sheehy	_	_
Mark Sullivan ⁽¹⁵⁾	357,646	*
Bilal Zuberi	_	_
Directors and executive officers as a group (14 individuals) ⁽¹⁶⁾	33,297,615	19.8%

^{*} Less than 1%.

(2) Consists of (i) 15,050,161 shares of the Company's Class A common stock and (ii) 1,789,822 earn-out shares pursuant to the Merger Agreement (the "Earn-Out Shares") issuable within 60 days of Closing. All shares of the Company's Class A common stock held by Gates Frontier, LLC may be deemed to be beneficially owned by William H. Gates III as the sole member of Gates Frontier, LLC. The address of Gates Frontier, LLC is 2365 Carillon Point, Kirkland, WA 98033.

5

- (3) Consists of (i) 10,601,281 shares of the Company's Class A common stock and (ii) 1,269,961 Earn-Out Shares issuable within 60 days of Closing. The registered holder of the referenced shares is Data Collective IV, L.P. ("DCVC IV"). Data Collective IV GP, LLC ("DCVC IV GP") is the general partner of DCVC IV. Zachary Bogue and Matthew Ocko are the managing members of DCVC IV GP and share voting and dispositive power over the shares held by DCVC IV. Zachary Bogue and Matthew Ocko disclaim beneficial ownership of the shares held by DCVC IV, except to the extent of their indirect pecuniary interest therein, if any. The address of the entities listed herein is 270 University Avenue, Palo Alto, California 94301.
- (4) Consists of (i) 13,296,147 shares of the Company's Class A common stock and (ii) 1,600,510 Earn-Out Shares issuable within 60 days of Closing. Represents shares of the Company's Class A common stock owned by General Catalyst Group V, L.P. ("GCG V"). General Catalyst GP V, LLC ("GCGP V") is the general partner of General Catalyst Partners V, L.P. Lawrence Bohn, Joel Cutler, and David Fialkow are managing directors of GCGP V, and, as a result, may be deemed to have voting and dispositive power over the shares held by GCG V. The address for General Catalyst is 20 University Road, Suite 450, Cambridge, MA 02138.
- (5) Consists of (i) 15,431,758 shares of the Company's Class A common stock, (ii) 1,811,584 Earn-Out Shares issuable within 60 days of Closing and (iii) 2,421,199 warrants to purchase shares of the Company's Class A common stock within 60 days of Closing. The registered holders of the referenced shares will be funds and accounts under management by Finback Investment Partners, LLC and related parties. The applicable portfolio managers, as managing directors of such entity, will have voting and investment power over the shares held by the funds and accounts, which will be the registered holders of the referenced shares. Such portfolio managers expressly disclaim beneficial ownership of all shares to be held by such funds and accounts. The address of such funds and accounts and such portfolio managers is 1200 Anastasia Avenue, Suite 500, Coral Gables, FL 33134.
- (6) Consists of shares of the Company's Class A common stock held by Lux Ventures III, L.P. Lux Venture Partners III, LLC is the general partner of Lux Ventures III L.P. and exercises voting and dispositive power over the shares noted herein held by Lux Ventures III, L.P. Peter Hebert, and Josh Wolfe are the individual managing members of Lux Venture Partners III, LLC. The individual managers, as the sole managers of Lux Venture Partners III, LLC, may be deemed to share voting and dispositive power for the shares noted herein held by Lux Ventures III, L.P. Each of Lux Venture Partners III, LLC, and the individual managers separately disclaim beneficial ownership over the shares noted herein except to the extent of their pecuniary interest therein. The address for these entities and individuals is c/o Lux Capital Management, 920 Broadway, 11th Floor, New York, NY 10010.
- (7)Consists of (i) 7,419,132 shares of the Company's Class A common stock and (ii) 889,791 Earn-Out Shares issuable within 60 days of Closing. The registered holders of referenced shares are funds and accounts under the management of SineWave Ventures, LLC. The address of SineWave Ventures, LLC is: 1390 Chain Bridge Road, #A177, McLean, VA 22101.
- (8)Consists of (i) 633,427 shares of the Company's Class A common stock, (ii) 8,062,260 shares of the Company's Class A common stock subject to options held by Mr. George that are exercisable within 60 days of Closing and (iii) 653,847 Earn-Out Shares issuable within 60 days of Closing.
- (9)Consists of (i) 1,663,974 shares of the Company's Class A common stock subject to options held by Mr. Faubert that are exercisable within 60 days of Closing and (ii) 144,311 Earn-Out Shares issuable within 60 days of Closing.
- (10)Consists of (i) 1,096,068 shares of the Company's Class A common stock, (ii) 4,211,718 shares of the Company's Class A common stock subject to options held by Mr. Chitkara that are exercisable within 60 days of Closing and (iii) 353,513 Earn-Out Shares issuable within 60 days of Closing.
- (11)Consists of (i) 5,268,110 shares of the Company's Class A common stock, (ii) 9,646,703 shares of the Company's Class A common stock subject to options held by Mr. Ellenbogen and his affiliates that are exercisable within 60 days of Closing and (iii) 904,814 Earn-Out Shares issuable to Mr. Ellenbogen and his affiliates within 60 days of Closing.

- (12)Consists of (i) 1,119,895 shares of the Company's Class A common stock subject to options held by Mr. DeRosa that are exercisable within 60 days of Closing and (iii) 129,700 Earn-Out Shares issuable within 60 days of Closing.
- (13)Consists of (i) 593,908 shares of the Company's Class A common stock, (ii) 579,779 shares of the Company's Class A common stock subject to options held by affiliates of Mr. Cohen that are exercisable within 60 days of Closing and (iii) 138,972 Earn-Out Shares issuable to affiliates of Mr. Cohen within 60 days of Closing.
- (14)Consists of (i) 280,918 shares of the Company's Class A common stock subject to options held by Ms. Saintil that are exercisable within 60 days of Closing and (ii)

⁽¹⁾ Unless otherwise noted, the business address of each of the following entities or individuals is 500 Totten Pond Road, 4th Floor, Waltham, MA 02451.

32,534 Earn-Out Shares issuable within 60 days of Closing.

(15)Consists of (i) 320,832 shares of the Company's Class A common stock subject to options held by Mr. Sullivan that are exercisable within 60 days of Closing and (ii) 36,814 Earn-Out Shares issuable within 60 days of Closing.

(16)Consists of (i) 6,997,623 shares of the Company's Class A common stock held by all directors and executive officers of the Company as a group, (ii) 23,905,487 shares of the Company's common stock subject to options held by all directors and executive officers of the Company as a group and that are exercisable within 60 days of Closing and (iii) 2,394,505 Earn-Out Shares issuable within 60 days of Closing.

Directors and Executive Officers

The Company's directors and executive officers upon the Closing are described in the Proxy Statement/Prospectus in the section entitled "Directors and Officers of the Combined Company After the Business Combination" beginning on page 171 thereof and that information is incorporated herein by reference.

Directors

Pursuant to the approval of NHIC stockholders from the Special Meeting, the following persons will constitute the Company's Board effective upon the Closing: Peter George, Michael Ellenbogen, Alan Cohen, Kevin Charlton, Neil Glat, David Orfao, Merline Saintil, Mahesh Saptharishi, Kimberly Sheehy, Mark Sullivan and Bilal Zuberi. Kevin Charlton, David Orfao and Bilal Zuberi were appointed to serve as ClassI directors, with terms expiring at the Company's first annual meeting of stockholders following the Closing; Neil Glat, Mark Sullivan, Alan Cohen and Merline Saintil were appointed to serve as ClassII directors, with terms expiring at the Company's second annual meeting of stockholders following the Closing; and Peter George, Michael Ellenbogen, Mahesh Saptharishi and Kimberly Sheehy were appointed to serve as ClassIII directors, with terms expiring at the Company's third annual meeting of stockholders following the Closing. Biographical information for these individuals is set forth in the Proxy Statement/Prospectus in the section titled "Directors and Executive Officers of the Combined Company After the Business Combination" beginning on page171, which is incorporated herein by reference.

Committees of the Board of Directors

Effective as of the Closing, the standing committees of the Company's Board consist of an audit committee (the "Audit Committee"), a compensation committee (the "Compensation Committee"), a nominating and corporate governance committee (the "Nominating and Corporate Governance Committee") and a technology sub-committee (the "Technology Sub-Committee"). Each of the committees reports to the Board.

Effective as of the Closing, the Board appointed Mr. Glat and Mmes.Saintil and Sheehy to serve on the Audit Committee, with Ms. Sheehy as chair. The Board appointed Messrs. Charlton, Orfao and Sullivan to serve on the Compensation Committee, with Mr. Charlton as chair. The Board appointed Mr. Cohen and Ms. Saintil to serve on the Nominating and Corporate Governance Committee, with Ms. Saintil as chair. The Board appointed Messrs. Cohen, Glat, Saptharishi and Zuberi to serve on the Technology Sub-Committee, with Mr. Saptharishi as chair.

7

Executive Officers

Effective as of the Closing, each of Messrs. Charlton, Sullivan, Goldman, Baynes-Reid and Deutsch resigned as the Chief Executive Officer, Chairman of the Board of Directors, Vice Chairman of the Board of Directors, Chief Operating Officer and Chief Financial Officer, respectively. Effective as of the Closing, the Board appointed Peter George to serve as Chief Executive Officer, Peter Faubert to serve as Chief Financial Officer, Anthony John DeRosa to serve as Chief Revenue Officer, Anil Chitkara to serve as Head of Corporate Development and Michael Ellenbogen to serve as Head of Advanced Technology.

Executive Compensation

Executive Compensation

The executive compensation of the Company's named executive officers and directors is described in the Proxy/Prospectus in the section entitled "Directors and Executive Officers of the Combined Company After the Business Combination—Officer and Director Compensation Following the Merger" beginning on page 176 thereof and that information is incorporated herein by reference.

Compensation Committee Interlocks and Insider Participation

None of our executive officers serves as a member of the board of directors or compensation committee (or other committee performing equivalent functions) of any entity that has one or more executive officers serving on our Board or Compensation Committee.

Certain Relationships and Related Transactions, and Director Independence

Certain Relationships and Related Transactions

Certain relationships and related transactions are described in the Proxy Statement/Prospectus in the section entitled "Certain Relationships and Related Transactions" beginning on page 193 thereof and are incorporated herein by reference.

Director Independence

Nasdaq listing standards require that a majority of our Board be independent. An "independent director" is defined generally as a person other than an officer or employee of the company or its subsidiaries or any other individual having a relationship which in the opinion of the company's board of directors, would interfere with the director's exercise of independent judgment in carrying out the responsibilities of a director. Our Board has determined that Alan Cohen, Kevin Charlton, Neil Glat, David Orfao, Merline Saintil, Kimberly Sheehy, Mark Sullivan, Bilal Zuberi and Mahesh Saptharishi are "independent directors" as defined in the Nasdaq listing standards and applicable SEC rules. Our independent directors will have regularly scheduled meetings at which only independent directors are present.

Risk Oversight

Our Board is responsible for overseeing our risk management process. Our Board focuses on our general risk management strategy, the most significant risks facing us, and oversees the implementation of risk mitigation strategies by management. Our Audit Committee is also responsible for discussing our policies with respect to risk assessment and risk management. Our Board believes its administration of its risk oversight function has not negatively affected our Board's leadership structure.

Committees of the Board of Directors

The committees of our Board are described in the Proxy Statement/Prospectus in the section entitled "Directors and Executive Officers of the Combined Company After the Business Combination—Board of Directors—Committees of the Board of Directors" beginning on page 175 thereof and that information is incorporated herein by reference.

Legal Proceedings

Reference is made to the disclosure regarding legal proceedings in the section of the Proxy Statement/Prospectus titled "Information About Evolv Technology—Legal Proceedings" beginning on page 121, which is incorporated herein by reference.

Market Price of and Dividends on the Registrant's Common Equity and Related Stockholder Matters

Price Range of Securities and Dividends

Shares of Evolv's common stock, Evolv's warrants and Evolv's units began trading on the Nasdaq under the symbols "EVLV," "EVLVW" and "EVLVU" respectively, on July 19, 2021, in lieu of the shares of common stock, warrants and units of NHIC. Evolv has not paid any cash dividends on its shares of common stock to date. It is the present intention of Evolv's board of directors to retain all earnings, if any, for use in Evolv's business operations and, accordingly, Evolv's Board does not anticipate declaring any dividends in the foreseeable future. The payment of cash dividends in the future will be dependent upon Evolv's revenues and earnings, if any, capital requirements and general financial condition. The payment of any cash dividends is within the discretion of Evolv's Board. Further, the ability of Evolv to declare dividends may be limited by the terms of financing or other agreements entered into by it or its subsidiaries from time to time.

Information respecting NHIC's common stock, warrants and units and related stockholder matters are described in the Proxy Statement/Prospectus in the section titled "Trading Market and Dividends" on page 25 and such information is incorporated herein by reference.

The Company's Class A common stock, warrants and units commenced trading on Nasdaq under the symbols "EVLV," "EVLVW" and "EVLVU" respectively, on July 19, 2021, subject to ongoing review of the Company's satisfaction of all listing criteria following the Business Combination, in lieu of the Class A common stock, warrants and units of NHIC. NHIC's units ceased trading separately on Nasdaq on July 19, 2021.

Holders of Record

As of the Closing and following the completion of the Transactions, including the redemption of public shares as described above and the consummation of the PIPE Investment, the Company had 142,260,102 shares of Class A common stock outstanding held of record by 88 holders and no shares of preferred stock outstanding. Such amounts do not include DTC participants or beneficial owners holding shares through nominee names.

Securities Authorized for Issuance Under Equity Compensation Plans

Reference is made to the disclosure described in the Proxy Statement/Prospectus in the sections entitled "Proposal No. 3—The Stock Plan Proposal" and "Proposal No. 4—The ESPP Proposal" beginning on pages 82 and 86 thereof, respectively, which are incorporated herein by reference. As described below, the Evolv Technologies Holdings, Inc. 2021 Incentive Award Plan and the Evolv Technologies Holdings, Inc. 2021 Employee Stock Purchase Plan and the material terms thereunder, including the authorization of the initial share reserves thereunder, were approved by NHIC's stockholders at the Special Meeting.

Recent Sales of Unregistered Securities

Reference is made to the disclosure set forth under Item 3.02 of this Report relating to the issuance of the Company's Class A common stock in connection with the Transactions, which is incorporated herein by reference.

9

Description of Registrant's Securities to be Registered

The Company's common stock is described in the Proxy Statement/Prospectus in the section entitled "Comparison of Stockholders' Rights" beginning on page 180 thereof and that information is incorporated herein by reference. The Company's warrants and units are described in NHIC's registration statement on Form S-1 (File No. 333-239822) initially filed with the SEC on July 10, 2021, as amended, in the section titled "Description of Securities" beginning on page 116 thereof. As described below, the Company's A&R Charter (as defined below) was approved by NHIC's stockholders at the Special Meeting and became effective as of the Closing.

Indemnification of Directors and Officers

Evolv has entered into indemnification agreements with each of its directors and executive officers. Under the terms of such indemnification agreements, we are required to indemnify each of our directors and executive officers, to the fullest extent permitted by the laws of the state of Delaware, if the basis of the indemnitee's involvement was by reason of the fact that the indemnitee is or was our director or officer or was serving at our request in an official capacity for another entity. Evolv must indemnify its directors and executive officers against all direct and indirect costs, fees and expenses of any type or nature whatsoever, including all other disbursements, obligations or expenses of the types customarily incurred in connection with prosecuting, defending, preparing to prosecute or defend, investigating, being or preparing to be witness in, settlement or appeal of, or otherwise participating in any threatened, pending or completed action, suit, claim, counterclaim, cross claim, arbitration, mediation, alternate dispute resolution mechanism, investigation, inquiry, administrative hearing or any other actual, threatened or completed proceeding. The indemnification agreements also require Evolv to advance, to the extent not prohibited by law, all direct and indirect costs, fees and expenses that such director or executive officer incurred, provided that such person will return any such advance if it is ultimately determined that such person is not entitled to indemnification by Evolv. The foregoing description of the indemnification agreements does not purport to be complete and is qualified in its entirety by the terms and conditions of the indemnification agreements, a form of which is attached hereto as Exhibit 10.10 and is incorporated herein by reference.

Further information about the indemnification of Evolv's directors and officers is set forth in the Proxy Statement/Prospectus in the section titled "Comparison of Stockholders' Rights—Indemnification of Directors, Officers, Employees and Agents" beginning on page 189 thereof and that information is incorporated herein by reference.

Financial Statements and Supplementary Data

Reference is made to the disclosure set forth under Item 9.01 of this report relating to the financial information of the Company, and to Exhibit 99.2, which is incorporated herein by reference.

Changes in and Disagreements with Accountants on Accounting and Financial Disclosure

Reference is made to the disclosure set forth under Item 4.01 of this report relating to the change in NHIC's certifying accountant, which is incorporated herein by reference.

Item 3.02. Unregistered Sales of Equity Securities.

At the Closing, the Company consummated the PIPE Investment. Additionally, the 4,312,500 shares of NHIC's Class B common stock held by the Sponsor automatically converted to shares of Class A common stock as of the Closing. The disclosure under Item 2.01 of this Report is incorporated into this Item 3.02 by reference.

The Company issued the foregoing securities under Section 4(a)(2) of the Securities Act and/or Rule 506 of Regulation D promulgated under the Securities Act, as a transaction not requiring registration under Section 5 of the Securities Act. The parties receiving the securities represented their intentions to acquire the securities for investment only and not with a view to or for sale in connection with any distribution, and appropriate restrictive legends were affixed to the certificates representing the securities (or reflected in restricted book entry with the Company's transfer agent). The parties also had adequate access, through business or other relationships, to information about the Company.

In addition, at the Closing, each issued and outstanding convertible promissory note of Evolv (collectively, the "2021 Evolv Convertible Notes") automatically converted into shares of Legacy Evolv common stock and, in connection therewith, holders of the 2021 Evolv Convertible Notes received an aggregate of 1,000,000 additional shares of NHIC Class A common stock as further consideration for the conversion of the 2021 Evolv Convertible Notes consistent with the terms thereof.

10

Item 3.03. Material Modification to Rights of Security Holders

The information set forth in Item 5.03 to this Current Report on Form 8-K is incorporated herein by reference.

Item 4.01 Changes in Registrant's Certifying Accountant

On July 16, 2021, WithumSmith+Brown PC ("Withum"), NHIC's independent registered public accounting firm prior to the Business Combination, was dismissed as NHIC's independent registered public accounting firm, which dismissal will become effective following the completion of NHIC's review of the quarter ended June 30, 2021, which consists only of the accounts of the pre-Business Combination special purpose acquisition company, NHIC.

Withum's report on NHIC's financial statements as of December 31, 2020, the related statements of operations, changes in stockholders' equity and cash flows for the period from January 24, 2020 (inception) through December 31, 2020 did not contain any adverse opinion or disclaimer of opinion, nor were they qualified or modified as to uncertainty, audit scope or accounting principles.

During the period from January 24, 2020 (inception) through December 31, 2020 and the subsequent period through July 16, 2021, there were no: (i) disagreements with Withum on any matter of accounting principles or practices, financial statement disclosures or audited scope or procedures, which disagreements if not resolved to Withum's satisfaction would have caused Withum to make reference to the subject matter of the disagreement in connection with its report or (ii) reportable events as defined in Item 304(a)(1)(v) of Regulation S-K other than as described immediately below.

On April 12, 2021, the SEC issued a Staff Statement on Accounting and Reporting Considerations for Warrants Issued by Special Purpose Acquisition Companies ("SPAC's"), the "Statement." In the Statement, the SEC indicates its view that certain terms of the warrants issued in connection with a SPAC Initial Public Offering ("Public Warrants") and private placement warrants typically issued by a SPAC ("Private Warrants") call for such warrants to be accounted for as liabilities and not as equity. The Company has recorded such warrants as equity. This determination (liability vs. equity) has caused the Company to restate previously issued financial statements that showed the warrants as equity. Therefore, management concluded that the Company did not maintain effective internal control over financial reporting as of March 31, 2021, December 31, 2020 and September 30, 2020, due to a material weakness in the Company's internal control over financial reporting related to a lack of an effectively designed control over the evaluation of settlement features used to determine the classification of warrant instruments. The need to restate financial statements in this instance constitutes a material weakness in internal control.

In light of this material weakness at March 31, 2021, December 31, 2020 and September 30, 2020, the Company modified its procedures to include employing consultants on valuation and specialized knowledge on warrant accounting and the Company performed additional analysis as deemed necessary to ensure that its financial statements at March 31, 2021 were prepared in accordance with U.S. generally accepted accounting principles. Accordingly, management believes that the financial statements included in the Company's Quarterly Report on Form 10-Q for the quarterly period ended March 31, 2021 presents fairly in all material respects the Company's financial position, results of operations and cash flows for the period presented.

The Company has provided Withum with a copy of the disclosures made by the Company in response to this Item 4.01 and has requested that Withum furnish the Company with a letter addressed to the SEC stating whether it agrees with the statements made by the registrant in response to Item 304(a) and, if not, stating the respects in which it does not agree. A letter from Withum is attached as Exhibit 16.1 to this Report.

Item 5.01. Changes in Control of the Registrant.

The information set forth above under Item 1.01 and Item 2.01 of this Report is incorporated herein by reference.

Item 5.02. Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers.

The information set forth above in the sections titled "Directors and Officers," "Executive Compensation," "Certain Relationships and Related Transactions" and "Indemnification of Directors and Officers" in Item 2.01 to this Report is incorporated herein by reference.

11

Item 5.03. Amendments to Articles of Incorporation or Bylaws; Change in Fiscal Year.

On July 16, 2021, in connection with the consummation of the Transactions, the Company amended and restated its bylaws (as amended, the "A&R Bylaws") prior to the closing and amended and restated its certificate of incorporation effective as of the Closing (as amended, the "A&R Charter").

Copies of the A&R Charter and the A&R Bylaws are attached as Exhibit 3.1 and Exhibit 3.2 to this Report, respectively, and are incorporated herein by reference.

The material terms of each of the A&R Charter and the A&R Bylaws and the general effect upon the rights of holders of the Company's capital stock are included in the Proxy Statement/Prospectus under the sections titled "Proposal No. 2—The Charter Approval Proposal" and "Comparison of Stockholders' Rights" beginning on pages 79 and 180 of the Proxy Statement/Prospectus, respectively, which are incorporated herein by reference.

Item 5.06 Change in Shell Company Status

As a result of the Business Combination, the Company ceased to be a shell company. Reference is made to the disclosure in the Proxy Statement/Prospectus in the sections entitled "Proposal No. 1—The Business Combination Proposal" beginning on page 58 thereof, which is incorporated herein by reference.

Item 8.01. Other Events.

On July 19, 2021, the parties issued a joint press release announcing the completion of the Business Combination, a copy of which is furnished as Exhibit 99.1 hereto.

Item 9.01. Financial Statement and Exhibits.

(a) Financial statements of businesses acquired.

The audited consolidated financial statements of Legacy Evolv for the years ended December 31, 2020 and 2019 are included in the Proxy Statement/Prospectus beginning on page F-40 and are incorporated herein by reference.

The unaudited condensed consolidated financial statements of Legacy Evolv for the three months ended March 31, 2021 and 2020 are included in the Proxy Statement/Prospectus beginning on page F-22 and are incorporated herein by reference.

(b) Pro forma financial information.

The unaudited pro forma condensed combined financial information of the Company for the year ended December 31, 2020 and the three months ended March 31, 2021 is filed as Exhibit 99.2 and is incorporated herein by reference.

(d) Exhibits.

			<u>Incorporated</u>	by Reference
Exhibit				
Number	<u>Description</u>	<u>Form</u>	<u>Exhibit</u>	Filing Date
2.1*	Agreement and Plan of Merger, dated as of March 5, 2021, by and among the NHIC, Merger	<u>8-K</u>	<u>2.1</u>	<u>3/8/21</u>
	Sub and Legacy Evolv.			
<u>2.2</u>	First Amendment to Agreement and Plan of Merger dated June 5, 2021, by and among the			
	Company, Merger Sub and Legacy Evolv.			
3.1	Amended and Restated Certificate of Incorporation.			
<u>3.2</u>	Amended and Restated By-Laws.			
<u>4.1</u>	Specimen Class A Common Stock Certificate.	<u>S-4/A</u>	<u>4.4</u>	6/9/21
<u>4.2</u>	Specimen Warrant Certificate.	<u>S-1/A</u>	<u>4.3</u>	<u>7/27/20</u>

<u>4.3</u>	Warrant Agreement, dated July 30, 2020, by and between NHIC and Continental Stock Transfer & Trust Company, as warrant agent.	<u>S-</u>	<u>4.4</u>	7/27/20
<u>10.1</u>	Letter Agreement, dated July 30, 2020, by and among NHIC, its officers, its directors and NewHold Industrial Technology Holdings LLC.	1/A S- 1/A S-	<u>10.1</u>	<u>7/27/20</u>
<u>10.2</u>	Investment Management Trust Agreement, dated July 30, 2020, by and between Continental Stock Transfer & Trust Company and NewHold.	1/A S-	10.3	7/27/20
<u>10.3</u>	Registration Rights Agreement, dated July 30, 2020, by and among NHIC and certain stockholders.	1/A S-	<u>10.4</u>	<u>7/27/20</u>
<u>10.4</u>	Promissory Note, dated July 30, 2020, issued to NewHold Industrial Technology Holdings LLC.	1/A S-	10.2	7/27/20
10.5	Administrative Support Agreement, dated July 30, 2020, by and between the Registrant and NewHold Enterprises LLC.	1/A S-	10.8	7/27/20
<u>10.6</u>	Subscription Agreement, dated July 30, 2020, by and among the Registrant, NewHold Industrial Technology Holdings LLC and each of the	1/A S-	10.8	7/27/20
<u>10.7</u>	Anchor Investors. Form of Indemnity Agreement of NHIC.	1/A S-	<u>10.7</u>	7/27/20
10.8	Sponsor Support Agreement, dated March 5, 2021, by and among NHIC and the Sponsor.	1/A 8-K	10.1	3/8/21
<u>10.9</u>	Letter Agreement, dated March 5, 2021, by and among NewHold Industrial Technology Holdings LLC, Legacy Evolv, NHIC and certain other parties thereto.	<u>8-K</u>	<u>10.2</u>	<u>3/8/21</u>
<u>10.10</u>	Form of Indemnification Agreement.	<u>S-</u> <u>4/A</u>	10.10	6/9/21
<u>10.11</u>	Amended and Restated Executive Employment Agreement between Legacy Evolv and Peter George.	<u>S-</u> 4/A	10.11	6/9/21
<u>10.12</u>	Executive Employment Agreement between Legacy Evolv and Anil R. Chitkara.	<u>S-</u> 4/A	<u>10.12</u>	6/9/21
<u>10.13</u>	Executive Employment Agreement between Legacy Evolv and Peter Faubert.	<u>S-</u>	<u>10.13</u>	6/9/21
<u>10.14</u>	Form of 2021 Evolv Technologies Holdings, Inc. Incentive Award Plan.	4/A S-	<u>10.14</u>	6/9/21
<u>10.15</u>	Form of 2021 Evolv Technologies Holdings, Inc. Employee Stock Purchase Plan.	4/A S-	10.15	6/9/21
10.16	Form of Amended and Decisted Decistration Digital Agreement by and among NIHIC Evoly and contain steakholders	<u>4/A</u>		

- 10.16 Form of Amended and Restated Registration Rights Agreement by and among NHIC, Evolv and certain stockholders.
- 16.1 Letter from WithumSmith+Brown PC to the U.S. Securities and Exchange Commission dated July 21, 2021.
- 21.1 Subsidiaries of the Company.
- 99.1 Press release dated July 19, 2021.
- 99.2 Unaudited pro forma condensed combined financial information of the Company for the year ended December 31, 2020 and the three months ended March 31, 2021.

* Certain of the exhibits and schedules to this Exhibit have been omitted in accordance with Regulation S-K Item 601(a)(5). The Company agrees to furnish a copy of all omitted exhibits and schedules to the SEC upon its request.
13

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Evolv Technologies Holdings, Inc.

Date: July 22, 2021

By: /s/ Peter George
Name: Peter George

Title: Chief Executive Officer

FIRST AMENDMENT TO AGREEMENT AND PLAN OF MERGER

THIS FIRST AMENDMENT TO AGREEMENT AND PLAN OF MERGER (this "Amendment") is made as of June 5, 2021 (the "Amendment Date") by and among NewHold Investment Corp., a Delaware corporation ("Parent"), NHIC Sub Inc., a Delaware limited liability company and wholly-owned subsidiary of Parent ("Merger Sub"), and Evolv Technologies, Inc., dba Evolv Technology, Inc., a Delaware corporation (the "Company"). Each of Parent, Merger Sub and the Company are referred to herein as a "Party" and together as the "Parties." Capitalized terms used herein and not otherwise defined shall have the meanings ascribed to them in the Merger Agreement (as defined below).

WHEREAS, the Parties entered into that certain Agreement and Plan of Merger dated as of March 5, 2021 (as may be amended, restated, or otherwise supplemented from time to time, including pursuant to this Amendment, the "Merger Agreement");

WHEREAS, pursuant to Section 11.1 of the Merger Agreement, the Merger Agreement may be amended or modified, in whole or in part, by a duly authorized agreement in writing executed by each of the Parties; and

WHEREAS, the Parties wish to amend the Merger Agreement as set forth in this Amendment.

NOW, THEREFORE, intending to be legally bound and in consideration of the mutual provisions set forth in this Amendment and the Merger Agreement and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby agree as follows:

ARTICLE 1 AMENDMENTS TO THE MERGER AGREEMENT

- Section 1.1 Amendment to Section 2.1(a)(iii) of the Merger Agreement. Section 2.1(a)(iii) of the Merger Agreement is hereby amended by deleting Section 2.1(a)(iii) of the Merger Agreement and replacing it in its entirety with the following:
 - (iii) Company Warrants. Immediately prior to the Effective Time, (x) the Company shall cause each Company Warrant (other than the Finback Warrant) that is issued and outstanding immediately prior to the Effective Time to be either exercised in full on a cash or cashless basis or terminated without exercise and (y) (A) the Company shall cause the portion of the Finback Warrant that is vested as of immediately prior to the Effective Time to be either exercised in full on a cash or cashless basis or terminated without exercise and (B) the portion of the Finback Warrant that is unvested as of immediately prior to the Effective Time (such portion, the "Unvested Finback Warrant") shall be automatically, without any action on the part of Parent, the Company or the Company Warrant holder, converted into a warrant (the "Parent Finback Warrant") to acquire shares of Parent Common Stock in accordance with this Section 2.1(a)(iii), in each case, in accordance with the respective terms of such Company Warrant (such actions, collectively the "Company Warrant Settlement"). Such Parent Finback Warrant as so assumed and converted shall continue to have, and shall be subject to, the same terms and conditions as applied to the Finback Warrant immediately prior to the Effective Time. As of the Effective Time, such Parent Finback Warrant as so assumed and converted shall be for that number of shares of Parent Common Stock determined by multiplying the number of shares of the Company Common Stock subject to the unvested portion of such Finback Warrant immediately prior to the Effective Time by the Exchange Ratio, which product shall be rounded down to the nearest whole number of shares, at a per share exercise price determined by dividing the per share exercise price of the Unvested Finback Warrant immediately prior to the Effective Time by the Exchange Ratio, which quotient shall be rounded down to the nearest whole cent. After the Company Warrant Settlement, all of the Company Warrants shall thereafter cease to have any rights wit
- Section 1.2 Amendment to Section 2.1 of the Merger Agreement. Section 2.1 of the Merger Agreement is hereby amended by (x) deleting section 2.1(c)(ii) of the Merger Agreement and (y) adding the following provisions as new Section 2.1(d) and Section 2.1(e) of the Merger Agreement and the current Section 2.1(d) of the Merger Agreement (and all references thereto) shall be changed to Section 2.1(f):
 - (d) Treatment of Company RSUs. At the Effective Time, each outstanding award of Company RSUs, whether vested or unvested, shall, automatically and without any required action on the part of the holder thereof, cease to represent an award of restricted stock units covering shares of Company Common Stock and shall be converted into (i) an award of restricted stock units covering a number of shares of Parent Common Stock determined in accordance with this Section 2.1(d) (each, an "Assumed RSU"), and (ii) the right to receive a number of Earn-Out Shares in accordance with Section 2.8. Each award of Assumed RSUs shall be subject to the same terms and conditions as were applicable to such corresponding award of Company RSUs immediately prior to the Effective Time (including applicable vesting conditions), except to the extent such terms or conditions are rendered inoperative by the Transactions. Each Assumed RSU shall represent an award of restricted stock units of Parent Common Stock at such exercise price, in each case, determined as follows and as set forth in the Allocation Statement: the number of shares of Parent Common Stock awarded pursuant to the Assumed RSU shall be equal to (rounded down to the nearest whole number): (A) the number of shares of Company Common Stock covered by such Company RSU immediately prior to the Effective Time, multiplied by (B) the Exchange Ratio.
 - (e) <u>Parent Actions.</u> Parent shall take all actions that are necessary for the assumption of the Company Options and the Company RSUs pursuant to this <u>Section 2.1</u>, including the reservation, issuance and listing of Parent Common Stock as necessary to consummate the transactions contemplated by this <u>Section 2.1</u>. If registration of any plan interests in the Stock Plan or other Company Benefit Plans or the shares of Parent Common Stock issuable thereunder is required under the Securities Act, Parent shall file with the SEC on the Closing Date a registration statement on Form S-8 with respect to such interests or Parent Common Stock, and shall use its reasonable best efforts to maintain the effectiveness of such registration statement for so long as the relevant Stock Plan or other Company Benefit Plans, as applicable, remain in effect and such registration of interests therein or the shares of Parent Common Stock issuable thereunder continues to be required."

- Section 1.3 Amendment to Section 2.5(a) of the Merger Agreement. Section 2.5(a) of the Merger Agreement is hereby amended by deleting Section 2.5(a) of the Merger Agreement and replacing it in its entirety with the following:
 - (a) No later than the third (3rd) Business Day preceding the anticipated Closing Date, the Company shall prepare and deliver to Parent a statement containing the following information (the "<u>Allocation Statement</u>"):
 - (i) The allocation of the Aggregate Merger Consideration to the holders of Company Common Stock, including, for the avoidance of doubt, to

the holders of any outstanding award of Company RSUs, whether vested or unvested, after giving effect to the Convertible Notes Conversion, the Preferred Stock Conversion, and the Company Warrant Settlement;

- (ii) Each holder and the number of shares of Parent Common Stock constituting the Per Share Merger Consideration receivable by such holder of Company Common Stock pursuant to the terms of this Agreement, including, for the avoidance of doubt, to the holders of any outstanding award of Company RSUs, whether vested or unvested, after giving effect to the Preferred Stock Conversion, the Convertible Notes Conversion and the Company Warrant Settlement.
- (iii) Each Assumed Option that will be outstanding as of the Closing, and with respect to such Assumed Option, the number of shares of Parent Common Stock issuable upon exercise of such Assumed Option and the exercise price of such Assumed Option, in each case computed in accordance with Section 2.1(c).
- (iv) The number of shares of Company Common Stock issuable under the Unvested Finback Warrant (assuming it were fully vested as of immediately prior to the Effective Time) and the number of shares of Parent Common Stock issuable upon the exercise of the Parent Finback Warrant and the exercise price of such Parent Finback Warrant computed in accordance with Section 2.1(a)(iii).

3

- Section 1.4 Amendment to Section 2.8(a)(i) of the Merger Agreement. Section 2.8(a)(i) of the Merger Agreement is hereby amended by deleting Section 2.8(a) (i) of the Merger Agreement and replacing it in its entirety with the following:
 - (i) Following the Closing, and as additional consideration in respect of shares of Company Common Stock, including the aggregate number of shares of Company Common Stock issuable upon exercise or settlement of all Company RSUs (vested or unvested) held by the Company Earn-Out Holder as of immediately prior to the Effective Time (after giving effect to the Preferred Stock Conversion, the Convertible Notes Conversion and the Company Warrant Settlement), the Unvested Finback Warrant and the Company Options, within ten (10) Business Days after the occurrence of a Triggering Event, Parent shall issue or cause to be issued to Persons who held such shares of Company Common Stock(after giving effect to the Preferred Stock Conversion, the Convertible Notes Conversion and the Company Warrant Settlement), the Unvested Finback Warrant, Company Options and Company RSUs (vested or unvested) immediately prior to the Effective Time (the "Company Earn-Out Holders"), in accordance with their respective Earn-Out Pro Rata Shares, the following shares of Parent Common Stock, as applicable (which shall be equitably adjusted for stock splits, reverse stock splits, stock dividends, recapitalizations, reclassifications, combinations, exchanges of shares or other like changes or transactions with respect to shares of Parent Common Stock) (as so adjusted, the "Earn-Out Shares"), upon the terms and subject to the conditions set forth in this Agreement and the other agreements contemplated hereby:
 - (A) upon the occurrence of Triggering Event I, a one-time aggregate issuance of 5,000,000 Earn-Out Shares;
 - (B) upon the occurrence of Triggering Event II, a one-time aggregate issuance of 5,000,000 Earn-Out Shares; and
 - (C) upon the occurrence of Triggering Event III, a one-time aggregate issuance of 5,000,000 Earn-Out Shares.
- Section 1.5 Amendment to Section 2.8(c) of the Merger Agreement. Section 2.8(c) of the Merger Agreement is hereby amended by deleting Section 2.8(c) of the Merger Agreement and replacing it in its entirety with the following:
 - (c) <u>Earn-Out Cap</u>; <u>Service Requirements</u>. For the avoidance of doubt, the Company Earn-Out Holders shall be entitled to receive Earn-Out Shares upon the occurrence of each Triggering Event (or Acceleration Event, if applicable); <u>provided, however</u>, that each Triggering Event (or Acceleration Event, if applicable) shall only occur once, if at all, and in no event shall the Company Earn-Out Holders be entitled to receive more than 15,000,000 Earn-Out Shares (subject to adjustment for stock splits, reverse stock splits, stock dividends, reorganizations, recapitalizations, combinations, exchanges of shares or other like changes or transactions with respect to shares of Parent Common Stock). Notwithstanding anything in this Agreement to the contrary, any Earn-Out Shares issuable under this <u>Section 2.8</u> to any Company Earn-Out Holder in respect of Company Options or Company RSUs held by such Company Earn-Out Holder as of immediately prior to the Effective Time shall be issued to such Company Earn-Out Holder only if such Company Earn-Out Holder continues to provide services (whether as an employee, director or individual independent contractor) to Parent or one of its Subsidiaries through the date of the occurrence of the corresponding Triggering Event (or Acceleration Event, if applicable) that causes such Earn-Out Shares to become issuable. Any Earn-Out Shares that are forfeited pursuant to the preceding sentence shall be reallocated to the other Company Earn-Out Holders who remain entitled to receive Earn-Out Shares in accordance with their respective Earn-Out PoRata Shares.

- Section 1.6 Amendment to Section 2.8(d)(iii) of the Merger Agreement Section 2.8(d)(iii) of the Merger Agreement is hereby amended by deleting Section 2.8(d)(iii) of the Merger Agreement and replacing it in its entirety with the following:
 - (iii) "Earn-Out Pro Rata Share" means, for each Company Earn-Out Holder, a percentage determined by the quotient of:
 - (D) The sum of (i) the total number of shares of Company Common Stock held by the Company Earn-Out Holder as of immediately prior to the Effective Time, including, for the avoidance of doubt, the aggregate number of shares of Company Common Stock issuable upon exercise or settlement of all Company RSUs (vested or unvested) held by the Company Earn-Out Holder as of immediately prior to the Effective Time (after giving effect to the Convertible Notes Conversion, the Preferred Stock Conversion and the Company Warrant Settlement); <u>plus</u> (ii) (x) the total number of shares of Company Common Stock issuable assuming full exercise of all Company Options that are held by the Company Earn-Out Holder as of immediately prior to the Effective Time and that have an exercise price less than the Per Share Value, <u>minus</u> (y) the number of shares of Company Common Stock equal to (I) the aggregate of the exercise prices of the Company Options described in clause (ii)(x) above, <u>divided by</u> (II) the Per Share Value; <u>plus</u> (iii) with respect to Finback Evolv OBH, LLC, only (x) the total number of shares of Company Common Stock equal to (I) the exercise price of the Unvested Finback Warrant were fully vested and fully exercised <u>minus</u> (y) the number of shares of Company Common Stock equal to (I) the exercise price of the Unvested Finback Warrant described in clause (iii)(x) above, <u>divided by</u> (II) the Per Share Value; <u>divided by</u>
 - (E) The sum of (i) the total number of shares of Company Common Stock held by all Company Earn-Out Holders as of immediately prior to the Effective Time, including, for the avoidance of doubt, the aggregate number of shares of Company Common Stock issuable upon exercise or settlement of all Company RSUs (vested or unvested) held by the Company Earn-Out Holder as of immediately prior to the Effective Time (after giving effect to the Convertible Notes Conversion, the Preferred Stock Conversion and the Company Warrant Settlement); <u>plus</u> (ii) (x) the total number of shares of Company Common Stock issuable assuming full exercise of all Company Options that are held by all Company Earn-Out Holders as of immediately prior to the Effective Time and that have an exercise price less than the Per Share Value, <u>minus</u> (y) the number of shares of Company Common Stock equal to (I) the aggregate of the exercise prices of the Company Options described in clause (x) above, <u>divided by</u> (II) the Per Share Value <u>plus</u> (iii) (x) the total number of shares of Company Common Stock issuable assuming the Unvested Finback Warrant were fully vested and fully exercised <u>minus</u> (y) the number of shares of Company Common Stock equal to (I) the exercise price of the Unvested

- Section 1.7 Amendment to Section 7.1(a) of the Merger Agreement. Section 7.1 (a) of the Merger Agreement is hereby amended by deleting Section 7.1(a) of the Merger Agreement and replacing it in its entirety with the following:
 - (a) As promptly as practicable following the execution and delivery of this Agreement, Parent shall prepare, with the assistance of the Company, and cause to be filed with the SEC a registration statement on Form S-4 (as amended or supplemented from time to time, and including the Proxy Statement contained therein, the "Registration Statement") in connection with the registration under the Securities Act of certain shares of the Parent Common Stock to be issued under this Agreement, which Registration Statement will also contain the Proxy Statement. Each of Parent and the Company shall use its reasonable best efforts to cause the Registration Statement and the Proxy Statement to comply with the rules and regulations promulgated by the SEC, to have the Registration Statement declared effective under the Securities Act as promptly as practicable after such filing and to keep the Registration Statement effective as long as is necessary to consummate the Merger. In addition to Section 5.5(b), each of Parent and the Company shall furnish all information concerning it as may reasonably be requested by the other Party in connection with such actions and the preparation of the Registration Statement and the Proxy Statement. Promptly after the Registration Statement is declared effective under the Securities Act, Parent will cause the Proxy Statement to be mailed to stockholders of Parent.
- Section 1.8 Amendment to Section 7.9 of the Merger Agreement Section 7.9 of the Merger Agreement is hereby amended by deleting Section 7.9 of the Merger Agreement and replacing it in its entirety with the following:
 - 7.9 Parent Incentive Plan. Parent shall, prior to the Effective Time, approve and adopt the Parent Incentive Plan and the Parent ESPP, in each case to be effective in connection with the Closing. The Parent Incentive Plan shall provide for an initial aggregate share reserve thereunder equal to approximately 9.53% of the number of shares of Parent Common Stock on a fully diluted basis at the Closing, and shall otherwise be in the form mutually acceptable to Parent and the Company. The Parent ESPP shall provide for an initial aggregate share reserve thereunder equal to 1.50% of the number of shares of Parent Common Stock on a fully diluted basis at the Closing, and shall otherwise be in the form mutually acceptable to Parent and the Company. Subject to approval of the Parent Incentive Plan and the Parent ESPP by the Parent stockholders, following the Effective Time Parent shall file an effective Form S-8 Registration Statement with the SEC with respect to the shares of Parent Class A Stock issuable under the Parent Incentive Plan and the Parent ESPP and shall use commercially reasonable efforts to maintain the effectiveness of such Form S-8 Registration Statement for so long as awards granted pursuant to the Parent Incentive Plan and/or Parent ESPP remain outstanding.
 - Section 1.9 Amendments to Exhibit A to the Merger Agreement.
- (a) The definition of "Company Fully Diluted Capital Stock" in Exhibit A to the Merger Agreement is hereby amended and restated in its entirety as follows:

"Company Fully Diluted Capital Stock" means, without duplication, a number of shares of Company Common Stock equal to (a) the aggregate number of shares of Company Common Stock that are issued and outstanding as of immediately prior to the Effective Time after giving effect to the Convertible Notes Conversion, the Preferred Stock Conversion and the Company Warrant Settlement; <u>plus</u> (b) the aggregate number of shares of Company Common Stock equal to the total number of shares of Company Common Stock issuable assuming full exercise of all Company Options that are outstanding as of immediately prior to the Effective Time and that have an exercise price less than the Per Share Value; <u>minus</u> (c) the Treasury Shares outstanding immediately prior to the Effective Time; <u>plus</u> (d) the aggregate number of shares of Company Common Stock issuable upon exercise or settlement of all Company RSUs (vested and unvested) outstanding as of immediately prior to the Effective Time; <u>plus</u> (e) the aggregate number of shares of Company Common Stock issuable assuming the Unvested Finback Warrant were fully vested and fully exercised.

 ϵ

(b) The definition of "Parent Restated Bylaws" in Exhibit A to the Merger Agreement is hereby amended and restated in its entirety as follows:

"Parent Restated Bylaws" means that certain second amended and restated bylaws of Parent, in form and substance reasonably acceptable to Parent and the Company, which, except as otherwise expressly provided therein or otherwise approved by the board of directors of Parent in accordance with the terms thereof, shall include customary lock-up restrictions for a period of 180 days following the Closing on those shares of Parent Common Stock issued at or in connection with Closing (i) which constitute Per Share Merger Consideration or (ii) which are issued to directors, officers and employees of Parent upon the settlement or exercise of stock options or other equity awards outstanding as of immediately following the Closing in respect of awards of Company equity interests outstanding as of immediately prior to the Closing.

(c) Exhibit A to the Merger Agreement shall be amended by adding the following definitions in the proper alphabetical order:

"Assumed RSUs" has the meaning set forth in Section 2.8(d).

"Company RSUs" means restricted stock units covering shares of Company Common Stock granted pursuant to the Stock Plan.

"Unvested Finback Warrant" has the meaning set forth in Section 2.1(a)(iii).

ARTICLE 2 MISCELLANEOUS

Section 2.1 No Other Amendment. Except to the extent that any provisions of or any Exhibits or Schedules to the Merger Agreement are expressly amended by Article 1 of this Amendment, all terms and conditions of the Merger Agreement including, without limitation, the last sentence of Section 2.1(b), and all other documents, instruments and agreements executed thereunder, shall remain in full force and effect pursuant to the terms thereof. In the event of any inconsistency or contradiction between the terms of this Amendment and the Merger Agreement, the provisions of this Amendment shall prevail and control.

"herewith," "hereunder" and words of similar import shall, unless otherwise stated, be construed to refer to the Merger Agreement as amended by this Amendment. No reference to this Amendment need be made in any instrument or document at any time referring to the Merger Agreement and a reference to the Merger Agreement in any such instrument or document shall be deemed to be a reference to the Merger Agreement as amended by this Amendment.

Section 2.3 General Provisions. Except as set forth in Article 1 of this Amendment, the provisions of Article XI (Miscellaneous) and Exhibit A (Certain Definitions) of the Merger Agreement apply equally to this Amendment and are hereby deemed incorporated by reference.

[Signature Page Follows]

8

IN WITNESS WHEREOF, the Parties have duly executed this Amendment to be effective as of the Amendment Date.

NEWHOLD INVESTMENT CORP.

By: /s/ Kevin Charlton
Name: Kevin Charlton
Title: Chief Executive Officer

NHIC SUB INC.

By: /s/ Charlie Baynes-Reid

Name: Charlie Baynes-Reid

Title: President, Secretary and Treasurer

[Signature Page to First Amendment to Agreement and Plan Merger]

IN WITNESS WHEREOF, the Parties have duly executed this Amendment to be effective as of the Amendment Date.

EVOLV TECHNOLOGIES, INC.

By: /s/ Peter George

Name: Peter George

Title: Chief Executive Officer

[Signature Page to First Amendment to Agreement and Plan Merger]

SECOND AMENDED AND RESTATED CERTIFICATE OF INCORPORATION OF NEWHOLD INVESTMENT CORP.

NewHold Investment Corp. (the "Corporation"), a corporation organized and existing under the General Corporation Law of the State of Delaware (the "DGCL"), does hereby certify as follows:

- 1. The name of the Corporation is NewHold Investment Corp. The Corporation was incorporated under the name NewHold Investment Corp. by the filing of its original Certificate of Incorporation with the Secretary of State of the State of Delaware on January 24, 2020 (the "Original Certificate").
- 2. An Amended and Restated Certificate of Incorporation, which amended and restated the Original Certificate in its entirety, was filed with the Secretary of State of the State of Delaware on July 30, 2020 (as amended from time to time, the "Existing Certificate").
- 3. This Second Amended and Restated Certificate of Incorporation (the "Second Amended and Restated Certificate"), which amends and restates the Existing Certificate in its entirety, has been approved by the Board of Directors of the Corporation (the "Board of Directors") in accordance with Sections 242 and 245 of the DGCL and has been adopted by the stockholders of the Corporation at a meeting of the stockholders of the Corporation in accordance with the provisions of Section 211 of the DGCL.
- 4. The text of the Existing Certificate is hereby amended and restated by this Second Amended and Restated Certificate to read in its entirety as set forth in EXHIBIT A attached hereto.
 - 5. This Second Amended and Restated Certificate shall become effective on the date of filing with the Secretary of State of the State of Delaware.

IN WITNESS WHEREOF, NewHold Investment Corp. has caused this Second Amended and Restated Certificate to be signed by a duly authorized officer of the Corporation, on July 16, 2021.

NEWHOLD INVESTMENT CORP.

By: /s/ Peter George
Name: Peter George

Title: Chief Executive Officer

[Signature Page to Second Amended and Restated Certificate of Incorporation]

EXHIBIT A

ARTICLE I NAME

The name of the corporation is Evolv Technologies Holdings, Inc. (the "Corporation").

ARTICLE II REGISTERED OFFICE AND AGENT

The address of the Corporation's registered office in the State of Delaware is [3411 Silverside Road, Tatnall Building, #104, in the City of Wilmington, County of New Castle, State of Delaware, 19810, and the name of its registered agent at such address is Corporation Creations Network Inc].

ARTICLE III PURPOSE

The purpose of the Corporation is to engage in any lawful act or activity for which corporations may be organized under the General Corporation Law of the State of Delaware (the "DGCL") as it now exists or may hereafter be amended and supplemented.

ARTICLE IV CAPITAL STOCK

The Corporation is authorized to issue two classes of stock to be designated, respectively, 'Class A Common Stock' and "Preferred Stock." The total number of shares of capital stock which the Corporation shall have authority to issue is [1,200,000,000]. The total number of shares of Class A Common Stock that the Corporation is authorized to issue is [1,100,000,000], having a par value of \$0.0001 per share, and the total number of shares of Preferred Stock that the Corporation is authorized to issue is [100,000,000], having a par value of \$0.0001 per share.

The designations and the powers, privileges and rights, and the qualifications, limitations or restrictions thereof in respect of each class of capital stock of the Corporation are as follows:

A. CLASS A COMMON STOCK.

1. General. The voting, dividend, liquidation, and other rights and powers of the Class A Common Stock are subject to and qualified by the rights, powers and preferences of any series of Preferred Stock as may be designated by the Board of Directors of the Corporation (the "Board of Directors") and outstanding from time to time and that certain Stockholder Agreement, dated as of March 5, 2021, by and among NewHold Investment Corp. and Motorola Solutions, Inc. (as such agreement may be amended, restated, amended and restated, supplemented or otherwise modified from time to time, the "Stockholder Agreement").

on each matter submitted to a vote of stockholders and shall be entitled to one (1) vote for each share of Class A Common Stock held of record by such holder as of the record date for determining stockholders entitled to vote on such matter. Except as otherwise required by law, holders of Class A Common Stock, as such, shall not be entitled to vote on any amendment to this Second Amended and Restated Certificate (including any Certificate of Designation (as defined below)) that relates solely to the rights, powers, preferences (or the qualifications, limitations or restrictions thereof) or other terms of one or more outstanding series of Preferred Stock if the holders of such affected series are entitled, either separately or together with the holders of one or more other such series, to vote thereon pursuant to this Second Amended and Restated Certificate (including any Certificate of Designation) or pursuant to the DGCL.

Subject to the rights of any holders of any outstanding series of Preferred Stock, the number of authorized shares of Class A Common Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the stock of the Corporation entitled to vote, irrespective of the provisions of Section 242(b)(2) of the DGCL.

- 3. <u>Dividends</u>. Subject to applicable law and the rights and preferences of any holders of any outstanding series of Preferred Stock, the holders of Class A Common Stock, as such, shall be entitled to the payment of dividends on the Class A Common Stock when, as and if declared by the Board of Directors in accordance with applicable law.
- 4. <u>Liquidation</u>. Subject to the rights and preferences of any holders of any shares of any outstanding series of Preferred Stock, in the event of any liquidation, dissolution or winding up of the Corporation, whether voluntary or involuntary, the funds and assets of the Corporation that may be legally distributed to the Corporation's stockholders shall be distributed among the holders of the then outstanding Class A Common Stock *pro rata* in accordance with the number of shares of Class A Common Stock held by each such holder.

B. PREFERRED STOCK

Shares of Preferred Stock may be issued from time to time in one or more series, each of such series to have such terms as stated or expressed herein and in the resolution or resolutions providing for the creation and issuance of such series adopted by the Board of Directors as hereinafter provided.

Authority is hereby expressly granted to the Board of Directors from time to time to issue the Preferred Stock in one or more series, and in connection with the creation of any such series, by adopting a resolution or resolutions providing for the issuance of the shares thereof and by filing a certificate of designation relating thereto in accordance with the DGCL (a "Certificate of Designation"), to determine and fix the number of shares of such series and such voting powers, full or limited, or no voting powers, and such designations, preferences and relative participating, optional or other special rights, and qualifications, limitations or restrictions thereof, including without limitation thereof, dividend rights, conversion rights, redemption privileges and liquidation preferences, and to increase or decrease (but not below the number of shares of such series then outstanding) the number of shares of any series as shall be stated and expressed in such resolutions, all to the fullest extent now or hereafter permitted by the DGCL. Without limiting the generality of the foregoing, the resolution or resolutions providing for the creation and issuance of any series of Preferred Stock may provide that such series shall be superior or rank equally or be junior to any other series of Preferred Stock to the extent permitted by law and this Second Amended and Restated Certificate (including any Certificate of Designation). Except as otherwise required by law, holders of any series of Preferred Stock shall be entitled only to such voting rights, if any, as shall expressly be granted thereto by this Second Amended and Restated Certificate (including any Certificate of Designation).

The number of authorized shares of Preferred Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the stock of the Corporation entitled to vote, irrespective of the provisions of Section 242(b)(2) of the DGCL.

ARTICLE V BOARD OF DIRECTORS

For the management of the business and for the conduct of the affairs of the Corporation it is further provided that:

A. Subject to the Stockholder Agreement and the special rights of the holders of one or more outstanding series of Preferred Stock to elect directors, the initial directors of the Corporation shall be classified with respect to the time for which they severally hold office into three classes, designated as Class I, Class II and Class III. The initial Class I directors shall serve for a term expiring at the first annual meeting of the stockholders following the date of this Second Amended and Restated Certificate; and the initial Class III directors shall serve for a term expiring at the third annual meeting following the date of this Second Amended and Restated Certificate; and the initial Class III directors shall serve for a term expiring at the third annual meeting following the date of this Second Amended and Restated Certificate, subject to the Stockholders of the Corporation beginning with the first annual meeting of the stockholders following the date of this Second Amended and Restated Certificate, subject to the Stockholder Agreement and the special rights of the holders of one or more outstanding series of Preferred Stock to elect directors, the successors of the class of directors whose term expires at that meeting shall be elected to hold office for a term expiring at the annual meeting of the stockholders held in the third year following the year of their election. Each director shall hold office until his or her successor is duly elected and qualified or until his or her earlier death, resignation, disqualification or removal. No decrease in the number of directors shall shorten the term of any incumbent director. Subject to the Stockholders Agreement, the Board of Directors is authorized to assign members of the Board of Directors already in office to Class I, Class II and Class III.

B. Except as otherwise expressly provided by the DGCL or this Second Amended and Restated Certificate, the business and affairs of the Corporation shall be managed by or under the direction of the Board of Directors. The number of directors which shall constitute the whole Board of Directors shall be fixed exclusively by one or more resolutions adopted from time to time by the Board of Directors.

- C. Subject to the Stockholder Agreement and the special rights of the holders of one or more outstanding series of Preferred Stock to elect directors, the Board of Directors or any individual director may be removed from office at any time, but only for cause and only by the affirmative vote of the holders of at least two-thirds (66 and 2/3%) of the voting power of all of the then outstanding shares of voting stock of the Corporation entitled to vote at an election of directors.
- D. Subject to the Stockholder Agreement and the special rights of the holders of one or more outstanding series of Preferred Stock to elect directors, except as otherwise provided by law, any vacancies on the Board of Directors resulting from death, resignation, disqualification, retirement, removal or other causes and any newly created directorships resulting from any increase in the number of directors shall be filled exclusively by the affirmative vote of a majority of the directors then in office, even though less than a quorum, or by a sole remaining director (other than any directors elected by the separate vote of one or more outstanding series of Preferred Stock), and shall not be filled by the stockholders. Subject to the Stockholder Agreement, any director appointed in accordance with the preceding sentence shall hold office until the expiration of the term to which such director shall have been appointed or until his or her earlier death, resignation, retirement, disqualification, or removal.
- E. Whenever the holders of any one or more series of Preferred Stock issued by the Corporation shall have the right, voting separately as a series or separately as a class with one or more such other series, to elect directors at an annual or special meeting of stockholders, the election, term of office, removal and other features of such

directorships shall be governed by the terms of this Second Amended and Restated Certificate (including any Certificate of Designation). Notwithstanding anything to the contrary in this Article V, the number of directors that may be elected by the holders of any such series of Preferred Stock shall be in addition to the number fixed pursuant to paragraph B of this Article V, and the total number of directors constituting the whole Board of Directors shall be automatically adjusted accordingly. Except as otherwise provided in the Certificate of Designation(s) in respect of one or more series of Preferred Stock, whenever the holders of any series of Preferred Stock having such right to elect additional directors are divested of such right pursuant to the provisions of such Certificate of Designation(s), the terms of office of all such additional directors elected by the holders of such series of Preferred Stock, or elected to fill any vacancies resulting from the death, resignation, disqualification or removal of such additional directors, shall forthwith terminate (in which case each such director thereupon shall cease to be qualified as, and shall cease to be, a director) and the total authorized number of directors of the Corporation shall automatically be reduced accordingly.

F. In furtherance and not in limitation of the powers conferred by statute, the Board of Directors is expressly authorized, subject to the Stockholder Agreement, to adopt, amend or repeal the Amended and Restated Bylaws of the Corporation (as amended and/or restated from time to time, the "Bylaws"). In addition to any vote of the holders of any class or series of stock of the Corporation required by applicable law or by this Second Amended and Restated Certificate (including any Certificate of Designation in respect of one or more series of Preferred Stock) or the Bylaws of the Corporation, the adoption, amendment or repeal of the Bylaws of the Corporation by the stockholders of the Corporation shall require the affirmative vote of the holders of at least two-thirds (66 and 2/3%) of the voting power of all of the then outstanding shares of voting stock of the Corporation entitled to vote generally in an election of directors.

G. The directors of the Corporation need not be elected by written ballot unless the Bylaws so provide.

ARTICLE VI STOCKHOLDERS

- A. Any action required or permitted to be taken by the stockholders of the Corporation must be effected at an annual or special meeting of the stockholders of the Corporation, and shall not be taken by written consent in lieu of a meeting. Notwithstanding the foregoing, any action required or permitted to be taken by the holders of any series of Preferred Stock, voting separately as a series or separately as a class with one or more other such series, may be taken without a meeting, without prior notice and without a vote, to the extent expressly so provided by the applicable Certificate of Designation relating to such series of Preferred Stock, if a consent or consents in writing, setting forth the action so taken, shall be signed by the holders of outstanding shares of the relevant series of Preferred Stock having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares entitled to vote thereon were present and voted and shall be delivered to the Corporation in accordance with the applicable provisions of the DGCL.
- B. Subject to the Stockholder Agreement and the special rights of the holders of one or more series of Preferred Stock, special meetings of the stockholders of the Corporation may be called, for any purpose or purposes, at any time only by or at the direction of the Board of Directors, the Chairperson of the Board of Directo
- C. Advance notice of stockholder nominations for the election of directors and of other business proposed to be brought by stockholders before any meeting of the stockholders of the Corporation shall be given in the manner provided in the Bylaws of the Corporation.

ARTICLE VII LIABILITY

No director of the Corporation shall have any personal liability to the Corporation or its stockholders for monetary damages for any breach of fiduciary duty as a director, except to the extent such exemption from liability or limitation thereof is not permitted under the DGCL as the same exists or hereafter may be amended. Any amendment, repeal or modification of this <u>Article VII</u>, or the adoption of any provision of the Second Amended and Restated Certificate inconsistent with this <u>Article VII</u>, shall not adversely affect any right or protection of a director of the Corporation with respect to any act or omission occurring prior to such amendment, repeal, modification or adoption. If the DGCL is amended after approval by the stockholders of this <u>Article VII</u> to authorize corporate action further eliminating or limiting the personal liability of directors, then the liability of a director of the Corporation shall be eliminated or limited to the fullest extent permitted by the DGCL as so amended.

ARTICLE VIII INDEMNIFICATION

The Corporation shall have the power to provide rights to indemnification and advancement of expenses to its current and former officers, directors, employees and agents and to any person who is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise.

ARTICLE IX AMENDMENTS

- A. Notwithstanding anything contained in this Second Amended and Restated Certificate to the contrary, in addition to any vote required by applicable law, the following provisions in this Second Amended and Restated Certificate may be amended, altered, repealed or rescinded, in whole or in part, or any provision inconsistent therewith or herewith may be adopted, only by the affirmative vote of the holders of at least two-thirds (66 and 2/3%) of the total voting power of all the then outstanding shares of stock of the Corporation entitled to vote thereon, voting together as a single class: Part B of Article IV, Article VI, Article VII, Article VIII, and this Article IX.
- B. If any provision or provisions of this Second Amended and Restated Certificate shall be held to be invalid, illegal or unenforceable as applied to any circumstance for any reason whatsoever: (i) the validity, legality and enforceability of such provisions in any other circumstance and of the remaining provisions of this Second Amended and Restated Certificate (including, without limitation, each portion of any paragraph of this Second Amended and Restated Certificate containing any such provision held to be invalid, illegal or unenforceable that is not itself held to be invalid, illegal or unenforceable with any paragraph of the fullest extent permitted by applicable law, in any way be affected or impaired thereby and (ii) to the fullest extent permitted by applicable law, the provisions of this Second Amended and Restated Certificate (including, without limitation, each such portion of any paragraph of this Second Amended and Restated Certificate containing any such provision held to be invalid, illegal or unenforceable) shall be construed so as to permit the Corporation to protect its directors, officers, employees and agents from personal liability in respect of their good faith service to or for the benefit of the Corporation to the fullest extent permitted by law.

Amended and Restated Bylaws of

Evolv Technologies Holdings, Inc.

(a Delaware corporation)

Table of Contents

		<u>Page</u>
Article I - Co	Corporate Offices	1
1.1 1.2	Registered Office Other Offices	1 1
Article II - N	Meetings of Stockholders	1
2.1 2.2 2.3 2.4 2.5 2.6 2.7 2.8 2.9 2.10 2.11 2.12 2.13 2.14	Place of Meetings Annual Meeting Special Meeting Notice of Business to be Brought before a Meeting. Notice of Nominations for Election to the Board. Notice of Stockholders' Meetings Quorum Adjourned Meeting; Notice Conduct of Business Voting Record Date for Stockholder Meetings and Other Purposes Proxies List of Stockholders Entitled to Vote Inspectors of Election	1 1 1 2 5 8 9 9 9 9 10 10 11
2.15 Article III - 1	Delivery to the Corporation.	12
3.1 3.2 3.3 3.4 3.5 3.6 3.7	Powers Number of Directors Election, Qualification and Term of Office of Directors Resignation and Vacancies Place of Meetings; Meetings by Telephone Regular Meetings Special Meetings; Notice Ouorum	12 12 12 12 13 13 13
3.8 3.9 3.10	Board Action without a Meeting Fees and Compensation of Directors	14 14 14
Article IV -	Committees	15
4.1 4.2 4.3 4.4	Committees of Directors Committee Minutes Meetings and Actions of Committees Subcommittees.	15 15 15
Article V - C	Officers	16
5.1 5.2 5.3 5.4	Officers Appointment of Officers Subordinate Officers Removal and Resignation of Officers	16 16 16
	i	

Table of Contents (continued)

		<u>Page</u>
5.5	Vacancies in Offices	16
5.6	Representation of Shares of Other Corporations	17
5.7	Authority and Duties of Officers	17
5.8	Compensation.	17
Article V	VI - Records	17

Article V	I - General Matters	17
7.1	Execution of Corporate Contracts and Instruments	17
7.2	Stock Certificates	18
7.2	Special Designation of Certificates.	18
	Lost Certificates Lost Certificates	18
7.5	Shares Without Certificates	19
7.6	Construction; Definitions	19
7.7	Dividends	19
	Fiscal Year	19
7.9	Seal Seal	19
	Transfer of Stock	19
	Stock Transfer Agreements	20
	Registered Stockholders	20
	Waiver of Notice	20
7.13	Thanks of A color	20
Article V	II - Notice	20
8.1	Delivery of Notice; Notice by Electronic Transmission	20
Article IX	- Lock-Up	21
9.1	Lock-Up	21
Article X	- Indemnification	23
10.1	Indemnification of Directors and Officers	23
	Indemnification of Others	23
	Prepayment of Expenses	23
	Determination; Claim	23
	Non-Exclusivity of Rights	23
	Insurance	24
	Other Indemnification	24
	Continuation of Indemnification	24
	Amendment or Repeal; Interpretation	24
Article X	- Amendments	25
Article X	I - Forum Selection	25
Article X	II - Definitions	26

Amended and Restated Bylaws of Evolv Technologies Holdings, Inc.

Article I - Corporate Offices

1.1 Registered Office.

The address of the registered office of Evolv Technologies Holdings, Inc. (the "Corporation") in the State of Delaware, and the name of its registered agent at such address, shall be as set forth in the Corporation's certificate of incorporation, as the same may be amended and/or restated from time to time (the "Certificate of Incorporation").

1.2 Other Offices

The Corporation may have additional offices at any place or places, within or outside the State of Delaware, as the Corporation's board of directors (the "Board") may from time to time establish or as the business of the Corporation may require.

Article II - Meetings of Stockholders

2.1 <u>Place of Meetings</u>.

Meetings of stockholders shall be held at any place, within or outside the State of Delaware, designated by the Board. The Board may, in its sole discretion, determine that a meeting of stockholders shall not be held at any place, but may instead be held solely by means of remote communication as authorized by Section 211(a)(2) of the General Corporation Law of the State of Delaware (the "DGCL"). In the absence of any such designation or determination, stockholders' meetings shall be held at the Corporation's principal executive office.

2.2 <u>Annual Meeting</u>.

The Board shall designate the date and time of the annual meeting. At the annual meeting, directors shall be elected and other proper business properly brought before the meeting in accordance with Section 2.4 may be transacted. The Board may postpone, reschedule or cancel any previously scheduled annual meeting of stockholders.

2.3 Special Meeting.

Special meetings of the stockholders may be called only by such persons and only in such manner as set forth in the Certificate of Incorporation.

No business may be transacted at any special meeting of stockholders other than the business specified in the notice of such meeting. The Board may postpone, reschedule or cancel any previously scheduled special meeting of stockholders.

2.4 Notice of Business to be Brought before a Meeting

- At an annual meeting of the stockholders, only such business shall be conducted as shall have been properly brought before the meeting. To be properly (a) brought before an annual meeting, business must be (i) specified in a notice of meeting given by or at the direction of the Board, (ii) if not specified in a notice of meeting, otherwise brought before the meeting by the Board or the Chairman of the Board or (iii) otherwise properly brought before the meeting by a stockholder present in person who (A) (1) was a record owner of shares of the Corporation both at the time of giving the notice provided for in this Section 2.4 and at the time of the meeting, (2) is entitled to vote at the meeting and (3) has complied with this Section 2.4 in all applicable respects or (B) properly made such proposal in accordance with Rule 14a-8 under the Securities Exchange Act of 1934, as amended, and the rules and regulations thereunder (as so amended and inclusive of such rules and regulations, the "Exchange Act"). The foregoing clause (iii) shall be the exclusive means for a stockholder to propose business to be brought before an annual meeting of the stockholders. The only matters that may be brought before a special meeting are the matters specified in the notice of meeting given by or at the direction of the person calling the meeting pursuant to Section 2.3, and stockholders shall not be permitted to propose business to be brought before a special meeting of the stockholders. For purposes of this Section 2.4, "present in person" shall mean that the stockholder proposing that the business be brought before the annual meeting of the Corporation, or a qualified representative of such proposing stockholder, appear at such annual meeting. A "qualified representative" of such proposing stockholder shall be a duly authorized officer, manager or partner of such stockholder or any other person authorized by a writing executed by such stockholder or an electronic transmission delivered by such stockholder to act for such stockholder as proxy at the meeting of stockholders and such person must produce such writing or electronic transmission, or a reliable reproduction of the writing or electronic transmission, at the meeting of stockholders. Stockholders seeking to nominate persons for election to the Board must comply with Section 2.5, and this Section 2.4 shall not be applicable to nominations except as expressly provided in Section 2.5.
- (b) Without qualification, for business to be properly brought before an annual meeting by a stockholder, the stockholder must (i) provide Timely Notice (as defined below) thereof in writing and in proper form to the Secretary of the Corporation and (ii) provide any updates or supplements to such notice at the times and in the forms required by this Section 2.4. To be timely, a stockholder's notice must be delivered to, or mailed and received at, the principal executive offices of the Corporation not less than ninety (90) days nor more than one hundred twenty (120) days prior to the one-year anniversary of the preceding year's annual meeting; provided, however, that if no annual meeting was held in the preceding year, to be timely, a stockholder's notice must be so delivered, or mailed and received, not earlier than the close of business on the one hundred and twentieth (120th) day prior to such annual meeting and not later than the close of business on the later of the ninetieth (90th) day prior to such annual meeting or, if later, the tenth (10th) day following the day on which public disclosure of the date of such annual meeting was first made by the Corporation; provided, further, that if the date of the annual meeting is more than thirty (30) days before or more than sixty (60) days after such anniversary date, to be timely, a stockholder's notice must be so delivered, or mailed and received, not later than the ninetieth (90th) day prior to such annual meeting or, if later, the tenth (10th) day following the day on which public disclosure of the date of such annual meeting was first made by the Corporation (such notice within such time periods, "Timely Notice"). In no event shall any adjournment or postponement of an annual meeting or the announcement thereof commence a new time period for the giving of Timely Notice as described above.

2

- (c) To be in proper form for purposes of this Section 2.4, a stockholder's notice to the Secretary of the Corporation shall set forth:
- (i) As to each Proposing Person (as defined below), (A) the name and address of such Proposing Person (including, if applicable, the name and address that appear on the Corporation's books and records); and (B) the class or series and number of shares of the Corporation that are, directly or indirectly, owned of record or beneficially owned (within the meaning of Rule 13d-3 under the Exchange Act) by such Proposing Person, except that such Proposing Person shall in all events be deemed to beneficially own any shares of any class or series of the Corporation as to which such Proposing Person has a right to acquire beneficial ownership at any time in the future (the disclosures to be made pursuant to the foregoing clauses (A) and (B) are referred to as "Stockholder Information");
- As to each Proposing Person, (A) the full notional amount of any securities that, directly or indirectly, underlie any "derivative security" (as such term is defined in Rule 16a-1(c) under the Exchange Act) that constitutes a "call equivalent position" (as such term is defined in Rule 16a-1(b) under the Exchange Act) ("Synthetic Equity Position") and that is, directly or indirectly, held or maintained by such Proposing Person with respect to any shares of any class or series of shares of the Corporation, provided that, for the purposes of the definition of "Synthetic Equity Position," the term "derivative security" shall also include any security or instrument that would not otherwise constitute a "derivative security" as a result of any feature that would make any conversion, exercise or similar right or privilege of such security or instrument becoming determinable only at some future date or upon the happening of a future occurrence, in which case the determination of the amount of securities into which such security or instrument would be convertible or exercisable shall be made assuming that such security or instrument is immediately convertible or exercisable at the time of such determination; and, provided, further, that any Proposing Person satisfying the requirements of Rule 13d-1(b)(1) under the Exchange Act (other than a Proposing Person that so satisfies Rule 13d-1(b)(1) under the Exchange Act solely by reason of Rule 13d-1(b)(1)(ii)(E)) shall not be deemed to hold or maintain the notional amount of any securities that underlie a Synthetic Equity Position held by such Proposing Person as a hedge with respect to a bona fide derivatives trade or position of such Proposing Person arising in the ordinary course of such Proposing Person's business as a derivatives dealer, (B) any rights to dividends on the shares of any class or series of shares of the Corporation owned beneficially by such Proposing Person that are separated or separable from the underlying shares of the Corporation, (C) any material pending or threatened legal proceeding in which such Proposing Person is a party or material participant involving the Corporation or any of its officers or directors, or any affiliate of the Corporation, (D) any other material relationship between such Proposing Person, on the one hand, and the Corporation or any affiliate of the Corporation, on the other hand, (E) any direct or indirect material interest in any material contract or agreement of such Proposing Person with the Corporation or any affiliate of the Corporation (including, in any such case, any employment agreement, collective bargaining agreement or consulting agreement), (F) a representation that such Proposing Person intends or is part of a group that intends to deliver a proxy statement or form of proxy to holders of at least the percentage of the Corporation's outstanding capital stock required to approve or adopt the proposal or otherwise solicit proxies from stockholders in support of such proposal and (G) any other information relating to such Proposing Person that would be required to be disclosed in a proxy statement or other filing required to be made in connection with solicitations of proxies or consents by such Proposing Person in support of the business proposed to be brought before the meeting pursuant to Section 14(a) of the Exchange Act (the disclosures to be made pursuant to the foregoing clauses (A) through (G) are referred to as "Disclosable Interests"); provided, however, that Disclosable Interests shall not include any such disclosures with respect to the ordinary course business activities of any broker, dealer, commercial bank, trust company or other nominee who is a Proposing Person solely as a result of being the stockholder directed to prepare and submit the notice required by these bylaws on behalf of a beneficial owner; and

among any of the Proposing Persons or (y) between or among any Proposing Person and any other person or entity (including their names) in connection with the proposal of such business by such stockholder; and (D) any other information relating to such item of business that would be required to be disclosed in a proxy statement or other filing required to be made in connection with solicitations of proxies in support of the business proposed to be brought before the meeting pursuant to Section 14(a) of the Exchange Act; provided, however, that the disclosures required by this Section 2.4(c)(iii) shall not include any disclosures with respect to any broker, dealer, commercial bank, trust company or other nominee who is a Proposing Person solely as a result of being the stockholder directed to prepare and submit the notice required by these bylaws on behalf of a beneficial owner.

For purposes of this <u>Section 2.4</u>, the term "<u>Proposing Person</u>" shall mean (i) the stockholder providing the notice of business proposed to be brought before an annual meeting, (ii) the beneficial owner or beneficial owners, if different, on whose behalf the notice of the business proposed to be brought before the annual meeting is made, and (iii) any participant (as defined in paragraphs (a)(ii)-(vi) of Instruction 3 to Item 4 of Schedule 14A) with such stockholder in such solicitation.

- A Proposing Person shall update and supplement its notice to the Corporation of its intent to propose business at an annual meeting, if necessary, so that the information provided or required to be provided in such notice pursuant to this Section 2.4 shall be true and correct as of the record date for stockholders entitled to vote at the meeting and as of the date that is ten (10) business days prior to the meeting or any adjournment or postponement thereof, and such update and supplement shall be delivered to, or mailed and received by, the Secretary of the Corporation at the principal executive offices of the Corporation not later than five (5) business days after the record date for stockholders entitled to vote at the meeting (in the case of the update and supplement required to be made as of such record date), and not later than eight (8) business days prior to the date for the meeting or, if practicable, any adjournment or postponement thereof (and, if not practicable, on the first practicable date prior to the date to which the meeting has been adjourned or postponed) (in the case of the update and supplement required to be made as of ten (10) business days prior to the meeting or any adjournment or postponement thereof). For the avoidance of doubt, the obligation to update and supplement as set forth in this paragraph or any other Section of these bylaws shall not limit the Corporation's rights with respect to any deficiencies in any notice provided by a stockholder, extend any applicable deadlines hereunder or enable or be deemed to permit a stockholder who has previously submitted notice hereunder to amend or update any proposal or to submit any new proposal, including by changing or adding matters, business or resolutions proposed to be brought before a meeting of the stockholders.
- (e) Notwithstanding anything in these bylaws to the contrary, no business shall be conducted at an annual meeting that is not properly brought before the meeting in accordance with this Section 2.4. The presiding officer of the meeting shall, if the facts warrant, determine that the business was not properly brought before the meeting in accordance with this Section 2.4, and if he or she should so determine, he or she shall so declare to the meeting and any such business not properly brought before the meeting shall not be transacted.

4

- (f) This <u>Section 2.4</u> is expressly intended to apply to any business proposed to be brought before an annual meeting of stockholders other than any proposal made in accordance with Rule 14a-8 under the Exchange Act and included in the Corporation's proxy statement. In addition to the requirements of this <u>Section 2.4</u> with respect to any business proposed to be brought before an annual meeting, each Proposing Person shall comply with all applicable requirements of the Exchange Act with respect to any such business. Nothing in this <u>Section 2.4</u> shall be deemed to affect the rights of stockholders to request inclusion of proposals in the Corporation's proxy statement pursuant to Rule 14a-8 under the Exchange Act.
- (g) For purposes of these bylaws, "public disclosure" shall mean disclosure in a press release reported by a national news service or in a document publicly filed by the Corporation with the Securities and Exchange Commission pursuant to Sections 13, 14 or 15(d) of the Exchange Act.

2.5 Notice of Nominations for Election to the Board.

- Nominations of any person for election to the Board at an annual meeting or at a special meeting (but only if the election of directors is a matter specified in the notice of meeting given by or at the direction of the person calling such special meeting) may be made at such meeting only (i) as provided in that certain Stockholder Agreement, dated as of March 5, 2021, by and among the NewHold Investment Corp. (predecessor to the Corporation) and Motorola Solutions, Inc. (as such agreement may be amended, restated, amended and restated, supplemented or otherwise modified from time to time, the "Stockholder Agreement"), (ii) by or at the direction of the Board, including by any committee or persons authorized to do so by the Board or these bylaws, or (iii) by a stockholder present in person (A) who was a record owner of shares of the Corporation both at the time of giving the notice provided for in this Section 2.5 and at the time of the meeting, (B) is entitled to vote at the meeting, and (C) has complied with this Section 2.5 as to such notice and nomination. For purposes of this Section 2.5, "present in person" shall mean that the stockholder proposing that the business be brought before the meeting of the Corporation, or a qualified representative of such stockholder, appear at such meeting. A "qualified representative" of such proposing stockholder shall be a duly authorized officer, manager or partner of such stockholder or any other person authorized by a writing executed by such stockholder or an electronic transmission delivered by such stockholder to act for such stockholder as proxy at the meeting of stockholders and such person must produce such writing or electronic transmission, or a reliable reproduction of the writing or electronic transmission, at the meeting of stockholders. Other than as provided in the Stockholder Agreement, the foregoing clause (iii) shall be the exclusive means for a stockholder to make any nomination of a person or persons for election to the Board at an annual meeting or s
- (b) (i) Without qualification, for a stockholder to make any nomination of a person or persons for election to the Board at an annual meeting, the stockholder must (1) provide Timely Notice (as defined in Section 2.4) thereof in writing and in proper form to the Secretary of the Corporation, (2) provide the information, agreements and questionnaires with respect to such stockholder and its candidate for nomination as required to be set forth by this Section 2.5 and (3) provide any updates or supplements to such notice at the times and in the forms required by this Section 2.5.
 - (ii) Without qualification, if the election of directors is a matter specified in the notice of meeting given by or at the direction of the person calling a special meeting, then for a stockholder to make any nomination of a person or persons for election to the Board at a special meeting, the stockholder must (i) provide Timely Notice thereof in writing and in proper form to the Secretary of the Corporation at the principal executive offices of the Corporation, (ii) provide the information with respect to such stockholder and its candidate for nomination as required by this Section 2.5 and (iii) provide any updates or supplements to such notice at the times and in the forms required by this Section 2.5. To be timely, a stockholder's notice for nominations to be made at a special meeting must be delivered to, or mailed and received at, the principal executive offices of the Corporation not earlier than the one hundred twentieth (120th) day prior to such special meeting and not later than the ninetieth (90th) day prior to such special meeting or, if later, the tenth (10th) day following the day on which public disclosure (as defined in Section 2.4) of the date of such special meeting was first made.

- (iii) In no event shall any adjournment or postponement of an annual meeting or special meeting or the announcement thereof commence a new time period for the giving of a stockholder's notice as described above.
- (iv) In no event may a Nominating Person provide Timely Notice with respect to a greater number of director candidates than are subject to election by shareholders at the applicable meeting. If the Corporation shall, subsequent to such notice, increase the number of directors subject to election at the meeting, such notice as to any additional nominees shall be due on the later of (i) the conclusion of the time period for Timely Notice, (ii) the date set forth in Section 2.5(b)(ii) or (iii) the tenth day following the date of public disclosure (as defined in Section 2.4) of such increase.

- (c) To be in proper form for purposes of this Section 2.5, a stockholder's notice to the Secretary of the Corporation shall set forth:
- (i) As to each Nominating Person (as defined below), the Stockholder Information (as defined in Section 2.4(c)(i), except that for purposes of this Section 2.5, the term "Nominating Person" shall be substituted for the term "Proposing Person" in all places it appears in Section 2.4(c)(i));
- (ii) As to each Nominating Person, any Disclosable Interests (as defined in <u>Section 2.4(c)(ii)</u>, except that for purposes of this <u>Section 2.5</u>, the term "<u>Nominating Person</u>" shall be substituted for the term "<u>Proposing Person</u>" in all places it appears in <u>Section 2.4(c)(ii)</u> and the disclosure with respect to the business to be brought before the meeting in <u>Section 2.4(c)(ii)</u> shall be made with respect to the election of directors at the meeting); and
- (iii) As to each candidate whom a Nominating Person proposes to nominate for election as a director, (A) all information with respect to such candidate for nomination that would be required to be set forth in a stockholder's notice pursuant to this Section 2.5 if such candidate for nomination were a Nominating Person, (B) all information relating to such candidate for nomination that is required to be disclosed in a proxy statement or other filings required to be made in connection with solicitations of proxies for election of directors in a contested election pursuant to Section 14(a) under the Exchange Act (including such candidate's written consent to being named in the proxy statement as a nominee and to serving as a director if elected), (C) a description of any direct or indirect material interest in any material contract or agreement between or among any Nominating Person, on the one hand, and each candidate for nomination or his or her respective associates or any other participants in such solicitation, on the other hand, including, without limitation, all information that would be required to be disclosed pursuant to Item 404 under Regulation S-K if such Nominating Person were the "registrant" for purposes of such rule and the candidate for nomination were a director or executive officer of such registrant and (D) a completed and signed questionnaire, representation and agreement as provided in Section 2.5(f).

For purposes of this <u>Section 2.5</u>, the term "<u>Nominating Person</u>" shall mean (i) the stockholder providing the notice of the nomination proposed to be made at the meeting, (ii) the beneficial owner or beneficial owners, if different, on whose behalf the notice of the nomination proposed to be made at the meeting is made, and (iii) any other participant in such solicitation.

6

- (d) A stockholder providing notice of any nomination proposed to be made at a meeting shall further update and supplement such notice, if necessary, so that the information provided or required to be provided in such notice pursuant to this Section 2.5 shall be true and correct as of the record date for stockholders entitled to vote at the meeting and as of the date that is ten (10) business days prior to the meeting or any adjournment or postponement thereof, and such update and supplement shall be delivered to, or mailed and received by, the Secretary of the Corporation at the principal executive offices of the Corporation not later than five (5) business days after the record date for stockholders entitled to vote at the meeting (in the case of the update and supplement required to be made as of such record date), and not later than eight (8) business days prior to the date for the meeting or, if practicable, any adjournment or postponement thereof (and, if not practicable, on the first practicable date prior to the date to which the meeting has been adjourned or postponed) (in the case of the update and supplement required to be made as of ten (10) business days prior to the meeting or any adjournment or postponement thereof). For the avoidance of doubt, the obligation to update and supplement as set forth in this paragraph or any other Section of these bylaws shall not limit the Corporation's rights with respect to any deficiencies in any notice provided by a stockholder, extend any applicable deadlines hereunder or enable or be deemed to permit a stockholder who has previously submitted notice hereunder to amend or update any nomination or to submit any new nomination.
- (e) In addition to the requirements of this Section 2.5 with respect to any nomination proposed to be made at a meeting, each Nominating Person shall comply with all applicable requirements of the Exchange Act with respect to any such nominations.
- (f) To be eligible to be a candidate for election as a director of the Corporation at an annual or special meeting, a candidate must be nominated in the manner prescribed in Section 2.5 and the candidate for nomination, whether nominated by the Board or by a stockholder of record, must have previously delivered (in accordance with the time period prescribed for delivery in a notice to such candidate given by or on behalf of the Board), to the Secretary of the Corporation at the principal executive offices of the Corporation, (i) a completed written questionnaire (in a form provided by the Corporation) with respect to the background, qualifications, stock ownership and independence of such proposed nominee and (ii) a written representation and agreement (in form provided by the Corporation) that such candidate for nomination (A) is not and, if elected as a director during his or her term of office, will not become a party to (1) any agreement, arrangement or understanding with, and has not given and will not give any commitment or assurance to, any person or entity as to how such proposed nominee, if elected as a director of the Corporation, will act or vote on any issue or question (a "Voting Commitment") or (2) any Voting Commitment that could limit or interfere with such proposed nominee's ability to comply, if elected as a director of the Corporation, with such proposed nominee's fiduciary duties under applicable law, (B) is not, and will not become a party to, any agreement, arrangement or understanding with any person or entity other than the Corporation with respect to any direct or indirect compensation or reimbursement for service as a director that has not been disclosed to the Corporation and (C) if elected as a director of the Corporation, will comply with all applicable corporate governance, conflict of interest, confidentiality, stock ownership and trading and other policies and guidelines of the Corporation shall provide to such candidate for nomination all such policies and guidelines then in eff

- (g) The Board may also require any proposed candidate for nomination as a Director to furnish such other information as may reasonably be requested by the Board in writing prior to the meeting of stockholders at which such candidate's nomination is to be acted upon in order for the Board to determine the eligibility of such candidate for nomination to be an independent director of the Corporation in accordance with the Corporation's corporate governance guidelines.
- (h) A candidate for nomination as a director shall further update and supplement the materials delivered pursuant to this Section 2.5, if necessary, so that the information provided or required to be provided pursuant to this Section 2.5 shall be true and correct as of the record date for stockholders entitled to vote at the meeting and as of the date that is ten (10) business days prior to the meeting or any adjournment or postponement thereof, and such update and supplement shall be delivered to, or mailed and received by, the Secretary of the Corporation at the principal executive offices of the Corporation (or any other office specified by the Corporation in any public announcement) not later than five (5) business days after the record date for stockholders entitled to vote at the meeting (in the case of the update and supplement required to be made as of such record date), and not later than eight (8) business days prior to the date for the meeting or, if practicable, any adjournment or postponement thereof (and, if not practicable, on the first practicable date prior to the date to which the meeting has been adjourned or postponed) (in the case of the update and supplement required to be made as of ten (10) business days prior to the meeting or any adjournment or postponement thereof). For the avoidance of doubt, the obligation to update and supplement as set forth in this paragraph or any other Section of these bylaws shall not limit the Corporation's rights with respect to any deficiencies in any notice provided by a stockholder, extend any applicable deadlines hereunder or enable or be deemed to permit a stockholder who has previously submitted notice hereunder to amend or update any proposal or to submit any new proposal, including by changing or adding nominees, matters, business or resolutions proposed to be brought before a meeting of the stockholders.
- (i) No candidate shall be eligible for nomination as a director of the Corporation unless such candidate for nomination and the Nominating Person seeking to place such candidate's name in nomination has complied with this Section 2.5. The presiding officer at the meeting shall, if the facts warrant, determine that a nomination was not properly made in accordance with Section 2.5, and if he or she should so determine, he or she shall so declare such determination to the meeting, the defective nomination shall be disregarded and any ballots cast for the candidate in question (but in the case of any form of ballot listing other qualified nominees, only the ballots cast for the nominee in question) shall be void and of no force or effect.

(j) Notwithstanding anything in these bylaws to the contrary, no candidate for nomination shall be eligible to be seated as a director of the Corporation unless nominated and elected in accordance with <u>Section 2.5.</u>

2.6 Notice of Stockholders' Meetings.

Unless otherwise provided by law, the Certificate of Incorporation or these bylaws, the notice of any meeting of stockholders shall be sent or otherwise given in accordance with Section 8.1 not less than ten (10) nor more than sixty (60) days before the date of the meeting to each stockholder entitled to vote at such meeting. The notice shall specify the place, if any, date and time of the meeting, the means of remote communication, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such meeting, and, in the case of a special meeting, the purpose or purposes for which the meeting is called.

8

2.7 Quorum.

Unless otherwise provided by law, the Certificate of Incorporation or these bylaws, the holders of a majority in voting power of the stock issued and outstanding and entitled to vote, present in person, or by remote communication, if applicable, or represented by proxy, shall constitute a quorum for the transaction of business at all meetings of the stockholders. A quorum, once established at a meeting, shall not be broken by the withdrawal of enough votes to leave less than a quorum. If, however, a quorum is not present or represented at any meeting of the stockholders, then either (i) the person presiding over the meeting or (ii) a majority in voting power of the stockholders entitled to vote at the meeting, present in person, or by remote communication, if applicable, or represented by proxy, shall have power to recess the meeting or adjourn the meeting from time to time in the manner provided in Section 2.8 until a quorum is present or represented. At any recessed or adjourned meeting at which a quorum is present or represented, any business may be transacted that might have been transacted at the meeting as originally noticed.

2.8 Adjourned Meeting; Notice.

When a meeting is adjourned to another time or place, unless these bylaws otherwise require, notice need not be given of the adjourned meeting if the time, place, if any, thereof, and the means of remote communications, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such adjourned meeting are announced at the meeting at which the adjournment is taken. At any adjourned meeting, the Corporation may transact any business which might have been transacted at the original meeting. If the adjournment is for more than thirty (30) days, a notice of the adjourned meeting shall be given to each stockholder of record entitled to vote at the meeting. If after the adjournment a new record date for determination of stockholders entitled to vote is fixed for the adjourned meeting, the Board shall fix as the record date for determining stockholders entitled to notice of such adjourned meeting the same or an earlier date as that fixed for determination of stockholders entitled to vote at the adjourned meeting, and shall give notice of the adjourned meeting to each stockholder of record entitled to vote at such meeting as of the record date so fixed for notice of such adjourned meeting.

2.9 Conduct of Business.

The date and time of the opening and the closing of the polls for each matter upon which the stockholders will vote at a meeting shall be announced at the meeting by the person presiding over the meeting. The Board may adopt by resolution such rules and regulations for the conduct of the meeting of stockholders as it shall deem appropriate. Except to the extent inconsistent with such rules and regulations as adopted by the Board, the person presiding over any meeting of stockholders shall have the right and authority to convene and (for any or no reason) to recess and/or adjourn the meeting, to prescribe such rules, regulations and procedures (which need not be in writing) and to do all such acts as, in the judgment of such presiding person, are appropriate for the proper conduct of the meeting. Such rules, regulations or procedures, whether adopted by the Board or prescribed by the person presiding over the meeting, may include, without limitation, the following: (i) the establishment of an agenda or order of business for the meeting; (ii) rules and procedures for maintaining order at the meeting and the safety of those present (including, without limitation, rules and procedures for removal of disruptive persons from the meeting); (iii) limitations on attendance at or participation in the meeting to stockholders entitled to vote at the meeting, their duly authorized and constituted proxies or such other persons as the person presiding over the meeting shall determine; (iv) restrictions on entry to the meeting after the time fixed for the commencement thereof; and (v) limitations on the time allotted to questions or comments by participants. The presiding person at any meeting of stockholders, in addition to making any other determinations that may be appropriate to the conduct of the meeting (including, without limitation, determinations with respect to the administration and/or interpretation of any of the rules, regulations or procedures of the meeting, whether adopted by the Board or prescribed by the person presiding over the meeting), shall, if the facts warrant, determine and declare to the meeting that a matter of business was not properly brought before the meeting and if such presiding person should so determine, such presiding person shall so declare to the meeting and any such matter or business not properly brought before the meeting shall not be transacted or considered. Unless and to the extent determined by the Board or the person presiding over the meeting, meetings of stockholders shall not be required to be held in accordance with the rules of parliamentary procedure.

9

2.10 Voting.

Except as may be otherwise provided in the Certificate of Incorporation, these bylaws or the DGCL, each stockholder shall be entitled to one (1) vote for each share of capital stock held by such stockholder.

Except as otherwise provided by the Certificate of Incorporation, at all duly called or convened meetings of stockholders at which a quorum is present, for the election of directors, a plurality of the votes cast shall be sufficient to elect a director. Except as otherwise provided by the Certificate of Incorporation, these bylaws, the rules or regulations of any stock exchange applicable to the Corporation, or applicable law or pursuant to any regulation applicable to the Corporation or its securities, each other matter presented to the stockholders at a duly called or convened meeting at which a quorum is present shall be decided by the affirmative vote of the holders of a majority in voting power of the votes cast (excluding abstentions and broker non-votes) on such matter.

2.11 Record Date for Stockholder Meetings and Other Purposes.

In order that the Corporation may determine the stockholders entitled to notice of or to vote at any meeting of stockholders or any adjournment thereof, the Board may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted by the Board, and which record date shall, unless otherwise required by law, not be more than sixty (60) days nor less than ten (10) days before the date of such meeting. If the Board so fixes a date, such date shall also be the record date for determining the stockholders entitled to vote at such meeting unless the Board determines, at the time it fixes such record date, that a later date on or before the date of the meeting shall be the date for making such determination. If no record date is fixed by the Board, the record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be the close of business on the next day preceding the day on which notice is first given, or, if notice is waived, at the close of business on the day next preceding the day on which the meeting is held. A determination of stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting; provided, however, that the Board may fix a new record date for determination of stockholders entitled to vote at the adjourned meeting; and in such case shall also fix as the record date for stockholders entitled to notice of such adjourned meeting the same or an earlier date as that fixed for determination of

stockholders entitled to vote in accordance herewith at the adjourned meeting.

In order that the Corporation may determine the stockholders entitled to receive payment of any dividend or other distribution or allotment or any rights or the stockholders entitled to exercise any rights in respect of any change, conversion or exchange of capital stock, or for the purposes of any other lawful action, the Board may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted, and which record date shall be not more than sixty (60) days prior to such action. If no record date is fixed, the record date for determining stockholders for any such purpose shall be at the close of business on the day on which the Board adopts the resolution relating thereto.

10

2.12 Proxies.

Each stockholder entitled to vote at a meeting of stockholders may authorize another person or persons to act for such stockholder by proxy authorized by an instrument in writing or by a transmission permitted by law filed in accordance with the procedure established for the meeting, but no such proxy shall be voted or acted upon after three (3) years from its date, unless the proxy provides for a longer period. The revocability of a proxy that states on its face that it is irrevocable shall be governed by the provisions of Section 212 of the DGCL. A proxy may be in the form of an electronic transmission which sets forth or is submitted with information from which it can be determined that the transmission was authorized by the stockholder.

2.13 <u>List of Stockholders Entitled to Vote</u>.

The Corporation shall prepare, at least ten (10) days before every meeting of stockholders, a complete list of the stockholders entitled to vote at the meeting *provided*, *however*, that if the record date for determining the stockholders entitled to vote is less than ten (10) days before the date of the meeting, the list shall reflect the stockholders entitled to vote as of the tenth (10th) day before the meeting date), arranged in alphabetical order, and showing the address of each stockholder and the number of shares registered in the name of each stockholder. The Corporation shall not be required to include electronic mail addresses or other electronic contact information on such list. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting for a period of at least ten (10) days prior to the meeting: (i) on a reasonably accessible electronic network, *provided* that the information required to gain access to such list is provided with the notice of the meeting, or (ii) during ordinary business hours, at the Corporation's principal executive office. In the event that the Corporation determines to make the list available on an electronic network, the Corporation may take reasonable steps to ensure that such information is available only to stockholders of the Corporation. If the meeting is to be held at a place, then the list shall be produced and kept at the time and place of the meeting during the whole time thereof, and may be inspected by any stockholder who is present. If the meeting is to be held solely by means of remote communication, then the list shall also be open to the examination of any stockholder during the whole time of the meeting on a reasonably accessible electronic network, and the information required to access such list shall be provided with the notice of the meeting. Such list shall presumptively determine the identity of the stockholders entitled to vote at the meeting and the number of shares held by each of them. Except as otherwise

2.14 <u>Inspectors of Election.</u>

Before any meeting of stockholders, the Corporation shall appoint an inspector or inspectors of election to act at the meeting or its adjournment and make a written report thereof. The Corporation may designate one or more persons as alternate inspectors to replace any inspector who fails to act. If any person appointed as inspector or any alternate fails to appear or fails or refuses to act, then the person presiding over the meeting shall appoint a person to fill that vacancy.

Such inspectors shall:

- (i) determine the number of shares outstanding and the voting power of each, the number of shares represented at the meeting and the validity of any proxies and ballots;
 - (ii) count all votes or ballots;

11

- (iii) count and tabulate all votes:
- (iv) determine and retain for a reasonable period a record of the disposition of any challenges made to any determination by the inspector(s); and
- (v) certify its or their determination of the number of shares represented at the meeting and its or their count of all votes and ballots.

Each inspector, before entering upon the discharge of the duties of inspector, shall take and sign an oath faithfully to execute the duties of inspection with strict impartiality and according to the best of such inspector's ability. Any report or certificate made by the inspectors of election is *prima facie* evidence of the facts stated therein. The inspectors of election may appoint such persons to assist them in performing their duties as they determine.

2.15 <u>Delivery to the Corporation.</u>

Whenever this Article II requires one or more persons (including a record or beneficial owner of stock) to deliver a document or information to the Corporation or any officer, employee or agent thereof (including any notice, request, questionnaire, revocation, representation or other document or agreement), such document or information shall be in writing exclusively (and not in an electronic transmission) and shall be delivered exclusively by hand (including, without limitation, overnight courier service) or by certified or registered mail, return receipt requested, and the Corporation shall not be required to accept delivery of any document not in such written form or so delivered. For the avoidance of doubt, the Corporation expressly opts out of Section 116 of the DGCL with respect to the delivery of information and documents to the Corporation required by this Article II.

Article III - Directors

3.1 Powers.

Except as otherwise provided by the Certificate of Incorporation or the DGCL, the business and affairs of the Corporation shall be managed by or under the direction of the Board.

3.2 Number of Directors.

Subject to the Certificate of Incorporation and the Stockholder Agreement, the total number of directors constituting the Board shall be determined from time to time by resolution of the Board. No reduction of the authorized number of directors shall have the effect of removing any director before that director's term of office expires.

3.3 <u>Election, Qualification and Term of Office of Directors.</u>

Except as provided in Section 3.4, and subject to the Certificate of Incorporation and the Stockholder Agreement, each director, including a director elected to fill a vacancy or newly created directorship, shall hold office until the expiration of the term of the class, if any, for which such director is elected and until such director's successor is elected and qualified or until such director's earlier death, resignation, disqualification or removal. Directors need not be stockholders. The Certificate of Incorporation or these bylaws may prescribe qualifications for directors. Unless otherwise restricted by the Certificate of Incorporation or bylaws, any director or the entire Board of Directors may be removed, with or without cause, by the holders of a majority of shares entitled to vote at an election of directors.

12

3.4 Resignation and Vacancies.

Any director may resign at any time upon notice given in writing or by electronic transmission to the Corporation. The resignation shall take effect at the time specified therein or upon the happening of an event specified therein, and if no time or event is specified, at the time of its receipt. Subject to the Stockholder Agreement, when one or more directors so resigns and the resignation is effective at a future date or upon the happening of an event to occur on a future date, a majority of the directors then in office, including those who have so resigned, shall have power to fill such vacancy or vacancies, the vote thereon to take effect when such resignation or resignations shall become effective, and each director so chosen shall hold office as provided in Section 3.3.

Subject to the Stockholder Agreement, and unless otherwise provided in the Certificate of Incorporation or these bylaws, vacancies resulting from the death, resignation, disqualification or removal of any director, and newly created directorships resulting from any increase in the authorized number of directors shall be filled only by a majority of the directors then in office, although less than a quorum, or by a sole remaining director.

3.5 <u>Place of Meetings; Meetings by Telephone.</u>

The Board may hold meetings, both regular and special, either within or outside the State of Delaware.

Unless otherwise restricted by the Certificate of Incorporation or these bylaws, members of the Board, or any committee designated by the Board, may participate in a meeting of the Board, or any committee, by means of conference telephone or other communications equipment by means of which all persons participating in the meeting can hear each other, and such participation in a meeting pursuant to this bylaw shall constitute presence in person at the meeting.

3.6 Regular Meetings.

Regular meetings of the Board may be held within or outside the State of Delaware and at such time and at such place as which has been designated by the Board and publicized among all directors, either orally or in writing, by telephone, including a voice-messaging system or other system designed to record and communicate messages, facsimile, telegraph or telex, or by electronic mail or other means of electronic transmission. No further notice shall be required for regular meetings of the Board.

3.7 Special Meetings; Notice.

Special meetings of the Board for any purpose or purposes may be called at any time by the chairperson of the Board, the Chief Executive Officer, the President or the Secretary of the Corporation or a majority of the total number of directors constituting the Board.

Notice of the time and place of special meetings shall be:

(i) delivered personally by hand, by courier or by telephone;

13

- (ii) sent by United States first-class mail, postage prepaid;
- (iii) sent by facsimile or electronic mail; or
- (iv) sent by other means of electronic transmission,

directed to each director at that director's address, telephone number, facsimile number or electronic mail address, or other address for electronic transmission, as the case may be, as shown on the Corporation's records.

If the notice is (i) delivered personally by hand, by courier or by telephone, (ii) sent by facsimile or electronic mail, or (iii) sent by other means of electronic transmission, it shall be delivered or sent at least twenty-four (24) hours before the time of the holding of the meeting. If the notice is sent by U.S. mail, it shall be deposited in the U.S. mail at least four (4) days before the time of the holding of the meeting. The notice need not specify the place of the meeting (if the meeting is to be held at the Corporation's principal executive office) nor the purpose of the meeting.

3.8 Quorum

At all meetings of the Board, unless otherwise provided by the Certificate of Incorporation, a majority of the total number of directors shall constitute a quorum for the transaction of business. The vote of a majority of the directors present at any meeting at which a quorum is present shall be the act of the Board, except as may be otherwise specifically provided by statute, the Certificate of Incorporation or these bylaws. If a quorum is not present at any meeting of the Board, then the directors present thereat may adjourn the meeting from time to time, without notice other than announcement at the meeting, until a quorum is present.

3.9 <u>Board Action without a Meeting.</u>

Unless otherwise restricted by the Certificate of Incorporation or these bylaws, any action required or permitted to be taken at any meeting of the Board, or of any committee thereof, may be taken without a meeting if all members of the Board or committee, as the case may be, consent thereto in writing or by electronic transmission. After an action is taken, the consent or consents relating thereto shall be filed with the minutes of the proceedings of the Board, or the committee thereof, in the same paper or electronic form as the minutes are maintained. Such action by written consent or consent by electronic transmission shall have the same force and effect as a unanimous vote of

3.10 Fees and Compensation of Directors.

Unless otherwise restricted by the Certificate of Incorporation or these bylaws, the Board shall have the authority to fix the compensation, including fees and reimbursement of expenses, of directors for services to the Corporation in any capacity.

14

Article IV - Committees

4.1 <u>Committees of Directors.</u>

The Board may designate one (1) or more committees, each committee to consist, subject to the Stockholder Agreement, of one (1) or more of the directors of the Corporation. The Board may designate one (1) or more directors as alternate members of any committee, who may replace any absent or disqualified member at any meeting of the committee. In the absence or disqualification of a member of a committee, the member or members thereof present at any meeting and not disqualified from voting, whether or not such member or members constitute a quorum, may unanimously appoint another member of the Board to act at the meeting in the place of any such absent or disqualified member. Any such committee, to the extent provided in the resolution of the Board or in these bylaws, shall have and may exercise all the powers and authority of the Board in the management of the business and affairs of the Corporation, and may authorize the seal of the Corporation to be affixed to all papers that may require it; but no such committee shall have the power or authority to (i) approve or adopt, or recommend to the stockholders, any action or matter expressly required by the DGCL to be submitted to stockholders for approval, or (ii) adopt, amend or repeal any bylaw of the Corporation.

4.2 <u>Committee Minutes.</u>

Each committee shall keep regular minutes of its meetings and report the same to the Board when required.

4.3 <u>Meetings and Actions of Committees.</u>

Meetings and actions of committees shall be governed by, and held and taken in accordance with, the provisions of:

- (i) <u>Section 3.5 (place of meetings; meetings by telephone);</u>
- (ii) <u>Section 3.6 (regular meetings);</u>
- (iii) <u>Section 3.7</u> (special meetings; notice);
- (iv) Section 3.9 (board action without a meeting); and
- (v) <u>Section 7.13 (waiver of notice)</u>,

with such changes in the context of those bylaws as are necessary to substitute the committee and its members for the Board and its members*provided*, however, that:

- (i) the time of regular meetings of committees may be determined either by resolution of the Board or by resolution of the committee;
- (ii) special meetings of committees may also be called by resolution of the Board or the chairperson of the applicable committee; and
- (iii) the Board may adopt rules for the governance of any committee to override the provisions that would otherwise apply to the committee pursuant to this <u>Section 4.3</u>, *provided* that such rules do not violate the provisions of the Certificate of Incorporation or applicable law.

15

4.4 <u>Subcommittees.</u>

Unless otherwise provided in the Certificate of Incorporation, these bylaws or the resolutions of the Board designating the committee, a committee may create one (1) or more subcommittees, each subcommittee to consist, subject to the Stockholder Agreement, of one (1) or more members of the committee, and delegate to a subcommittee any or all of the powers and authority of the committee.

Article V - Officers

5.1 Officers.

The officers of the Corporation shall include a Chief Executive Officer, a President and a Secretary. The Corporation may also have, at the discretion of the Board, a Chairperson of the Board, a Vice Chairperson of the Board, a Chief Financial Officer, a Treasurer, one (1) or more Vice Presidents, one (1) or more Assistant Vice Presidents, one (1) or more Assistant Treasurers, one (1) or more Assistant Secretaries, and any such other officers as may be appointed in accordance with the provisions of these bylaws. Any number of offices may be held by the same person. No officer need be a stockholder or director of the Corporation.

5.2 <u>Appointment of Officers</u>.

The Board shall appoint the officers of the Corporation, except such officers as may be appointed in accordance with the provisions of Section 5.3.

5.3 <u>Subordinate Officers</u>.

The Board may appoint, or empower the Chief Executive Officer or, in the absence of a Chief Executive Officer, the President, to appoint, such other officers and agents as the business of the Corporation may require. Each of such officers and agents shall hold office for such period, have such authority, and perform such duties as are provided in these bylaws or as the Board may from time to time determine.

5.4 <u>Removal and Resignation of Officers.</u>

Subject to the rights, if any, of an officer under any contract of employment, any officer may be removed, either with or without cause, by the Board or, except in the case of an officer chosen by the Board, by any officer upon whom such power of removal may be conferred by the Board.

Any officer may resign at any time by giving written notice to the Corporation. Any resignation shall take effect at the date of the receipt of that notice or at any later time specified in that notice. Unless otherwise specified in the notice of resignation, the acceptance of the resignation shall not be necessary to make it effective. Any resignation is without prejudice to the rights, if any, of the Corporation under any contract to which the officer is a party.

5.5 <u>Vacancies in Offices</u>.

Any vacancy occurring in any office of the Corporation shall be filled by the Board or as provided in Section 5.2.

16

5.6 Representation of Shares of Other Corporations.

The Chairperson of the Board, the Chief Executive Officer, or the President of this Corporation, or any other person authorized by the Board, the Chief Executive Officer or the President, is authorized to vote, represent and exercise on behalf of this Corporation all rights incident to any and all shares or voting securities of any other corporation or other person standing in the name of this Corporation. The authority granted herein may be exercised either by such person directly or by any other person authorized to do so by proxy or power of attorney duly executed by such person having the authority.

5.7 <u>Authority and Duties of Officers</u>.

All officers of the Corporation shall respectively have such authority and perform such duties in the management of the business of the Corporation as may be provided herein or designated from time to time by the Board and, to the extent not so provided, as generally pertain to their respective offices, subject to the control of the Board.

5.8 <u>Compensation.</u>

The compensation of the officers of the Corporation for their services as such shall be fixed from time to time by or at the direction of the Board. An officer of the Corporation shall not be prevented from receiving compensation by reason of the fact that he or she is also a director of the Corporation.

Article VI - Records

A stock ledger consisting of one or more records in which the names of all of the Corporation's stockholders of record, the address and number of shares registered in the name of each such stockholder, and all issuances and transfers of stock of the corporation are recorded in accordance with Section 224 of the DGCL shall be administered by or on behalf of the Corporation. Any records administered by or on behalf of the Corporation in the regular course of its business, including its stock ledger, books of account, and minute books, may be kept on, or by means of, or be in the form of, any information storage device, or method, or one or more electronic networks or databases (including one or more distributed electronic networks or databases), provided that the records so kept can be converted into clearly legible paper form within a reasonable time and, with respect to the stock ledger, that the records so kept (i) can be used to prepare the list of stockholders specified in Sections 219 and 220 of the DGCL, (ii) record the information specified in Sections 156, 159, 217(a) and 218 of the DGCL, and (iii) record transfers of stock as governed by Article 8 of the Uniform Commercial Code as adopted in the State of Delaware.

Article VII - General Matters

7.1 Execution of Corporate Contracts and Instruments.

The Board, except as otherwise provided in these bylaws, may authorize any officer or officers, or agent or agents, to enter into any contract or execute any instrument in the name of and on behalf of the Corporation; such authority may be general or confined to specific instances.

17

7.2 Stock Certificates.

The shares of the Corporation shall be represented by certificates or shall be uncertificated. Certificates for the shares of stock, if any, shall be in such form as is consistent with the Certificate of Incorporation and applicable law. Every holder of stock represented by a certificate shall be entitled to have a certificate signed by, or in the name of the Corporation by, any two officers authorized to sign stock certificates representing the number of shares registered in certificate form. The Chairperson or Vice Chairperson of the Board, the Chief Executive Officer, the President, Vice President, the Treasurer, any Assistant Treasurer, the Secretary or any Assistant Secretary of the Corporation shall be specifically authorized to sign stock certificates. Any or all of the signatures on the certificate may be a facsimile. In case any officer, transfer agent or registrar who has signed or whose facsimile signature has been placed upon a certificate has ceased to be such officer, transfer agent or registrar before such certificate is issued, it may be issued by the Corporation with the same effect as if he or she were such officer, transfer agent or registrar at the date of issue.

The Corporation may issue the whole or any part of its shares as partly paid and subject to call for the remainder of the consideration to be paid therefor. Upon the face or back of each stock certificate issued to represent any such partly paid shares, or upon the books and records of the Corporation in the case of uncertificated partly paid shares, the total amount of the consideration to be paid therefor and the amount paid thereon shall be stated. Upon the declaration of any dividend on fully paid shares, the Corporation shall declare a dividend upon partly paid shares of the same class, but only upon the basis of the percentage of the consideration actually paid thereon.

7.3 Special Designation of Certificates.

If the Corporation is authorized to issue more than one class of stock or more than one series of any class, then the powers, the designations, the preferences and the relative, participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights shall be set forth in full or summarized on the face or on the back of the certificate that the Corporation shall issue to represent such class or series of stock (or, in the case of uncertificated shares, set forth in a notice provided pursuant to Section 151 of the DGCL); provided, however, that except as otherwise provided in Section 202 of the DGCL, in lieu of the foregoing requirements, there may be set forth on the face of back of the certificate that the Corporation shall issue to represent such class or series of stock (or, in the case of any uncertificated shares, included in the aforementioned notice) a statement that the Corporation will furnish without charge to each stockholder who so requests the powers, the designations, the preferences and the relative, participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights.

7.4 <u>Lost Certificates</u>.

Except as provided in this Section 7.4, no new certificates for shares shall be issued to replace a previously issued certificate unless the latter is surrendered to the Corporation and cancelled at the same time. The Corporation may issue a new certificate of stock or uncertificated shares in the place of any certificate theretofore issued by it, alleged to have been lost, stolen or destroyed, and the Corporation may require the owner of the lost, stolen or destroyed certificate, or such owner's legal representative, to give the Corporation a bond sufficient to indemnify it against any claim that may be made against it on account of the alleged loss, theft or destruction of any such certificate or the issuance of such new certificate or uncertificated shares.

18

7.5 Shares Without Certificates

The Corporation may adopt a system of issuance, recordation and transfer of its shares of stock by electronic or other means not involving the issuance of certificates, provided the use of such system by the Corporation is permitted in accordance with applicable law.

7.6 <u>Construction; Definitions.</u>

Unless the context requires otherwise, the general provisions, rules of construction and definitions in the DGCL shall govern the construction of these bylaws. Without limiting the generality of this provision, the singular number includes the plural and the plural number includes the singular.

7.7 Dividends

The Board, subject to any restrictions contained in either (i) the DGCL or (ii) the Certificate of Incorporation, may declare and pay dividends upon the shares of its capital stock. Dividends may be paid in cash, in property or in shares of the Corporation's capital stock.

The Board may set apart out of any of the funds of the Corporation available for dividends a reserve or reserves for any proper purpose and may abolish any such reserve. Such purposes shall include but not be limited to equalizing dividends, repairing or maintaining any property of the Corporation, and meeting contingencies.

7.8 Fiscal Year.

The fiscal year of the Corporation shall be fixed by resolution of the Board and may be changed by the Board.

7.9 Seal

The Corporation may adopt a corporate seal, which shall be adopted and which may be altered by the Board. The Corporation may use the corporate seal by causing it or a facsimile thereof to be impressed or affixed or in any other manner reproduced.

7.10 <u>Transfer of Stock</u>.

Shares of the stock of the Corporation shall be transferable in the manner prescribed by law and in these bylaws. Shares of stock of the Corporation shall be transferred on the books of the Corporation only by the holder of record thereof or by such holder's attorney duly authorized in writing, upon surrender to the Corporation of the certificate or certificates representing such shares endorsed by the appropriate person or persons (or by delivery of duly executed instructions with respect to uncertificated shares), with such evidence of the authenticity of such endorsement or execution, transfer, authorization and other matters as the Corporation may reasonably require, and accompanied by all necessary stock transfer stamps. No transfer of stock shall be valid as against the Corporation for any purpose until it shall have been entered in the stock records of the Corporation by an entry showing the names of the persons from and to whom it was transferred.

19

7.11 Stock Transfer Agreements.

The Corporation shall have power to enter into and perform any agreement with any number of stockholders of any one or more classes or series of stock of the Corporation to restrict the transfer of shares of stock of the Corporation of any one or more classes owned by such stockholders in any manner not prohibited by the DGCL.

7.12 <u>Registered Stockholders</u>.

The Corporation:

- (i) shall be entitled to recognize the exclusive right of a person registered on its books as the owner of shares to receive dividends and to vote as such owner; and
- (ii) shall not be bound to recognize any equitable or other claim to or interest in such share or shares on the part of another person, whether or not it shall have express or other notice thereof, except as otherwise provided by the laws of the State of Delaware.

7.13 <u>Waiver of Notice</u>.

Whenever notice is required to be given under any provision of the DGCL, the Certificate of Incorporation or these bylaws, a written waiver, signed by the person entitled to notice, or a waiver by electronic transmission by the person entitled to notice, whether before or after the time of the event for which notice is to be given, shall be deemed equivalent to notice. Attendance of a person at a meeting shall constitute a waiver of notice of such meeting, except when the person attends a meeting for the express purpose of objecting at the beginning of the meeting, to the transaction of any business because the meeting is not lawfully called or convened. Neither the business to be transacted at, nor the purpose of, any regular or special meeting of the stockholders need be specified in any written waiver of notice or any waiver by electronic transmission unless so required by the Certificate of Incorporation or these bylaws.

Article VIII - Notice

Without limiting the manner by which notice otherwise may be given effectively to stockholders, any notice to stockholders given by the Corporation under any provision of the DGCL, the Certificate of Incorporation or these bylaws shall be effective if given by a form of electronic transmission consented to by the stockholder to whom the notice is given. Any such consent shall be revocable by the stockholder by written notice or electronic transmission to the Corporation. Notwithstanding the provisions of this paragraph, the Corporation may give a notice by electronic mail in accordance with the first paragraph of this section without obtaining the consent required by this paragraph.

Any notice given pursuant to the preceding paragraph shall be deemed given:

- (i) if by facsimile telecommunication, when directed to a number at which the stockholder has consented to receive notice;
- (ii) if by a posting on an electronic network together with separate notice to the stockholder of such specific posting, upon the later of (A) such posting and (B) the giving of such separate notice; and
 - (iii) if by any other form of electronic transmission, when directed to the stockholder.

Notwithstanding the foregoing, a notice may not be given by an electronic transmission from and after the time that (1) the Corporation is unable to deliver by such electronic transmission two (2) consecutive notices given by the Corporation and (2) such inability becomes known to the Secretary or an Assistant Secretary of the Corporation or to the transfer agent, or other person responsible for the giving of notice; *provided, however*, that the inadvertent failure to discover such inability shall not invalidate any meeting or other action.

An affidavit of the Secretary or an Assistant Secretary of the Corporation or of the transfer agent or other agent of the Corporation that the notice has been given shall, in the absence of fraud, be prima facie evidence of the facts stated therein.

Article IX - Lock-Up

9.1 <u>Lock-Up</u>. Subject to this <u>Section 9.1</u>, the Lock-up Holders may not Transfer any Lock-up Shares until the first day following the expiration of the Lock-up Period (the "<u>Lock-up</u>").

Notwithstanding the provisions set forth in the paragraph above, the Lock-up Holders or their respective permitted transferees may Transfer the Lock-up Shares during the Lock-up Period (A) to any affiliates or immediate family members of the Corporation's officers or directors; (B) in the case of an individual, by gift to a member of such individual's immediate family or a charitable organization or to a trust, the beneficiary of which is such individual, a member of such individual's immediate family, an affiliate of such individual, or an entity wholly owned by such individual or a member of such individual's immediate family; (C) in the case of an individual, by virtue of laws of descent and distribution upon death of such individual; (D) in the case of an individual, pursuant to a qualified domestic relations order; or (E) in connection with a liquidation, merger, stock exchange, reorganization, tender offer approved by the Board or a duly authorized committee thereof or other similar transactions which results in all of the Corporation's stockholders having the right to exchange their shares of common stock for cash, securities or other property subsequent to the closing date of the Merger; provided, however, that in the case of clauses (A) through (D), these permitted transferees must enter into a written agreement with the Corporation agreeing to be bound by the terms of this Section 9.1.

21

Notwithstanding the other provisions set forth in this <u>Section 9.1</u>, the Board may, in its sole discretion, determine to waive, amend, or repeal the Lock-up obligations set forth herein.

For purposes of this <u>Section 9.1</u>:

- (ii) the term "Lock-up Period" means the period beginning on the closing date of the Merger and ending on the date that is 180 days after the closing date of the Merger;
- (iii) the term "Lock-up Holders" means the holders of common stock of the Corporation issued (a) at the closing of the Merger (as defined below) as Per Share Merger Consideration (as defined in the Merger Agreement) pursuant to the merger (the "Merger") of NHIC Sub Inc., a Delaware corporation, with and into Evolv Technologies, Inc., a Delaware corporation ("Evolv"), in accordance with the Merger Agreement (as defined below) or (b) to directors, officers and employees of the Corporation upon the settlement or exercise of stock options, restricted stock units or other equity awards outstanding as of immediately following the closing of the Merger in respect of awards of Evolv equity interests outstanding immediately prior to the closing of the Merger (such shares, the "Evolv Equity Award Shares");
- (iv) the term "Lock-up Shares" means all shares of common stock held by the Lock-up Holders immediately following the closing of the Merger (other than (a) shares of common stock acquired in the public market or pursuant to a transaction exempt from registration under the Securities Act of 1933, as amended, pursuant to a subscription agreement where the issuance of common stock occurs on or after the closing of the Merger and (b) shares of common stock issued at the closing of the Merger in exchange for shares of common stock of Evolv issued prior to the closing of the Merger upon conversion of those certain convertible promissory notes issued and sold by Evolv pursuant to that certain Convertible Note Purchase Agreement, dated as of January 21, 2021, by and among Evolv and the other parties thereto) and the Evolv Equity Award Shares; provided, that, for clarity, shares of common stock issued in connection with the PIPE Investment (as defined in the Agreement and Plan of Merger, entered into by and among the Corporation, Evolv and NHIC Sub Inc., dated as of March 5, 2021 (as may be amended from time to time, the "Merger Agreement")) shall not constitute Lock-up Shares;
- (v) the term "Transfer" shall mean the (a) sale of, offer to sell, contract or agreement to sell, hypothecate, pledge, grant of any option to purchase or otherwise dispose of or agreement to dispose of, directly or indirectly, or establishment or increase of a put equivalent position or liquidation with respect to or decrease of a call equivalent position within the meaning of Section 16 of the Exchange Act, and the rules and regulations promulgated thereunder with respect to, any security or the economic value thereof, (b) entry into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any security, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise, or (c) public announcement of any intention to effect any transaction specified

Article X - Indemnification

10.1 <u>Indemnification of Directors and Officers</u>.

The Corporation shall indemnify and hold harmless, to the fullest extent permitted by the DGCL as it presently exists or may hereafter be amended, any director or officer of the Corporation who was or is made or is threatened to be made a party or is otherwise involved in any action, suit or proceeding, whether civil, criminal, administrative or investigative (a "Proceeding") by reason of the fact that he or she, or a person for whom he or she is the legal representative, is or was a director or officer of the Corporation or, while serving as a director or officer of the Corporation, is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership (a "covered person"), joint venture, trust, enterprise or non-profit entity, including service with respect to employee benefit plans, against all liability and loss suffered and expenses (including, without limitation, attorneys' fees, judgments, fines, ERISA excise taxes or penalties, and amounts paid in settlement) reasonably incurred by such person in connection with any such Proceeding. Notwithstanding the preceding sentence, except as otherwise provided in Section 10.4, the Corporation shall be required to indemnify a person in connection with a Proceeding initiated by such person only if the Proceeding was authorized in the specific case by the Board.

10.2 <u>Indemnification of Others</u>.

The Corporation shall also have the power to indemnify and hold harmless, to the fullest extent permitted by applicable law as it presently exists or may hereafter be amended, any employee or agent of the Corporation who was or is made or is threatened to be made a party or is otherwise involved in any Proceeding by reason of the fact that he or she, or a person for whom he or she is the legal representative, is or was an employee or agent of the Corporation or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust, enterprise or non-profit entity, including service with respect to employee benefit plans, against all liability and loss suffered and expenses reasonably incurred by such person in connection with any such Proceeding.

10.3 <u>Prepayment of Expenses</u>.

The Corporation shall to the fullest extent not prohibited by applicable law pay the expenses (including, without limitation, attorneys' fees) incurred by any director or officer of the Corporation, and may also pay the expenses incurred by any employee or agent of the Corporation, in defending any Proceeding in advance of its final disposition; provided, however, that such payment of expenses in advance of the final disposition of the Proceeding shall be made only upon receipt of an undertaking by the person to repay all amounts advanced if it should be ultimately determined that the person is not entitled to be indemnified under this Article X or otherwise.

10.4 <u>Determination; Claim.</u>

If a claim for indemnification (following the final disposition of such Proceeding) under this $\underbrace{Article\ X}$ is not paid in full within sixty (60) days, or a claim for advancement of expenses under this $\underbrace{Article\ X}$ is not paid in full within thirty (30) days, after a written claim therefor has been received by the Corporation the claimant may thereafter (but not before) file suit to recover the unpaid amount of such claim and, if successful in whole or in part, shall be entitled to be paid the expense of prosecuting such claim to the fullest extent permitted by law. In any such action the Corporation shall have the burden of proving that the claimant was not entitled to the requested indemnification or payment of expenses under applicable law.

10.5 Non-Exclusivity of Rights.

The rights conferred on any person by this <u>Article X</u> shall not be exclusive of any other rights which such person may have or hereafter acquire under any statute, provision of the Certificate of Incorporation, these bylaws, agreement, vote of stockholders or disinterested directors or otherwise.

23

10.6 <u>Insurance</u>.

The Corporation may purchase and maintain insurance on behalf of any person who is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust enterprise or non-profit entity against any liability asserted against him or her and incurred by him or her in any such capacity, or arising out of his or her status as such, whether or not the Corporation would have the power to indemnify him or her against such liability under the provisions of the DGCL.

10.7 Other Indemnification.

The Corporation's obligation, if any, to indemnify or advance expenses to any person who was or is serving at its request as a director, officer, employee or agent of another corporation, partnership, joint venture, trust, enterprise or non-profit entity shall be reduced by any amount such person actually collects as indemnification or advancement of expenses from such other corporation, partnership, joint venture, trust, enterprise or non-profit enterprise.

10.8 <u>Continuation of Indemnification</u>.

The rights to indemnification and to prepayment of expenses provided by, or granted pursuant to, this Article X shall continue notwithstanding that the person has ceased to be a director or officer of the Corporation and shall inure to the benefit of the estate, heirs, executors, administrators, legatees and distributees of such person.

10.9 Amendment or Repeal; Interpretation.

The provisions of this Article X shall constitute a contract between the Corporation, on the one hand, and, on the other hand, each individual who serves or has served as a director or officer of the Corporation (whether before or after the adoption of these bylaws), in consideration of such person's performance of such services, and pursuant to this Article X the Corporation intends to be legally bound to each such current or former director or officer of the Corporation. With respect to current and former directors and officers of the Corporation, the rights conferred under this Article X are present contractual rights and such rights are fully vested, and shall be deemed to have vested fully, immediately upon adoption of these bylaws. With respect to any directors or officers of the Corporation who commence service following adoption of these bylaws, the rights conferred under this provision shall be present contractual rights and such rights shall fully vest, and be deemed to have vested fully, immediately upon such director or officer commencing service as a director or officer of the Corporation. Any repeal or modification of this Article X shall not adversely affect any right or protection (i) hereunder of any person in respect of any act or omission occurring prior to the time of such repeal or modification or (ii) under any agreement providing for indemnification or advancement of expenses to an officer or director of the Corporation in effect prior to the time of such repeal or modification.

Any reference to an officer of the Corporation in this Article X shall be deemed to refer exclusively to the Chief Executive Officer, the President and the Secretary of the Corporation, or other officer of the Corporation appointed by (x) the Board pursuant to Article V or (y) an officer to whom the Board has delegated the power to appoint officers pursuant to Article V, and any reference to an officer of any other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise shall be deemed to refer exclusively to an officer appointed by the board of directors (or equivalent governing body) of such other entity pursuant to the certificate of incorporation and bylaws (or equivalent organizational documents) of such other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise. The fact that any person who is or was an employee of the Corporation or an employee of any other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise has been given or has used the title of "Vice President" or any other title that could be construed to suggest or imply that such person is or may be an officer of the Corporation or of such other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise shall not result in such person being constituted as, or being deemed to be, an officer of the Corporation or of such other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise for purposes of this Article X.

Article XI - Amendments

The Board is expressly empowered, subject to the Stockholder Agreement, to adopt, amend or repeal the bylaws of the Corporation. The stockholders also shall have power to adopt, amend or repeal the bylaws of the Corporation; *provided, however*, that such action by stockholders shall require, in addition to any other vote required by the Certificate of Incorporation or applicable law, the affirmative vote of the holders of at least two-thirds of the voting power of all the then-outstanding shares of voting stock of the Corporation with the power to vote generally in an election of directors, voting together as a single class. Notwithstanding other provisions set forth in Section 9.1, the Board may, in its sole discretion, determine to waive, amend, or repeal the Lockup obligations set forth in Section 9.1.

Article XII - Forum Selection

Unless the Corporation consents in writing to the selection of an alternative forum, (a) the Court of Chancery (the 'Chancery Court') of the State of Delaware (or, in the event that the Chancery Court does not have jurisdiction, the federal district court for the District of Delaware or other state courts of the State of Delaware) shall, to the fullest extent permitted by law, be the sole and exclusive forum for (i) any derivative Proceeding brought on behalf of the Corporation, (ii) any Proceeding asserting a claim of breach of a fiduciary duty owed by any director, officer or stockholder of the Corporation to the Corporation or to the Corporation's stockholders, (iii) any Proceeding arising pursuant to any provision of the DGCL or the Certificate of Incorporation or these bylaws (as either may be amended from time to time) or (iv) any Proceeding asserting a claim against the Corporation governed by the internal affairs doctrine; and (b) subject to the preceding provisions of this Article XII, to the extent permitted by applicable law, the federal district courts of the United States of America shall be the exclusive forum for the resolution of any complaint asserting a cause of action arising under the Securities Act of 1933, as amended. If any action the subject matter of which is within the scope of clause (a) of the immediately preceding sentence is filed in a court other than the courts in the State of Delaware (a "Foreign Action"), such stockholder shall be deemed to have consented to (x) the personal jurisdiction of the state and federal courts in the State of Delaware in connection with any action brought in any such court to enforce the provisions of clause (a) of the immediately preceding sentence and (y) having service of process made upon such stockholder in any such action by service upon such stockholder's counsel in the Foreign Action as agent for such stockholder. If any action the subject matter of which is within the scope of clause (b) of the immediately preceding sentence is filed in a court other than the federal district courts of the United States of America (a "Foreign Securities Act Action") in the name of any stockholder, such stockholder shall be deemed to have consented to (i) the personal jurisdiction of the federal district courts of the United States of America in connection with any action brought in any such court to enforce clause (b) (a "Securities Act Enforcement Action"), and (ii) having service of process made upon such stockholder in any such Securities Act Enforcement Action by service upon such stockholder's counsel in the Foreign Securities Act Action as agent for such stockholder.

25

Any person or entity purchasing or otherwise acquiring any interest in any security of the Corporation shall be deemed to have notice of and consented to this Article XII. Notwithstanding the foregoing, the provisions of this Article XII shall not apply to suits brought to enforce any liability or duty created by the Securities Exchange Act of 1934, as amended, or any other claim for which the federal courts of the United States have exclusive jurisdiction.

Article XIII - Definitions

As used in these bylaws, unless the context otherwise requires, the following terms shall have the following meanings:

An "electronic transmission" means any form of communication, not directly involving the physical transmission of paper, including the use of, or participation in, one or more electronic networks or databases (including one or more distributed electronic networks or databases), that creates a record that may be retained, retrieved and reviewed by a recipient thereof, and that may be directly reproduced in paper form by such a recipient through an automated process.

An "<u>electronic mail</u>" means an electronic transmission directed to a unique electronic mail address (which electronic mail shall be deemed to include any files attached thereto and any information hyperlinked to a website if such electronic mail includes the contact information of an officer or agent of the Corporation who is available to assist with accessing such files and information).

An "electronic mail address" means a destination, commonly expressed as a string of characters, consisting of a unique user name or mailbox (commonly referred to as the "local part" of the address) and a reference to an internet domain (commonly referred to as the "domain part" of the address), whether or not displayed, to which electronic mail can be sent or delivered.

The term "person" means any individual, general partnership, limited partnership, limited liability company, corporation, trust, business trust, joint stock company, joint venture, unincorporated association, cooperative or association or any other legal entity or organization of whatever nature, and shall include any successor (by merger or otherwise) of such entity.

The undersigned hereby certifies that he is the duly elected, qualified, an	d acting Secretary of Evolv Technologies Holdings, Inc., a Delaware corporation (the
"Corporation"), and that the foregoing bylaws were approved on, 2021,	effective as of, 2021, by the Corporation's board of directors.
IN WITNESS WHEREOF, the undersigned has hereunto set his hand this	day of, 2021.
	[Name]
	[Full Title of Secretary]

AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT

THIS AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT (this "Agreement"), dated as of July 16, 2021, is made and entered into by and among Evolv Technologies Holdings, Inc., a Delaware corporation ("Evolv"), NewHold Investment Corp., a Delaware corporation (the "Company"), NewHold Industrial Technology Holdings LLC, a Delaware limited liability company (the "Sponsor") and each of the other undersigned parties listed as Existing Holders on the signature pages hereto (each such party, together with the Sponsor and any person or entity deemed an "Existing Holder" who hereafter becomes a party to this Agreement pursuant to Section 5.2 of this Agreement, an "Existing Holder" and, collectively, the "Existing Holders") and the undersigned parties listed as New Holders on the signature pages hereto (each such party, together with any person or entity deemed a "New Holder" who hereafter becomes a party to this Agreement pursuant to Section 5.2 of this Agreement, a "New Holder" and collectively the "New Holders"). Existing Holders, collectively with New Holders, are referred to herein as "Holders". Capitalized terms used but not otherwise defined in this Agreement shall have the meaning ascribed to such terms in the Merger Agreement (as defined below).

RECITALS

WHEREAS, on July 30, 2020, the Company, the Sponsor and certain other parties thereto entered into that certain Registration Rights Agreement (the 'Existing Registration Rights Agreement'), pursuant to which the Company granted the Existing Holders certain registration rights with respect to certain securities of the Company;

WHEREAS, the Company, NHIC Sub Inc., a Delaware corporation and wholly-owned subsidiary of the Company ("Merger Sub"), and Evolv Technologies, Inc., a Delaware corporation ("Legacy Evolv") entered that certain agreement and plan of merger, dated as of March 5, 2021, as amended by that certain First Amendment to Agreement and Plan of Merger dated June 5, 2021 (as so amended, the "Merger Agreement"), pursuant to which Merger Sub will merge with and into Evolv (the "Merger") with Legacy Evolv surviving the Merger as a wholly owned subsidiary of NHIC;

WHEREAS, in connection with the closing of the transactions contemplated by the Merger Agreement and subject to the terms and conditions set forth therein, the Existing Holders and New Holders were issued shares of common stock, par value \$0.0001 per share, of the Company ("Common Stock"), in each case, in such amounts and subject to such terms and conditions as set forth in the Merger Agreement;

WHEREAS, pursuant to Section 5.5 of the Existing Registration Rights Agreement, any of the provisions, covenants and conditions set forth therein may be amended or modified upon the written consent of the Company and the Holders of at least a majority of the Registrable Securities at the time in question; and

1

WHEREAS, the Company, Sponsor and the other parties to the Existing Registration Rights Agreement desire to amend and restate the Existing Registration Rights Agreement in order to provide the Existing Holders and the New Holders certain registration rights with respect to certain securities of the Company, as set forth in this Agreement.

NOW, THEREFORE, in consideration of the representations, covenants and agreements contained herein, and certain other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

ARTICLE I DEFINITIONS

- 1.1 <u>Definitions</u>. The terms defined in this <u>Article I</u> shall, for all purposes of this Agreement, have the respective meanings set forth below:
- "2021 Notes" shall mean those certain Convertible Promissory Notes issued by Legacy Evolv to the certain note holders pursuant to that certain Convertible Note Purchase Agreement, dated as of January 21, 2021.
- "Adverse Disclosure" shall mean any public disclosure of material non-public information, which disclosure, in the good faith judgment of the Chief Executive Officer or principal financial officer of the Company, after consultation with counsel to the Company, (i) would be required to be made in any Registration Statement or Prospectus in order for the applicable Registration Statement or Prospectus not to contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements contained therein (in the case of any prospectus and any preliminary prospectus, in the light of the circumstances under which they were made) not misleading, (ii) would not be required to be made at such time if the Registration Statement were not being filed, and (iii) the Company has a bona fide business purpose for not making such information public.
 - "Agreement" shall have the meaning given in the Preamble.
- "Anchor Investor Private Placement Warrants" shall mean the 1,400,000 warrants (or up to 1,520,000 warrants pro rata to the extent that the over-allotment option of the underwriters in connection with the Company's initial public offering was exercised) agreed to be purchased by the Anchor Investors pursuant to that certain Subscription Agreement dated July 8, 2020 by and among the Company, the Sponsor and the Anchor Investors.
 - "Anchor Investors" shall mean certain funds and accounts managed by Magnetar Financial LLC, UBS O'Connor LLC and Mint Tower Capital Management B.V..
 - "Anchor Investor Subscription Agreements" shall have the meaning given the Recitals hereto.
 - "Board" shall mean the Board of Directors of the Company.

- "Business Combination" shall mean any merger, capital stock exchange, asset acquisition, stock purchase, reorganization or other similar business combination with one or more businesses, involving the Company.
 - "Class B Common Stock" shall mean the Company's Class B common stock, par value \$0.0001 per share.
 - "Commission" shall mean the Securities and Exchange Commission.

- "Common Stock" shall have the meaning given in the Recitals hereto.
- "Company" shall have the meaning given in the Preamble.
- "Demand Registration" shall have the meaning given in subsection 2.1.1.
- "Demanding Holder" shall have the meaning given in subsection 2.1.1.
- "Exchange Act" shall mean the Securities Exchange Act of 1934, as it may be amended from time to time.
- "Existing Holders" shall have the meaning given in the Preamble.
- "Form S-1" shall have the meaning given in subsection 2.1.1.
- "Form S-3" shall have the meaning given in subsection 2.3.
- "Founder Shares" shall mean (i) the 4,312,500 shares of the Company's Class B common stock initially purchased by the Sponsor, plus (ii) the 920,000 shares of Class B Common Stock purchased from the Company in July 2020 by the Anchor Investors, less (iii) the 920,000 shares of Class B Common Stock forfeited by the Sponsor in July 2020. The term "Founder Shares" shall be deemed to include the shares of Common Stock issuable upon conversion thereof.
- "Founder Shares Lock-up Period" shall mean, with respect to the Founder Shares, the period ending on the earlier of (A) one year after the completion of the Company's initial Business Combination, (x) if the last reported sale price of the Common Stock equals or exceeds \$12.00 per share (as adjusted for stock splits, stock dividends, reorganizations, recapitalizations and the like) for any 20 trading days within any 30-trading day period commencing at least 150 days after the Company's initial Business Combination or (y) the date on which the Company completes a liquidation, merger, capital stock exchange, reorganization or other similar transaction that results in all of the Company's stockholders having the right to exchange their shares of Common Stock for cash, securities or other property.
 - "Holders" shall have the meaning given in the Preamble.

3

- "Insider Letter" shall mean that certain letter agreement, dated as of July 30, 2020, by and among the Company, the Sponsor, each of the Company's officers and directors and certain other parties thereto.
 - "Maximum Number of Securities" shall have the meaning given in subsection 2.1.4.
- "Misstatement" shall mean an untrue statement of a material fact or an omission to state a material fact required to be stated in a Registration Statement or Prospectus, or necessary to make the statements in a Registration Statement or Prospectus (in the light of the circumstances under which they were made) not misleading.
- "New Holder Lock-up Period" shall mean, with respect to the shares of Common Stock issued to the New Holders at or in connection with Closing ((i) which constitute Per Share Merger Consideration or (ii) which are issued to directors, officers and employees of Evolv upon the settlement or exercise of stock options or other equity awards outstanding as of immediately following the Closing in respect of awards of Company equity interests outstanding as of immediately prior to the Closing and held by the New Holders or their Permitted Transferees, the period ending one hundred eighty (180) days after the date hereof, in each case unless expressly excluded from any lock-up obligations pursuant to the Parent Restated Bylaws).
 - "New Holders" shall have the meaning given in the Preamble.
- "Noteholders' Consent" shall mean that certain Noteholders' Consent dated June 21, 2021, by and among the holders of Company Convertible Notes identified therein, Legacy Evolv and the Company regarding the 2021 Notes.
- "Permitted Transferees" shall mean (i) any person or entity to whom a Holder of Registrable Securities is permitted to transfer such Registrable Securities prior to the expiration of the Founder Shares Lock-up Period, the New Holder Lock-up Period or Private Placement Lock-up Period, as the case may be, under the Insider Letter, the Anchor Investor Subscription Agreements, this Agreement, the Parent Restated Bylaws and any other applicable agreement between such Holder and the Company, and to any transferee thereafter and (b) with respect to a New Holder, any of such New Holder's Affiliates or any fund or investment account managed by such New Holder or the same management company that manages such New Holder; provided, that such transferee to which a transfer is being made pursuant to clause (a) or (b) above, if not a Holder, enters into a written agreement with the Company agreeing to be bound to the restrictions set forth herein.
 - "Piggyback Registration" shall have the meaning given in subsection 2.2.1.
- "Private Placement Lock-up Period" shall mean, with respect to Private Placement Warrants that are held by the initial purchasers of such Private Placement Warrants or their Permitted Transferees, and any of the Common Stock issued or issuable upon the exercise or conversion of the Private Placement Warrants and that are held by the initial purchasers of the Private Placement Warrants or their Permitted Transferees, the period ending 30 days after the completion of the Company's initial Business Combination.

- "Private Placement Warrants" shall mean the Anchor Investor Private Placement Warrants together with the Sponsor Private Placement Warrants.
- "Pro Rata" shall have the meaning given in subsection 2.1.4.
- "Prospectus" shall mean the prospectus included in any Registration Statement, as supplemented by any and all prospectus supplements and as amended by any and all post-effective amendments and including all material incorporated by reference in such prospectus.
- "Registrable Security" shall mean (a) the Founder Shares and the shares of Common Stock issued or issuable upon the conversion of any Founder Shares, (b) the Private Placement Warrants (including any shares of the Common Stock issued or issuable upon the exercise of any such Private Placement Warrant), (c) any outstanding share of the Common Stock or any other equity security (including the shares of Common Stock issued or issuable upon the exercise or conversion of any other equity security) of the Company held by a Holder as of the date of this Agreement (including, for avoidance of doubt, all shares of Common Stock to be issued to the New Holders at the Effective

Time pursuant to the Merger Agreement), (d) Working Capital Warrants (including any shares of the Common Stock issued or issuable upon the exercise of the Working Capital Warrants), and (e) any other equity security of the Company issued or issuable with respect to any of the securities described in the foregoing clauses (a) – (d) by way of a stock dividend or stock split or in connection with a combination of shares, recapitalization, merger, consolidation or reorganization; provided, however, that, as to any particular Registrable Security, such security shall cease to be a Registrable Security when: (A) a Registration Statement with respect to the sale of such security shall have become effective under the Securities Act and such security shall have been sold, transferred, disposed of or exchanged in accordance with such Registration Statement; (B) such security shall have been otherwise transferred, a new certificate for such security not bearing a legend restricting further transfer shall have been delivered by the Company and subsequent public distribution of such security shall not require registration under the Securities Act; (C) such security shall have ceased to be outstanding; (D) such securities may be sold without registration pursuant to Rule 144 promulgated under the Securities Act (or any successor rule promulgated thereafter by the Commission) (but with no volume or other restrictions or limitations); or (E) such securities have been sold to, or through, a broker, dealer or underwriter in a public distribution or other public securities transaction.

"Registration" shall mean a registration effected by preparing and filing a registration statement or similar document in compliance with the requirements of the Securities Act, and the applicable rules and regulations promulgated thereunder, and such registration statement becoming effective.

5

- "Registration Expenses" shall mean the out-of-pocket expenses of a Registration, including, without limitation, the following:
- (A) all registration and filing fees (including fees with respect to filings required to be made with the Financial Industry Regulatory Authority, Inc.) and any securities exchange on which the Common Stock is then listed;
- (B) fees and expenses of compliance with securities or blue sky laws (including reasonable fees and disbursements of counsel for the Underwriters in connection with blue sky qualifications of Registrable Securities);
 - (C) printing, messenger, telephone and delivery expenses;
 - (D) reasonable fees and disbursements of counsel for the Company;
 - (E) reasonable fees and disbursements of all independent registered public accountants of the Company incurred specifically in connection with such Registration; and
- (F) reasonable fees and expenses of one (1) legal counsel selected by the majority-in-interest of the Demanding Holders initiating a Demand Registration to be registered for offer and sale in the applicable Registration.
- "Registration Statement" shall mean any registration statement filed by the Company with the Commission in compliance with the Securities Act and the rules and regulations promulgated thereunder (other than a Registration Statement on Form S-4 or Form S-8, or their successors), which registration statement covers the Registrable Securities pursuant to the provisions of this Agreement, including the Prospectus included in such registration statement, amendments (including post-effective amendments) and supplements to such registration statement, and all exhibits to and all material incorporated by reference in such registration statement.
 - "Requesting Holder" shall have the meaning given in subsection 2.1.1.
 - "Securities Act" shall mean the Securities Act of 1933, as amended from time to time.
 - "Sponsor" shall have the meaning given in the Recitals hereto.
- "Sponsor Private Placement Warrants" shall mean the 3,850,000 warrants (or up to 4,180,000 warrants pro rata to the extent that the over-allotment option of the underwriters in connection with the Company's initial public offering was exercised) agreed to be purchased by the Sponsor at a price of \$1.00 per warrant pursuant to that certain Private Placement Warrant Subscription Agreement dated July 30, 2020 by and between the Company and the Sponsor.
- "Underwriter" shall mean a securities dealer who purchases any Registrable Securities as principal in an Underwritten Offering and not as part of such dealer's market-making activities.
- "Underwritten Registration" or "Underwritten Offering" shall mean a Registration in which securities of the Company are sold to an Underwriter in a firm commitment underwriting for distribution to the public.
- "Working Capital Warrants" shall mean any loan by the Sponsor or an affiliate of the Sponsor or certain of the Company's officers and directors to the Company of funds as the Company may require, of which up to \$100,000 of such loans may be convertible into warrants at a price of \$1.00 per warrant.

6

ARTICLE II REGISTRATIONS

2.1 Demand Registration.

2.1.1 Request for Registration. Subject to the provisions of subsection 2.1.4 and Section 2.4 hereof, at any time and from time to time on or after the date the Company consummates a Business Combination, either (i) the holders of a majority-in-interest of the Registrable Securities held by the Existing Holders, their affiliates and transferees, or (ii) the holders of a majority-in-interest of the Registrable Securities held by the New Holders, their affiliates and transferees (the "Demanding Holders") may make a written demand for Registration of all or part of their Registrable Securities, which written demand shall describe the amount and type of securities to be included in such Registration and the intended method(s) of distribution thereof (such written demand a "Demand Registration"). The Company shall, within ten (10) days of the Company's receipt of the Demand Registration, notify, in writing, all other Holders of Registrable Securities of such demand, and each Holder of Registrable Securities who thereafter wishes to include all or a portion of such Holder's Registrable Securities in a Registration pursuant to a Demand Registration (each such Holder that includes all or a portion of such Holder's Registrable Securities in such Registration, a "Requesting Holder") shall so notify the Company, in writing, within five (5) days after the receipt by the Holder of the notice from the Company. Upon receipt by the Company of any such written notification from a Requesting Holder(s), such Requesting Holder(s) shall be entitled to have their Registrable Securities included in a Registration pursuant to a Demand Registration and the Company shall effect, as soon thereafter as practicable, but not more than forty five (45) days immediately after the Company's receipt of the Demand Registration, the Registration of all Registrable Securities requested by the Demanding Holders and Requesting Holders pursuant to such Demand Registration. Under no circumstances shall the Company be obligated to effect more than an ag

be counted for such purposes unless a Form S-1 or any similar long-form registration statement that may be available at such time ("Form S-1") has become effective and all of the Registrable Securities requested by the Requesting Holders to be registered on behalf of the Requesting Holders in such Form S-1 Registration have been sold, in accordance with Section 3.1 of this Agreement. For the avoidance of doubt, each of (a) the holders of a majority-in-interest of the Registrable Securities held by the Existing Holders, and (b) the holders of a majority-in-interest of the Registrations pursuant to this Section 2.1.1 with respect to their respective Registrable Securities.

2.1.2 Effective Registration. Notwithstanding the provisions of subsection 2.1.1 above or any other part of this Agreement, a Registration pursuant to a Demand Registration shall not count as a Registration unless and until (i) the Registration Statement filed with the Commission with respect to a Registration pursuant to the Demand Registration has been declared effective by the Commission and (ii) the Company has complied with all of its obligations under this Agreement with respect thereto; provided, further, that if, after such Registration Statement has been declared effective, an offering of Registrable Securities in a Registration pursuant to a Demand Registration is interfered with by any stop order or injunction of the Commission, federal or state court or any other governmental agency the Registration Statement with respect to such Registration shall be deemed not to have been declared effective, unless and until, (i) such stop order or injunction is removed, rescinded or otherwise terminated, and (ii) a majority-in-interest of the Demanding Holders initiating such Demand Registration thereafter affirmatively elect to continue with such Registration and accordingly notify the Company in writing, but in no event later than five (5) days, of such election; and provided, further, that the Company shall not be obligated or required to file another Registration Statement until the Registration Statement that has been previously filed with respect to a Registration pursuant to a Demand Registration becomes effective or is subsequently terminated.

7

2.1.3 <u>Underwritten Offering</u>. Subject to the provisions of <u>subsection 2.1.4</u> and <u>Section 2.4</u> hereof, if a majority-in-interest of the Demanding Holders so advise the Company as part of their Demand Registration that the offering of the Registrable Securities pursuant to such Demand Registration shall be in the form of an Underwritten Offering, then the right of such Demanding Holder or Requesting Holder (if any) to include its Registrable Securities in such Registration shall be conditioned upon such Holder's participation in such Underwritten Offering and the inclusion of such Holder's Registrable Securities in such Underwritten Offering to the extent provided herein. All such Holders proposing to distribute their Registrable Securities through an Underwritten Offering under this <u>subsection 2.1.3</u> shall enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering by the majority-in-interest of the Demanding Holders initiating the Demand Registration.

2.1.4 Reduction of Underwritten Offering. If the managing Underwriter or Underwriters in an Underwritten Registration pursuant to a Demand Registration, in good faith, advises the Company, the Demanding Holders and the Requesting Holders (if any) in writing that the dollar amount or number of Registrable Securities that the Demanding Holders and the Requesting Holders (if any) desire to sell, taken together with all other Common Stock or other equity securities that the Company desires to sell and the Common Stock, if any, as to which a Registration has been requested pursuant to separate written contractual piggy-back registration rights held by any other stockholders of the Company who desire to sell, exceeds the maximum dollar amount or maximum number of equity securities that can be sold in the Underwritten Offering without adversely affecting the proposed offering price, the timing, the distribution method, or the probability of success of the Underwritten Offering (such maximum dollar amount or maximum number of such securities, as applicable, the "Maximum Number of Securities"), then the Company shall include in such Underwritten Offering, as follows: (i) first, the Registrable Securities of the Demanding Holders and the Requesting Holders (if any) (pro rata based on the respective number of Registrable Securities that each Demanding Holder and Requesting Holder (if any) has requested be included in such Underwritten Registration and the aggregate number of Registrable Securities that the Demanding Holders and Requesting Holders have requested be included in such Underwritten Registration (such proportion is referred to herein as "Pro Rata")) that can be sold without exceeding the Maximum Number of Securities; (ii) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (i), the Registrable Securities of Holders (Pro Rata, based on the respective number of Registrable Securities that each Holder has so requested) exercising their rights to register their Registrable Securities pursuant to subsection 2.2.1 hereof, without exceeding the Maximum Number of Securities; (iii) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i) and (ii), the Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; and (iv) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i), (ii) and (iii), the Common Stock or other equity securities of other persons or entities that the Company is obligated to register in a Registration pursuant to separate written contractual arrangements with such persons and that can be sold without exceeding the Maximum Number of Securities.

8

2.1.5 <u>Demand Registration Withdrawal</u>. A majority-in-interest of the Demanding Holders initiating a Demand Registration or a majority-in-interest of the Requesting Holders (if any), pursuant to a Registration under <u>subsection 2.1.1</u> shall have the right to withdraw from a Registration pursuant to such Demand Registration for any or no reason whatsoever upon written notification to the Company and the Underwriter or Underwriters (if any) of their intention to withdraw from such Registration prior to the effectiveness of the Registration Statement filed with the Commission with respect to the Registration of their Registrable Securities pursuant to such Demand Registration. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with a Registration pursuant to a Demand Registration prior to its withdrawal under this subsection 2.1.5.

2.2 Piggyback Registration.

2.2.1 Piggyback Rights. If, at any time on or after the date the Company consummates a Business Combination, the Company proposes to file a Registration Statement under the Securities Act with respect to an offering of equity securities, or securities or other obligations exercisable or exchangeable for, or convertible into, equity securities, for its own account or for the account of stockholders of the Company (or by the Company and by the stockholders of the Company including, without limitation, pursuant to Section 2.1 hereof), other than a Registration Statement (i) filed in connection with any employee stock option or other benefit plan, (ii) for an exchange offer or offering of securities solely to the Company's existing stockholders, (iii) for an offering of debt that is convertible into equity securities of the Company or (iv) for a dividend reinvestment plan, then the Company shall give written notice of such proposed filing to all of the Holders of Registrable Securities as soon as practicable but not less than ten (10) days before the anticipated filing date of such Registration Statement, which notice shall (A) describe the amount and type of securities to be included in such offering, the intended method(s) of distribution, and the name of the proposed managing Underwriter or Underwriters, if any, in such offering, and (B) offer to all of the Holders of Registrable Securities the opportunity to register the sale of such number of Registrable Securities as such Holders may request in writing within five (5) days after receipt of such written notice (such Registration as "Piggyback Registration"). The Company shall, in good faith, cause such Registrable Securities to be included in such Piggyback Registration and shall use its best efforts to cause the managing Underwriter or Underwriters of a proposed Underwritten Offering to permit the Registrable Securities of the Company included in such Piggyback Registration and to permit the sale or other disposition of such Registrable Securities in accordan

- 2.2.2 Reduction of Piggyback Registration. If the managing Underwriter or Underwriters in an Underwritten Registration that is to be a Piggyback Registration, in good faith, advises the Company and the Holders of Registrable Securities participating in the Piggyback Registration in writing that the dollar amount or number of the securities that the Company desires to sell, taken together with (i) the Common Stock or other equity securities, if any, as to which Registration has been demanded pursuant to separate written contractual arrangements with persons or entities other than the Holders of Registrable Securities hereunder (ii) the Registrable Securities as to which registration has been requested pursuant to Section 2.2 hereof, and (iii) the Common Stock or other equity securities, if any, as to which Registration has been requested pursuant to separate written contractual piggy-back registration rights of other stockholders of the Company, exceeds the Maximum Number of Securities, then:
- (a) If the Registration is undertaken for the Company's account, the Company shall include in any such Registration (A) first, the Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to subsection 2.2.1 hereof, Pro Rata, which can be sold without exceeding the Maximum Number of Securities; and (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), the Common Stock, if any, as to which Registration has been requested pursuant to written contractual piggy-back registration rights of other stockholders of the Company, which can be sold without exceeding the Maximum Number of Securities;
- (b) If the Registration is pursuant to a request by persons or entities other than the Holders of Registrable Securities, then the Company shall include in any such Registration (A) first, the Common Stock or other equity securities, if any, of such requesting persons or entities, other than the Holders of Registrable Securities, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to subsection 2.2.1, pro rata based on the number of Registrable Securities that each Holder has requested be included in such Underwritten Registration and the aggregate number of Registrable Securities that the Holders have requested to be included in such Underwritten Registration, which can be sold without exceeding the Maximum Number of Securities; (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), the Common Stock or other equity securities has not been reached under the foregoing clauses (A), (B) and (C), the Common Stock or other equity securities for the account of other persons or entities that the Company is obligated to register pursuant to separate written contractual arrangements with such persons or entities, which can be sold without exceeding the Maximum Number of Securities.

10

- 2.2.3 <u>Piggyback Registration Withdrawal</u>. Any Holder of Registrable Securities shall have the right to withdraw from a Piggyback Registration for any or no reason whatsoever upon written notification to the Company and the Underwriter or Underwriters (if any) of his, her or its intention to withdraw from such Piggyback Registration prior to the effectiveness of the Registration Statement filed with the Commission with respect to such Piggyback Registration. The Company (whether on its own good faith determination or as the result of a request for withdrawal by persons pursuant to separate written contractual obligations) may withdraw a Registration Statement filed with the Commission in connection with a Piggyback Registration at any time prior to the effectiveness of such Registration Statement. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with the Piggyback Registration prior to its withdrawal under this <u>subsection</u> 2.2.3.
- 2.2.4 <u>Unlimited Piggyback Registration Rights</u>. For purposes of clarity, any Registration effected pursuant to <u>Section 2.2</u> hereof shall not be counted as a Registration pursuant to a Demand Registration effected under <u>Section 2.1</u> hereof.
- 2.3 Registrations on Form S-3. Any Holder of Registrable Securities may at any time, and from time to time, request in writing that the Company, pursuant to Rule 415 under the Securities Act (or any successor rule promulgated thereafter by the Commission), register the resale of any or all of their Registrable Securities on Form S-3 or any similar short form registration statement that may be available at such time ("Form S-3"); provided, however, that the Company shall not be obligated to effect such request through an Underwritten Offering. Within five (5) days of the Company's receipt of a written request from a Holder or Holders of Registrable Securities for a Registration on Form S-3, the Company shall promptly give written notice of the proposed Registration on Form S-3 to all other Holders of Registrable Securities, and each Holder of Registrable Securities who thereafter wishes to include all or a portion of such Holder's Registrable Securities in such Registration on Form S-3 shall so notify the Company, in writing, within ten (10) days after the receipt by the Holder of the notice from the Company. As soon as practicable thereafter, but not more than twelve (12) days after the Company's initial receipt of such written request for a Registration on Form S-3, the Company shall register all or such portion of such Holder's Registrable Securities as are specified in such written request, together with all or such portion of Registrable Securities of any other Holder or Holders joining in such request as are specified in the written notification given by such Holder or Holders; provided, however, that the Company shall not be obligated to effect any such Registration pursuant to Section 2.3 hereof if (i) a Form S-3 is not available for such offering; or (ii) the Holders of Registrable Securities, together with the Holders of any other equity securities of the Company entitled to inclusion in such Registration, propose to sell the Registrable Securities and such other equity securities (if any) at any aggre
- 2.4 <u>Restrictions on Registration Rights.</u> If (A) during the period starting with the date sixty (60) days prior to the Company's good faith estimate of the date of the filing of, and ending on a date one hundred and twenty (120) days after the effective date of, a Company initiated Registration and provided that the Company has delivered written notice to the Holders prior to receipt of a Demand Registration pursuant to <u>subsection 2.1.1</u> and it continues to actively employ, in good faith, all reasonable efforts to cause the applicable Registration Statement to become effective; (B) the Holders have requested an Underwritten Registration and the Company and the Holders are unable to obtain the commitment of underwriters to firmly underwrite the offer; or (C) in the good faith judgment of the Board such Registration would be seriously detrimental to the Company and the Board concludes as a result that it is essential to defer the filing of such Registration Statement at such time, then in each case the Company shall furnish to such Holders a certificate signed by the Chairman of the Board stating that in the good faith judgment of the Board it would be seriously detrimental to the Company for such Registration Statement to be filed in the near future and that it is therefore essential to defer the filing of such Registration Statement. In such event, the Company shall have the right to defer such filing for a period of not more than thirty (30) days; <u>provided, however</u>, that the Company shall not defer its obligation in this manner more than once in any 12-month period.

11

ARTICLE III COMPANY PROCEDURES

- 3.1 General Procedures. If at any time on or after the date the Company consummates a Business Combination the Company is required to effect the Registration of Registrable Securities, the Company shall use its best efforts to effect such Registration to permit the sale of such Registrable Securities in accordance with the intended plan of distribution thereof, and pursuant thereto the Company shall, as expeditiously as possible:
- 3.1.1 prepare and file with the Commission as soon as practicable a Registration Statement with respect to such Registrable Securities and use its reasonable best efforts to cause such Registration Statement to become effective and remain effective until all Registrable Securities covered by such Registration Statement have been sold;
- 3.1.2 prepare and file with the Commission such amendments and post-effective amendments to the Registration Statement, and such supplements to the Prospectus, as may be requested by any Holder or any Underwriter of Registrable Securities or as may be required by the rules, regulations or instructions applicable to the registration form used by the Company or by the Securities Act or rules and regulations thereunder to keep the Registration Statement effective until all Registrable Securities

covered by such Registration Statement are sold in accordance with the intended plan of distribution set forth in such Registration Statement or supplement to the Prospectus;

- 3.1.3 prior to filing a Registration Statement or prospectus, or any amendment or supplement thereto, furnish without charge to the Underwriters, if any, and each Holder of Registrable Securities included in such Registration, and each such Holder's legal counsel, copies of such Registration Statement as proposed to be filed, each amendment and supplement to such Registration Statement (in each case including all exhibits thereto and documents incorporated by reference therein), the Prospectus included in such Registration Statement (including each preliminary Prospectus), and such other documents as the Underwriters and each Holder of Registrable Securities included in such Registration or the legal counsel for any such Holders may request in order to facilitate the disposition of the Registrable Securities owned by such Holders;
- 3.1.4 prior to any public offering of Registrable Securities, use its best efforts to (i) register or qualify the Registrable Securities covered by the Registration Statement under such securities or "blue sky" laws of such jurisdictions in the United States as any Holder of Registrable Securities included in such Registration Statement (in light of their intended plan of distribution) may request and (ii) take such action necessary to cause such Registrable Securities covered by the Registration Statement to be registered with or approved by such other governmental authorities as may be necessary by virtue of the business and operations of the Company and do any and all other acts and things that may be necessary or advisable to enable the Holders of Registrable Securities included in such Registration Statement to consummate the disposition of such Registrable Securities in such jurisdictions; provided, however, that the Company shall not be required to qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify or take any action to which it would be subject to general service of process or taxation in any such jurisdiction where it is not then otherwise so subject;
- 3.1.5 cause all such Registrable Securities to be listed on each securities exchange or automated quotation system on which similar securities issued by the Company are then listed;
- 3.1.6 provide a transfer agent or warrant agent, as applicable, and registrar for all such Registrable Securities no later than the effective date of such Registration Statement;

12

- 3.1.7 advise each seller of such Registrable Securities, promptly after it shall receive notice or obtain knowledge thereof, of the issuance of any stop order by the Commission suspending the effectiveness of such Registration Statement or the initiation or threatening of any proceeding for such purpose and promptly use its reasonable best efforts to prevent the issuance of any stop order or to obtain its withdrawal if such stop order should be issued;
- 3.1.8 at least five (5) days prior to the filing of any Registration Statement or Prospectus or any amendment or supplement to such Registration Statement or Prospectus, furnish a copy thereof to each seller of such Registrable Securities and its counsel, including, without limitation, providing copies promptly upon receipt of any comment letters received with respect to any such Registration Statement or Prospectus;
- 3.1.9 notify the Holders at any time when a Prospectus relating to such Registration Statement is required to be delivered under the Securities Act, of the happening of any event as a result of which the Prospectus included in such Registration Statement, as then in effect, includes a Misstatement, and then to correct such Misstatement as set forth in Section 3.4 hereof;
- 3.1.10 permit a representative of the Holders (such representative to be selected by a majority of the participating Holders), the Underwriters, if any, and any attorney or accountant retained by such Holders or Underwriter to participate, at each such person's own expense, in the preparation of the Registration Statement, and cause the Company's officers, directors and employees to supply all information reasonably requested by any such representative, Underwriter, attorney or accountant in connection with the Registration; provided, however, that such representatives or Underwriters enter into a confidentiality agreement, in form and substance reasonably satisfactory to the Company, prior to the release or disclosure of any such information; and provided further, the Company may not include the name of any Holder or Underwriter or any information regarding any Holder or Underwriter in any Registration Statement or Prospectus, any amendment or supplement to such Registration Statement or Prospectus, any amendment or underwriter in the prior written consent of such Holder or Underwriter and providing each such Holder or Underwriter a reasonable amount of time to review and comment on such applicable document, which comments the Company shall include unless contrary to applicable law;
- 3.1.11 obtain a "cold comfort" letter from the Company's independent registered public accountants in the event of an Underwritten Registration which the participating Holders may rely on, in customary form and covering such matters of the type customarily covered by "cold comfort" letters as the managing Underwriter may reasonably request, and reasonably satisfactory to a majority-in-interest of the participating Holders;
- 3.1.12 on the date the Registrable Securities are delivered for sale pursuant to such Registration, obtain an opinion, dated such date, of counsel representing the Company for the purposes of such Registration, addressed to the Holders, the placement agent or sales agent, if any, and the Underwriters, if any, covering such legal matters with respect to the Registration in respect of which such opinion is being given as the Holders, placement agent, sales agent, or Underwriter may reasonably request and as are customarily included in such opinions and negative assurance letters, and reasonably satisfactory to a majority in interest of the participating Holders;
- 3.1.13 in the event of any Underwritten Offering, enter into and perform its obligations under an underwriting agreement, in usual and customary form, with the managing Underwriter of such offering;
- 3.1.14 make available to its security holders, as soon as reasonably practicable, an earnings statement covering the period of at least twelve (12) months beginning with the first day of the Company's first full calendar quarter after the effective date of the Registration Statement which satisfies the provisions of Section 11(a) of the Securities Act and Rule 158 thereunder (or any successor rule promulgated thereafter by the Commission);

13

- 3.1.15 if the Registration involves the Registration of Registration of Registration gross proceeds in excess of \$50,000,000, use its reasonable efforts to make available senior executives of the Company to participate in customary "road show" presentations that may be reasonably requested by the Underwriter in any Underwritten Offering; and
- 3.1.16 otherwise, in good faith, cooperate reasonably with, and take such customary actions as may reasonably be requested by the Holders, in connection with such Registration.
- 3.2 <u>Registration Expenses</u>. The Registration Expenses of all Registrations shall be borne by the Company. It is acknowledged by the Holders that the Holders shall bear all incremental selling expenses relating to the sale of Registrable Securities, such as Underwriters' commissions and discounts, brokerage fees, Underwriter marketing costs and, other than as set forth in the definition of "Registration Expenses," all reasonable fees and expenses of any legal counsel representing the Holders.

- 3.3 Requirements for Participation in Underwritten Offerings No person may participate in any Underwritten Offering for equity securities of the Company pursuant to a Registration initiated by the Company hereunder unless such person (i) agrees to sell such person's securities on the basis provided in any underwriting arrangements approved by the Company and (ii) completes and executes all customary questionnaires, powers of attorney, indemnities, lock-up agreements, underwriting agreements and other customary documents as may be reasonably required under the terms of such underwriting arrangements.
- 3.4 Suspension of Sales; Adverse Disclosure. Upon receipt of written notice from the Company that a Registration Statement or Prospectus contains a Misstatement, each of the Holders shall forthwith discontinue disposition of Registrable Securities until it has received copies of a supplemented or amended Prospectus correcting the Misstatement (it being understood that the Company hereby covenants to prepare and file such supplement or amendment as soon as practicable after the time of such notice), or until it is advised in writing by the Company that the use of the Prospectus may be resumed. If the filing, initial effectiveness or continued use of a Registration Statement in respect of any Registration at any time would require the Company to make an Adverse Disclosure or would require the inclusion in such Registration Statement of financial statements that are unavailable to the Company for reasons beyond the Company's control, the Company may, upon giving prompt written notice of such action to the Holders, delay the filing or initial effectiveness of, or suspend use of, such Registration Statement for the shortest period of time, but in no event more than thirty (30) days, determined in good faith by the Company to be necessary for such purpose. In the event the Company exercises its rights under the preceding sentence, the Holders agree to suspend, immediately upon their receipt of the notice referred to above, their use of the Prospectus relating to any Registration in connection with any sale or offer to sell Registrable Securities. The Company shall immediately notify the Holders of the expiration of any period during which it exercised its rights under this Section 3.4 and, upon the expiration of any such period, the Holders shall be entitled to resume the use of any such Prospectus in connection with any sale or offer to sell Registrable Securities.
- 3.5 Reporting Obligations. As long as any Holder shall own Registrable Securities, the Company, at all times while it shall be a reporting company under the Exchange Act, covenants to file timely (or obtain extensions in respect thereof and file within the applicable grace period) all reports required to be filed by the Company after the date hereof pursuant to Sections 13(a) or 15(d) of the Exchange Act and to promptly furnish the Holders with true and complete copies of all such filings. The Company further covenants that it shall take such further action as any Holder may reasonably request, all to the extent required from time to time to enable such Holder to sell shares of the Common Stock held by such Holder without registration under the Securities Act within the limitation of the exemptions provided by Rule 144 promulgated under the Securities Act (or any successor rule promulgated thereafter by the Commission), including providing any legal opinions. Upon the request of any Holder, the Company shall deliver to such Holder a written certification of a duly authorized officer as to whether it has complied with such requirements.

14

ARTICLE IV INDEMNIFICATION AND CONTRIBUTION

4.1 Indemnification.

- 4.1.1 The Company agrees to indemnify, to the extent permitted by law, each Holder of Registrable Securities, its officers and directors and each person who controls such Holder (within the meaning of the Securities Act) against all losses, claims, damages, liabilities and expenses (including attorneys' fees) caused by any untrue or alleged untrue statement of material fact contained in any Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, except insofar as the same are caused by or contained in any information furnished in writing to the Company by such Holder expressly for use therein. The Company shall indemnify the Underwriters, their officers and each person who controls such Underwriters (within the meaning of the Securities Act) to the same extent as provided in the foregoing with respect to the indemnification of the Holder.
- 4.1.2 In connection with any Registration Statement in which a Holder of Registrable Securities is participating, such Holder shall furnish to the Company in writing such information and affidavits as the Company reasonably requests for use in connection with any such Registration Statement or Prospectus and, to the extent permitted by law, shall indemnify the Company, its directors and officers and agents and each person who controls the Company (within the meaning of the Securities Act) against any losses, claims, damages, liabilities and expenses (including without limitation reasonable attorneys' fees) resulting from any untrue statement of material fact contained in the Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, but only to the extent that such untrue statement or omission is contained in any information or affidavit so furnished in writing by such Holder expressly for use therein; The Holders of Registrable Securities shall indemnify the Underwriters, their officers, directors and each person who controls such Underwriters (within the meaning of the Securities Act) to the same extent as provided in the foregoing with respect to indemnification of the Company. For the avoidance of doubt, the obligation to indemnify under this Section 4.01(b) shall be several, not joint and several, among the Holders of Registrable Securities, and the total indemnification liability of a Holder under this Section 4.01(b) shall be in proportion to and limited to the net proceeds received by such Holder from the sale of Registrable Securities pursuant to such Registration Statement.
- 4.1.3 Any person entitled to indemnification herein shall (i) give prompt written notice to the indemnifying party of any claim with respect to which it seeks indemnification (provided that the failure to give prompt notice shall not impair any person's right to indemnification hereunder to the extent such failure has not materially prejudiced the indemnifying party) and (ii) unless in such indemnified party's reasonable judgment a conflict of interest between such indemnified and indemnifying parties may exist with respect to such claim, permit such indemnifying party to assume the defense of such claim with counsel reasonably satisfactory to the indemnified party. If such defense is assumed, the indemnifying party shall not be subject to any liability for any settlement made by the indemnified party without its consent (but such consent shall not be unreasonably withheld). An indemnifying party who is not entitled to, or elects not to, assume the defense of a claim shall not be obligated to pay the fees and expenses of more than one counsel (plus local counsel) for all parties indemnified by such indemnifying party with respect to such claim, unless in the reasonable judgment of any indemnified party a conflict of interest may exist between such indemnified party and any other of such indemnified parties with respect to such claim. No indemnifying party shall, without the consent of the indemnified party, consent to the entry of any judgment or enter into any settlement which cannot be settled in all respects by the payment of money (and such money is so paid by the indemnifying party pursuant to the terms of such settlement) or which settlement does not include as an unconditional term thereof the giving by the claimant or plaintiff to such indemnified party of a release from all liability in respect to such claim or litigation.

15

- 4.1.4 The indemnification provided for under this Agreement shall remain in full force and effect regardless of any investigation made by or on behalf of the indemnified party or any officer, director or controlling person of such indemnified party and shall survive the transfer of securities.
- 4.1.5 If the indemnification provided under Section 4.1 hereof from the indemnifying party is unavailable or insufficient to hold harmless an indemnified party in respect of any losses, claims, damages, liabilities and expenses referred to herein, then the indemnifying party, in lieu of indemnifying the indemnified party, shall contribute to the amount paid or payable by the indemnified party as a result of such losses, claims, damages, liabilities and expenses in such proportion as is appropriate to reflect the relative fault of the indemnifying party and the indemnified party, as well as any other relevant equitable considerations. The relative fault of the indemnifying party and indemnified party shall be determined by reference to, among other things, whether any action in question, including any untrue or alleged untrue statement of a material fact or omission or alleged omission to state a material fact, was made by, or relates to information supplied by, such indemnifying party or indemnified party, and the indemnifying party's and indemnified party's relative intent, knowledge, access to information and opportunity to correct or prevent such action; provided, however, that the liability of any Holder under this subsection 4.1.5 shall be limited to the amount of the net proceeds received by such Holder in such offering giving rise to such liability. The amount paid or payable by a

party as a result of the losses or other liabilities referred to above shall be deemed to include, subject to the limitations set forth in <u>subsections 4.1.1, 4.1.2</u> and <u>4.1.3</u> above, any legal or other fees, charges or expenses reasonably incurred by such party in connection with any investigation or proceeding. The parties hereto agree that it would not be just and equitable if contribution pursuant to this <u>subsection 4.1.5</u> were determined by pro rata allocation or by any other method of allocation, which does not take account of the equitable considerations referred to in this <u>subsection 4.1.5</u>. No person guilty of fraudulent misrepresentation (within the meaning of <u>Section 11(f)</u> of the Securities Act) shall be entitled to contribution pursuant to this <u>subsection 4.1.5</u> from any person who was not guilty of such fraudulent misrepresentation.

ARTICLE V MISCELLANEOUS

5.1 Notices. Any notice or communication under this Agreement must be in writing and given by (i) deposit in the United States mail, addressed to the party to be notified, postage prepaid and registered or certified with return receipt requested, (ii) delivery in person or by courier service providing evidence of delivery, or (iii) transmission by hand delivery, facsimile or electronic mail. Each notice or communication that is mailed, delivered, or transmitted in the manner described above shall be deemed sufficiently given, served, sent, and received, in the case of mailed notices, on the third business day following the date on which it is mailed and, in the case of notices delivered by courier service, hand delivery, facsimile or electronic mail, at such time as it is delivered to the addressee (with the delivery receipt or the affidavit of messenger) or at such time as delivery is refused by the addressee upon presentation. Any notice or communication under this Agreement must be addressed, if to the Company, to: 950 McCarty Street, Building A, Houston, TX 77029, and, if to any Holder, at such Holder's address or contact information as set forth in the Company's books and records. Any party may change its address for notice at any time and from time to time by written notice to the other parties hereto, and such change of address shall become effective thirty (30) days after delivery of such notice as provided in this Section 5.1.

5.2 Assignment; No Third Party Beneficiaries.

5.2.1 This Agreement and the rights, duties and obligations of the Company hereunder may not be assigned or delegated by the Company in whole or in part.

16

- 5.2.2 Prior to the expiration of the Founder Shares Lock-up Period, New Holder Lock-up Period or the Private Placement Lock-up Period, as the case may be, no Holder may assign or delegate such Holder's rights, duties or obligations under this Agreement, in whole or in part, except in connection with a transfer of Registrable Securities by such Holder to a Permitted Transferee but only if such Permitted Transferee agrees to become bound by the transfer restrictions set forth in this Agreement, the Insider Letter, the Anchor Investor Subscription Agreements and other applicable agreements (but only to the extent such Holder is a party thereto). Notwithstanding the foregoing, nothing in this Agreement shall restrict the rights of the New Holders to take any actions in respect of any shares of Common Stock issuable in exchange for the commons stock of Legacy Evolv issued in connection with the conversion of the 2021 Notes.
- 5.2.3 This Agreement and the provisions hereof shall be binding upon and shall inure to the benefit of each of the parties and its successors and the permitted assigns of the Holders, which shall include Permitted Transferees.
- 5.2.4 This Agreement shall not confer any rights or benefits on any persons that are not parties hereto, other than as expressly set forth in this Agreement and Section 5.2 hereof.
- 5.2.5 No assignment by any party hereto of such party's rights, duties and obligations hereunder shall be binding upon or obligate the Company unless and until the Company shall have received (i) written notice of such assignment as provided in <u>Section 5.1</u> hereof and (ii) the written agreement of the assignee, in a form reasonably satisfactory to the Company, to be bound by the terms and provisions of this Agreement (which may be accomplished by an addendum or certificate of joinder to this Agreement). Any transfer or assignment made other than as provided in this <u>Section 5.2</u> shall be null and void.
- 5.3 Counterparts; Electronic Signatures. This Agreement may be executed in counterparts, each of which when so executed shall be deemed to be an original and all of which when taken together shall constitute one and the same instrument. The words "execution," signed," "signature," and words of like import in this Agreement or in any other certificate, agreement or document related to this Agreement shall include images of manually executed signatures transmitted by facsimile or other electronic format (including, without limitation, "pdf", "tif" or "jpg") and other electronic signatures (including, without limitation, DocuSign and AdobeSign). The use of electronic signatures and electronic records (including, without limitation, any contract or other record created, generated, sent, communicated, received, or stored by electronic means) shall be of the same legal effect, validity and enforceability as a manually executed signature or use of a paper-based record-keeping system to the fullest extent permitted by applicable law, including, without limitation, any state law based on the Uniform Electronic Transactions Act or the Uniform Commercial Code.
- 5.4 Governing Law; Venue. NOTWITHSTANDING THE PLACE WHERE THIS AGREEMENT MAY BE EXECUTED BY ANY OF THE PARTIES HERETO, THE PARTIES EXPRESSLY AGREE THAT (I) THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED UNDER THE LAWS OF THE STATE OF NEW YORK AS APPLIED TO AGREEMENTS AMONG NEW YORK RESIDENTS ENTERED INTO AND TO BE PERFORMED ENTIRELY WITHIN NEW YORK, WITHOUT REGARD TO THE CONFLICT OF LAW PROVISIONS OF SUCH JURISDICTION AND (II) THE VENUE FOR ANY ACTION TAKEN WITH RESPECT TO THIS AGREEMENT SHALL BE ANY STATE OR FEDERAL COURT IN NEW YORK COUNTY IN THE STATE OF NEW YORK.

17

- 5.5 Amendments and Modifications. Upon the written consent of the Company and the Holders of at least a majority in interest of the Registrable Securities at the time in question, compliance with any of the provisions, covenants and conditions set forth in this Agreement may be waived, or any of such provisions, covenants or conditions may be amended or modified; provided, however, that notwithstanding the foregoing, any amendment hereto or waiver hereof that adversely affects one Holder, solely in its capacity as a holder of the shares of capital stock of the Company, in a manner that is materially different from the other Holders (in such capacity) shall require the consent of the Holder so affected. No course of dealing between any Holder or the Company and any other party hereto or any failure or delay on the part of a Holder or the Company in exercising any rights or remedies under this Agreement shall operate as a waiver of any rights or remedies of any Holder or the Company. No single or partial exercise of any rights or remedies under this Agreement by a party shall operate as a waiver or preclude the exercise of any other rights or remedies hereunder or thereunder by such party.
- 5.6 Other Registration Rights. Other than registration rights granted under the Subscription Agreements (as defined in the Merger Agreement) or the registration rights of certain New Holders in respect of certain Company Convertible Notes (as defined in the Merger Agreement) granted under the Noteholders' Consent, the Company represents and warrants that no person, other than a Holder of Registrable Securities, has any right to require the Company to register any securities of the Company for sale or to include such securities of the Company in any Registration filed by the Company for the sale of securities for its own account or for the account of any other person. Further, other than registration rights granted under the Subscription Agreements (as defined in the Merger Agreement) or the registration rights of certain New Holders in respect of certain Company Convertible Notes (as defined in the Merger Agreement) granted under the Noteholders' Consent, the Company represents and warrants that this Agreement supersedes any other registration rights agreement or agreement with similar terms and conditions and in the event of a conflict between any such agreement or agreements and this Agreement, the terms of this Agreement shall prevail.

5.7 Term. This Agreement shall terminate upon the earlier of (i) the tenth anniversary of the date of this Agreement or (ii) the date as of which (A) all of the Registrable Securities have been sold pursuant to a Registration Statement (but in no event prior to the applicable period referred to in Section 4(a)(3) of the Securities Act and Rule 174 thereunder (or any successor rule promulgated thereafter by the Commission)) or (B) the Holders of all Registrable Securities are permitted to sell the Registrable Securities under Rule 144 (or any similar provision) under the Securities Act without limitation on the amount of securities sold or the manner of sale and without compliance with the current public reporting requirements set forth under Rule 144(i)(2). The provisions of Section 3.5 and Article IV shall survive any termination.

[Signature Pages Follow]

	18
IN WITNESS WHEREOF, the undersigned have	ve caused this Agreement to be executed as of the date first written above.
	COMPANY:
	NEWHOLD INVESTMENT CORP.
	Ву:
	Name: Kevin Charlton Title: Chief Executive Officer
	EVOLV:
	EVOLV TECHNOLOGIES HOLDINGS, INC.
	Ву:
	Name: Title:
	SPONSOR:
	NEWHOLD INDUSTRIAL TECHNOLOGY HOLDINGS LLC
	By: NewHold Enterprises LLC, its Manager
	Ву:
	Name: Title:
[Signal	ture Page to Amended and Restated Registration Rights Agreement]
[Signal	ture Page to Amended and Restated Registration Rights Agreement]
[Signa.	ture Page to Amended and Restated Registration Rights Agreement]
	ve caused this Agreement to be executed as of the date first written above.
	we caused this Agreement to be executed as of the date first written above. HOLDERS:
	ve caused this Agreement to be executed as of the date first written above.
	ve caused this Agreement to be executed as of the date first written above. HOLDERS: The accounts listed on Schedule A attached hereto, acting by and through MAGNETAR FINANCIAL LLC, their investment manager/general partner/manager By:
	ve caused this Agreement to be executed as of the date first written above. HOLDERS: The accounts listed on Schedule A attached hereto, acting by and through MAGNETAR FINANCIAL LLC, their investment manager/general partner/manager
	ve caused this Agreement to be executed as of the date first written above. HOLDERS: The accounts listed on Schedule A attached hereto, acting by and through MAGNETAR FINANCIAL LLC, their investment manager/general partner/manager By:
	ve caused this Agreement to be executed as of the date first written above. HOLDERS: The accounts listed on Schedule A attached hereto, acting by and through MAGNETAR FINANCIAL LLC, their investment manager/general partner/manager By: Name: Title: The accounts listed on Schedule A attached hereto, acting by and through MINT TOWER CAPITAL MANAGEMENT B.V. By:
	we caused this Agreement to be executed as of the date first written above. HOLDERS: The accounts listed on Schedule A attached hereto, acting by and through MAGNETAR FINANCIAL LLC, their investment manager/general partner/manager By: Name: Title: The accounts listed on Schedule A attached hereto, acting by and through MINT TOWER CAPITAL MANAGEMENT B.V.
	ve caused this Agreement to be executed as of the date first written above. HOLDERS: The accounts listed on Schedule A attached hereto, acting by and through MAGNETAR FINANCIAL LLC, their investment manager/general partner/manager By: Name: Title: The accounts listed on Schedule A attached hereto, acting by and through MINT TOWER CAPITAL MANAGEMENT B.V. By: Name:
	we caused this Agreement to be executed as of the date first written above. HOLDERS: The accounts listed on Schedule A attached hereto, acting by and through MAGNETAR FINANCIAL LLC, their investment manager/general partner/manager By: Name: Title: The accounts listed on Schedule A attached hereto, acting by and through MINT TOWER CAPITAL MANAGEMENT B.V. By: Name: Title: The accounts listed on Schedule A attached hereto, acting by and through UBS O'CONNOR LLC By:
	we caused this Agreement to be executed as of the date first written above. HOLDERS: The accounts listed on Schedule A attached hereto, acting by and through MAGNETAR FINANCIAL LLC, their investment manager/general partner/manager By: Name: Title: The accounts listed on Schedule A attached hereto, acting by and through MINT TOWER CAPITAL MANAGEMENT B.V. By: Name: Title: The accounts listed on Schedule A attached hereto, acting by and through UBS O'CONNOR LLC

	Thomas J. Sullivan
	Kathleen Harris
	Neil Glat
	Brian Mathis
	Marc Saiontz
	Suzy Taherian
	Nick Petruska
	Charles Goldman
	Kevin Charlton
	John Charles Baynes-Reid
	Adam Deutsch
	Susan Quinn
IN WITNESS WHEREOF, the undersigned have caused this Agreement to be	be executed as of the date first written above.
	HOLDERS:
	MOTOROLA SOLUTIONS, INC.
	By: Name: Title:
[Signature Page to Amended and Re	estated Registration Rights Agreement]
IN WITNESS WHEREOF, the undersigned have caused this Agreement to be	be executed as of the date first written above.
	HOLDERS:
	FINBACK EVOLV II, LLC
	By: EIG Evolv, LLC its Manager
	By: Name: George B. Huber
	Title: Manager
[Signature Page to Amended and Re	estated Registration Rights Agreement]

HOLDERS:

IN WITNESS WHEREOF	the undersigned have caused this	Agreement to be executed as of the date first written above.
IN WITHESS WHEREUF.	the undersigned have caused this /	Agreement to be executed as of the date first written above.

HOLDERS:

GENERAL CATALYST GROUP V, L.P.

By: General Catalyst Partners V, L.P.

Its: General Partner

By: General Catalyst GP V, LLC

Its: General Partner

By:

Name: Christopher McCain Title: Chief Legal Officer

GC ENTREPRENEURS FUND V, L.P.

By: General Catalyst Partners V, L.P.

Its: General Partner

By: General Catalyst GP V, LLC

Its: General Partner

By:

Name: Christopher McCain Title: Chief Legal Officer

[Signature Page to Amended and Restated Registration Rights Agreement]

IN WITNESS WHEREOF, the undersigned have caused this Agreement to be executed as of the date first written above.

HOLDERS:

GATES FRONTIER, LLC

By:

Name: Alan Heuberger

Title: Authorized Representative

[Signature Page to Amended and Restated Registration Rights Agreement]

IN WITNESS WHEREOF, the undersigned have caused this Agreement to be executed as of the date first written above.

HOLDERS:

LUX VENTURES III SPECIAL FOUNDERS FUND, L.P.

By: Lux Venture Partners III, LLC, its General Partner

By:

Name: Peter Hebert

Title: Managing Member

LUX VENTURES CAYMAN III, L.P.

By: Lux Venture Partners Cayman III, L.P., its General Partner

By: Lux Venture Cayman III General Partners Limited, its General Partner

By:

Name: Peter Hebert Title: Managing Member

LUX CO-INVEST OPPORTUNITIES, L.P.

By: Lux Co-Invest Partners, LLC, its General Partner

By:

Name: Peter Hebert Title: Managing Member

LUX VENTURES III, L.P.

Lux Venture Partners III, LLC, its General Partner

By:

By:

		Peter Hebert Managing Member
[Signature Page to Amended and Res		
IN WITNESS WHEREOF, the undersigned have caused this Agreement to be	e executed a	s of the date first written above.
	HOLDER	S:
		PLLECTIVE IV, L.P. of itself and as nominee for certain affiliated entities
		ata Collective IV GP, LLC eneral Partner
	By:	
	Name: Z Title: N	Zachary Bogue Managing Member
[Signature Page to Amended and Res	stated Regist	ration Rights Agreement]
IN WITNESS WHEREOF, the undersigned have caused this Agreement to be	e executed a	s of the date first written above.
	HOLDER	S:
	SINEWAY	VE VENTURES FUND I, L.P.
	By:	Vanev Suissa
		Managing Partner
	SINEWAY	VE VENTURES DIRECT 5, L.P.
	By: Name:	Vanev Suissa
		Managing Partner
	SINEWAY	VE VENTURES DIRECT 5A, L.P.
	By:	Zaman Carina
		Yanev Suissa Managing Partner
[Signature Page to Amended and Res	stated Regist	ration Rights Agreement]
IN WITNESS WHEREOF, the undersigned have caused this Agreement to be	e executed a	s of the date first written above.
	HOLDER	S:
	OSAGE U	NIVERSITY PARTNERS I, LP
	By: O	sage University GP, LLC, its General Partner
	By:	Agus Cia agu
		Marc Singer Member
[Signature Page to Amended and Res		

IN WITNESS WHEREOF, the undersigned have caused this	s Agreement to be executed as of the date first written above.
	HOLDERS:
	JAWS EQUITY OWNERS 53, LLC
	Ву:
	Name: Title:
[Signature Page to	Amended and Restated Registration Rights Agreement
IN WITNESS WHEREOF, the undersigned have caused thi	is Agreement to be executed as of the date first written above.
	HOLDERS:
	FAMILY FISCHER INVESTMENTS, L.P.
	By: Name:
	Title:
[Signature Page to	Amended and Restated Registration Rights Agreement]
	_
IN WITNESS WHEREOF, the undersigned have caused this	s Agreement to be executed as of the date first written above.
	HOLDERS:
	NOVEL TMT VENTURES LIMITED
	By: Name:
	Title:
[Signature Page to	Amended and Restated Registration Rights Agreement]
IN WITNESS WHEREOF, the undersigned have caused this	s Agreement to be executed as of the date first written above.
	HOLDERS:
	BFF INVESTMENTS LIMITED
	Ву:
	Name: Title:
Signature Page to	Amended and Restated Registration Rights Agreement]
IN WITNESS WHEREOF, the undersigned have caused this	s Agreement to be executed as of the date first written above.
	HOLDERS:
	STONEHAVEN VENTURES LIMITED
	Ву:
	Name:

Title:

	HOLDERS:
	LIONHILL VENTURES LIMITED
	By:
[Signature Page	to Amended and Restated Registration Rights Agreement]
N WITNESS WHEREOF, the undersigned have caused	this Agreement to be executed as of the date first written above.
	HOLDERS:
	CREATIVE DEVELOPMENT ASSOCIATES LLP
	By:
[Signature Page	to Amended and Restated Registration Rights Agreement]
N WITNESS WHEREOF, the undersigned have caused	this Agreement to be executed as of the date first written above. HOLDERS: CF EVO INVESTMENTS, LLC By: Name: Title: to Amended and Restated Registration Rights Agreement]
N WITNESS WHEREOF, the undersigned have caused	this Agreement to be executed as of the date first written above. HOLDERS: CF EVO INVESTMENTS, LLC By: Name: Title:
N WITNESS WHEREOF, the undersigned have caused	this Agreement to be executed as of the date first written above. HOLDERS: CF EVO INVESTMENTS, LLC By: Name: Title: to Amended and Restated Registration Rights Agreement] this Agreement to be executed as of the date first written above.
N WITNESS WHEREOF, the undersigned have caused	this Agreement to be executed as of the date first written above. HOLDERS: CF EVO INVESTMENTS, LLC By: Name: Title: to Amended and Restated Registration Rights Agreement] this Agreement to be executed as of the date first written above. HOLDERS:

IN WITNESS WHEREOF, the undersigned have caused this Agreement to be executed as of the date first written above.

HOLDERS:

	Ву:
	Name: Title:
	riue.
[Signature Page to Amended and	Restated Registration Rights Agreement]
IN WITNESS WHEREOF, the undersigned have caused this Agreement t	o be executed as of the date first written above.
	HOLDERS:
	CTAN EV DI A CIV A DECIVED INC
	STANLEY BLACK & DECKER, INC.
	By:Name:
	Title:
[Signature Page to Amended and	Restated Registration Rights Agreement]
	,
IN WITNESS WHEREOF, the undersigned have caused this Agreement t	o be executed as of the date first written above.
, ,	
	HOLDERS:
	METAMATERIALS IP COMPANY LLC
	By:
	Name: Title:
	Titic.
[Signature Page to Amended and	Restated Registration Rights Agreement]
IN WITNESS WHEREOF, the undersigned have caused this Agreement t	o be executed as of the date first written above.
	HOLDEDS.
	HOLDERS:
	Patrick Ennis
	Guy Lipworth
	ouy Elphotai
	Jeffrey Porter
	Constantinos G. Zioze
	Level D. Linckin and Elling I. Pader
	Joseph D. Lipchitz and Jillian L. Erdos
	E VENTURES TRUST DATED SEPTEMBER 12, 2019
	By:
	Name: Michael Ellenbogen
	Title: Trustee
	Michael Ellenbogen
	MARY ELLEN NEYLON REVOCABLE TRUST OF 2006
	By:
	Name: Mary Ellen Neylon
	Title: Trustee

CHARLES T. O'NEILL REVOCABLE TRUST OF 2006

	Ву:
	Name: Charles T. O'Neill Title: Trustee
Signature Page to Amended and Re	estated Registration Rights Agreement]
IN WITNESS WHEREOF, the undersigned have caused this Agreement to be	ne executed as of the date first written above
IN WITH 255 WIE IN 2017, the underlyighted have ended this regreement to the	
	HOLDERS:
	John Pistole
	John Fistore
	Ronald Kravit
	Rolling Havil
	Tom Driscoll
	David Smith
	David Brady
	Matt Reynolds
	Jack Hunt
	Nathan Landy
	Alex Mrozack
	Guy Lipworth
	Michael Litchfield
[Signature Page to Amended and Re	estated Registration Rights Agreement]
IN WITNESS WHEREOF, the undersigned have caused this Agreement to be	be executed as of the date first written above.
	HOLDERS:
	Anil Chitkara
	Eben Frankenberg
	Todd McIntyre
	Casey Tegreene
	Peter Conway
	Peter L Onwall

 $[Signature\ Page\ to\ Amended\ and\ Restated\ Registration\ Rights\ Agreement]$

Scott Macintosh

SCHEDULE A

Accounts

[See attached]

EXHIBIT A

Founder Share Transfers

<u>Name</u>	Number of Founder Shares Held Post- Transfer
Thomas J. Sullivan	50,000
Kathleen Harris	40,000
Neil Glat	35,000
Brian Mathis	35,000
Marc Saiontz	32,500
Suzy Taherian	32,500
Nick Petruska	200,000
Charles Goldman	135,000
Kevin Charlton	135,000
Charlie Baynes-Reid	100,000
Adam Deutsch	67,500
Susan Quinn	5,000
TOTAL:	867,500

July 21, 2021

Office of the Chief Accountant Securities and Exchange Commission 100 F Street, NE Washington, D.C. 20549

Ladies and Gentlemen:

We have read Evolv Technologies Holdings, Inc.'s (formerly known as NewHold Investment Corp.) statements included under Item 4.01 of its Form 8-K dated July 16, 2021. We agree with the statements concerning our Firm under Item 4.01, in which we were informed of our dismissal on July 16, 2021, following completion of the Company's quarterly review for the period ended June 30, 2021, which consists only of the accounts of the pre-Business Combination Special Purpose Acquisition Company. We are not in a position to agree or disagree with other statements contained therein.

Very truly yours,

/s/ WithumSmith+Brown, PC

New York, New York

Subsidiary	Jurisdiction of Incorporation
Evolv Technologies, Inc., dba Evolv Technology, Inc.	Delaware
Evolv NewHold Benefit LLC	Delaware

Evolv Technology Publicly Listed on the NASDAQ Under Symbol EVLV

NewHold Investment Corp. and Evolv Technology Complete Business Combination

- · Evolv is the global leader in weapons detection security screening, dedicated to making the world a safer place to work, learn and play by helping to protect innocent people from mass shootings and terrorist attacks. Evolv has screened over 60 million people, second only to the TSA in screening people in the U.S.
- Combined company has a pro forma total enterprise value of approximately \$1.3 billion and total equity value of approximately \$1.7 billion
- · Combined company received approximately \$385 million in gross proceeds, net of redemptions
- · All current Evolv stockholders, including Cascade Asset Management Company, DCVC, Finback Investment Partners, General Catalyst, Lux Capital, Motorola Solutions, SineWave Ventures and STANLEY Ventures, all retained their equity holdings through closing
- · Proceeds will fund rapid growth, expand international markets and help accelerate adjacencies through partnerships and acquisitions
- · Evolv is the first company in the industry to introduce a subscription Security as a Service (SaaS) Model with its customers, including, Uber, Lincoln Center, L.L. Bean, and Six Flags

WALTHAM, MA & NEW YORK, NY – NewHold Investment Corp. (NASDAQ: NHIC, "NewHold"), a publicly-traded special purpose acquisition company, and Evolv Technologies, Inc. ("Evolv Technology" or "Evolv"), the global leader in weapons detection security screening, today announced that they have completed their business combination. The transaction has been unanimously approved by the board of directors of NewHold and was approved at a special meeting of NewHold shareholders on July 15, 2021. The common stock of the combined company will trade on the NASDAQ Stock Market under the symbol "EVLV", its warrants will trade under "EVLVW", and its units will trade under "EVLVU" beginning today, July 19, 2021. Peter George, who has served as Chief Executive Officer of the Company since January 2020, will continue to lead the business.

Evolv Technology - Leader with Unrivaled Technology and Market Presence

Evolv is transforming the physical security industry, as the first weapons detection security screening system in the market, providing continuous detection improvements and a frictionless visitor experience. The Evolv Express® system delivers up to a 70% reduction in cost and is up to ten times faster than traditional metal detectors, resulting in screening up to an unparalleled 3,600 people per hour, per system. The system allows for visitors to pass through screening without breaking stride and continuing the pace of life, improving security at the speed and scale required in this post-pandemic, increasingly volatile world. Built on its Evolv Cortex AITM software platform, Evolv is continually improving the security posture for customers through machine learning and analytical insights on-demand, an industry first.

The world's most iconic venues and companies place their trust in Evolv to protect their employees and visitors, including Lincoln Center for the Performing Arts, L.L. Bean, Six Flags and others. In addition, Evolv signed a significant technology and go-to-market partnership with Motorola Solutions, a global leader in land mobile radio mission-critical communications, video security and analytics, and command center software for enterprise and public safety customers. Evolv expects this partnership to accelerate growth and market presence going forward.

"Evolv's new capital supports our rapid growth in transforming security. The world is opening, visitors require a better experience and piece of mind to gather safely." said Peter George, CEO. "We are energized by our market opportunity and driven to fulfill our mission to make everywhere safer. And we are incredibly thankful to all the customers, employees, investors, and partners who have helped us establish a higher standard for physical security."

"Evolv is a category creating company, permanently changing the physical security market with its weapons detection systems. NewHold is proud to finalize our combination and partnership with Evolv, in order to help accelerate the company's growth and make more people safer, sooner" said Kevin Charlton, CEO of NewHold Investment Corp. "We could not be more excited about the future of the company and expect Evolv Technology to generate value for shareholders for years to come."

In addition to CEO Peter George, Evolv will continue to be led by its highly experienced management team including co-founders Mike Ellenbogen (Chief Innovation Officer) and Anil Chitkara (Head of Corporate Development), Chief Financial Officer Peter Faubert, General Counsel Eric Pyenson, Chief Scientist Alec Rose, Chief Marketing Officer Dana Loof, Chief Revenue Officer A.J. DeRosa and Head of People Liza Knapp.

Peter George will also serve as a Director of the combined company Board alongside Evolv's Co-Founder Mike Ellenbogen. Chairman of the former Evolv Board, Alan Cohen of DCVC, will continue as Chairman of the combined company Board. NewHold representatives, Kevin Charlton, CEO, and Neil Glat, a board member of NewHold will also join the Board of the combined company alongside David Orfao of General Catalyst, Bilal Zuberi of Lux Capital, Mahesh Saptharishi of Motorola Solutions, Merline Saintil, Mark Sullivan and Kimberly Sheehy.

In addition to a strong management team and board, Evolv partners with a group of experienced industry advisors including Juliette Kayyem, former DHS Assistant Secretary, Dan Coats, Former Director of US National Intelligence, John Pistole, Former FBI Deputy Director and TSA Administrator, General Tony Thomas, Former Commander, US SOCOM, David Cohen, Former NYPD Deputy Commissioner, and Jin Kim, Former FBI Active Shooter Expert.

Evolv Technology, which has over 100 employees worldwide, will maintain its global headquarters in Waltham, Massachusetts, in addition to offices in Singapore and London.

Transaction Overview

At the closing of the transaction, NewHold Investment Corp.'s name changed to Evolv Technologies Holdings Inc. (NASDAQ: EVLV).

The combined entity has a pro forma enterprise value of approximately \$1.3 billion. The upsized and oversubscribed \$300 million PIPE, which closed concurrently with the closing of the business combination, included investors such as Motorola Solutions, Inc., Magnetar Capital, Eldridge, Senator Investment Group and UBS O'Connor, in addition to investments from star athletes, franchise owners and managers. Evolv's existing stockholders own approximately 75% of the fully diluted shares of common stock of the

combined company immediately following the closing of the business combination.

The combined company received approximately \$385 million in gross proceeds from the trust account and PIPE investors, net of redemptions from NewHold's existing public shareholders.

A more detailed description of the transaction terms will be included in a current report on Form 8-K to be filed by Evolv with the U.S. Securities and Exchange Commission ("SEC") at www.sec.gov.

Advisors

Cowen acted as sole financial and capital markets advisor to Evolv and as sole placement agent to NewHold in connection with the \$300 million PIPE. Stifel acted as sole financial and capital markets advisor to NewHold. Latham & Watkins LLP acted as legal counsel to Evolv and Loeb & Loeb LLP acted as legal counsel to NewHold. Tarplin Consulting acted as Investor Relations advisor and partner to NewHold.

About Evolv Technology

Evolv Technology is the world's leading provider of weapons detection security screening systems that enhance safety without sacrificing the visitor, student and employee experience. Built on top of its Evolv Cortex AITM software platform, the company provides an array of AI touchless screening technologies for weapons detection, identity verification and health-related threats. Led by a team of security industry leaders with a track record for delivering first-to-market products, Evolv's investors include Cascade Asset Management Company, Florida Governor Jeb Bush's firm, Finback Investment Partners, DCVC, General Catalyst Partners, Lux Capital, SineWave Ventures, Motorola Solutions and STANLEY Ventures. The company's strategic channel partners include Motorola Solutions, STANLEY Security and Johnson Controls. Evolv Express® has earned industry accolades such as the 2020 Edison AwardsTM, two Campus Safety 2020 BEST Awards, Campus Security & Life Safety magazine's Secure Campus 2020 Awards and Best Places to Work by Inc. Magazine and Built in Boston.

Evolv Technology, Evolv Express®, Evolv InsightsTM, and Evolv Cortex AITM are registered trademarks or trademarks of Evolv Technologies, Inc. in the United States and other jurisdictions.

For more information, visit https://evolvtechnology.com.

About NewHold Investment Corp.

NewHold Investment Corp. is a blank check company formed in 2020 for the purpose of effecting a merger, capital stock exchange, asset acquisition, stock purchase, reorganization or similar business combination with one or more businesses. While the Company may pursue a business combination target in any business or industry, it focuses on identifying businesses in the industrial technology sector. NewHold Investment Corp. is led by an experienced management team with Kevin Charlton as CEO, Charlie Baynes-Reid as COO and Adam Deutsch as CFO, alongside a seasoned Board of Directors. For more information visit https://nhicspac.com.

IMPORTANT LEGAL INFORMATION

Forward-Looking Statements

This document contains certain forward-looking statements within the meaning of the federal securities laws, including statements regarding expected benefits of the transaction to Evolv and adoption of Evolv's technology and products. These forward-looking statements generally are identified by the words "believe," "project," "expect," "anticipate," "estimate," "intend," "strategy," "future," "opportunity," "plan," "may," "should," "will," "would," "will be," "will continue," "will likely result," and similar expressions. Forward-looking statements are predictions, projections and other statements about future events that are based on current expectations and assumptions and, as a result, are subject to risks and uncertainties. Many factors could cause actual future events to differ materially from the forward-looking statements in this document, including Evolv's ability to implement business plans after the transaction and changes and developments in the competitive security screening, weapons detection and related industries. The foregoing list of factors is not exhaustive. You should carefully consider the foregoing factors and the other risks and uncertainties described in the "Risk Factors" section of NewHold's definitive proxy statement/prospectus filed with the SEC on June 28, 2021, and other documents filed by the company from time to time with the SEC. These filings identify and address other important risks and uncertainties that could cause actual events and results to differ materially from those contained in the forward-looking statements. Forward-looking statements speak only as of the date they are made. Readers are cautioned not to put undue reliance on forward-looking statements, and the company assumes no obligation and do not intend to update or revise these forward-looking statements, whether as a result of new information, future events, or otherwise. The company does not give any assurance that it will achieve its expectations.

For Evolv Technology:

Investor Contact: Michael Bowen and Ryan Gardella EvolvIR@icrinc.com Media Contact: Jed Hamilton EvolvPR@icrinc.com

For NewHold Investment Corp.:

Investor & Media Contact: Amanda Tarplin amanda@tarplinconsulting.com

UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION

Capitalized terms used but not defined in this Exhibit 99.2 shall have the meanings ascribed to them in the Current Report on Form 8-K (the 'Form 8-K'') filed with the Securities and Exchange Commission (the "SEC") on July 21, 2021 and, if not defined in the Form 8-K, the proxy statement/prospectus.

The following unaudited pro forma condensed combined financial information is provided to aid you in your analysis of the financial aspects of the Business Combination and the PIPE Investment. The Business Combination will be accounted for as a reverse recapitalization, pursuant to which the Business Combination will be treated as the equivalent of Evolv issuing stock for the net assets of NewHold, accompanied by a recapitalization. The net assets of NewHold will be stated at historical cost, with no goodwill or other intangible assets recorded.

The following unaudited pro forma condensed combined balance sheet as of March 31, 2021 combines the unaudited historical condensed balance sheet of NewHold as of March 31, 2021 with the unaudited historical consolidated balance sheet of Evolv as of March 31, 2021, giving effect to the Business Combination, the PIPE Investment as if they had been consummated on March 31, 2021.

The following unaudited pro forma condensed combined statement of operations for the three months ended March 31, 2021 combines the unaudited historical condensed statement of operations of NewHold for the three months ended March 31, 2021 and the unaudited historical consolidated statement of operations of Evolv for such period, giving effect to the Business Combination and the PIPE Investment as if they had been consummated on January 1, 2020, the beginning of the earliest period presented. The following unaudited pro forma condensed combined statement of operations for the year ended December 31, 2020 combines the audited historical statement of operations of NewHold (as restated) for the period January 24, 2020 (inception) through December 31, 2020 and the audited historical consolidated statement of operations of Evolv for the year ended December 31, 2020, giving effect to the Business Combination and the PIPE Investment as if they had been consummated on January 1, 2020, the beginning of the earliest period presented.

The unaudited pro forma condensed combined financial information has been derived from and should be read in conjunction with:

- the accompanying notes to the unaudited pro forma condensed combined financial information;
- the unaudited historical condensed balance sheet of NewHold as of March 31, 2021 and the audited historical statement of operations (as restated) for the period from January 24, 2020 (inception) through December 31, 2020 and the unaudited historical condensed statement of operations for the three months ended March 31, 2021, and the related notes thereto, included in the proxy statement/prospectus and incorporated by reference;
- the unaudited historical consolidated financial statements of Evolv as of and for the three months ended March 31, 2021 and the audited historical consolidated financial statements for the year ended December 31, 2020, and the related notes thereto, included in the proxy statement/prospectus and incorporated by reference; and
- the sections entitled "The Merger Agreement," "Management's Discussion and Analysis of Financial Condition and Results of Operations of NewHold," "Management's Discussion and Analysis of Financial Condition and Results of Operation of Evolv," and other financial information relating to NewHold and Evolv included in the proxy statement/prospectus and incorporated by reference.

The unaudited pro forma condensed combined financial information below reflects stockholders' approval of the Business Combination on July 15, 2021 and 8,755,987 shares of outstanding NewHold Class A common stock that were redeemed, resulting in an aggregate payment of \$87.6 million out of the trust account, at a redemption price of \$10.00 per share.

The unaudited pro forma condensed combined financial information is for illustrative purposes only and is not necessarily indicative of what the actual results of operations and financial position would have been had the Business Combination, and the PIPE Investment taken place on the dates indicated, nor are they indicative of the future consolidated results of operations or financial position of the combined company. Further, it should be noted the Private Placement Warrants are currently classified as fair value liability instruments. The final accounting of the transaction, including the earnout, Founder Shares and private and public warrants, will be finalized by Evolv and reported on in the first reporting period following the consummation of the Business Combination.

2

UNAUDITED PRO FORMA CONDENSED COMBINED BALANCE SHEET AS OF MARCH 31, 2021

(in thousands, except share and per share data)

	Historical	Historical					
	(A) Evolv	(B) NHIC	_	Transaction Accounting Adjustments	PIPE Transaction Adjustments	_	Pro Forma Combined
Assets							
Current assets:							
Cash and cash equivalents	\$ 22,122	\$ 579	\$	(12,257) 5(b) \$	280,1145	(n) \$	366,614
				166,560 5(d)			
				(2,940) 5(k)			
				(87,564) 5(m)			
Accounts receivable	2,338	-		-	-		2,338
Inventory	3,175	-		-	-		3,175
Current portion of commission asset	690	-		-	-		690
Prepaid expenses and other current assets	5,007	193		-	-		5,200
Total current assets	33,332	772		63,799	280,114		378,017
Commission asset, noncurrent	1,993	-		-	-		1,993
Property and equipment, net	11,386	-		-	-		11,386
Cash and investments held in trust account	_	172,598		(6,038) 5(a)	-		-
		ĺ		(166,560) 5(d)			
Other long-term assets	119	-		-	-		119
Total assets	\$ 46,830	\$ 173,370	\$	(108,799) \$	280,114	\$	391,515
Liabilities, Convertible Preferred Stock and Stockholders'							

Liabilities, Convertible Preferred Stock and Stockholders

Equity (Deficit)

Current liabilities:					
Accounts payable	\$ 5,633	\$ 1,426	\$ -	\$ -	\$ 7,059
Accrued expenses and other current liabilities	5,540	-	-	-	5,540
Accrued income and franchise taxes	-	50	-	-	50
Current portion of deferred revenue	3,110	-	-	-	3,110
Current portion of long-term debt	24,515	-	(24,515) 5(f)	-	-
Total current liabilities	38,798	1,476	(24,515)	_	15,759
Deferred revenue, noncurrent	466	-	-	-	466
Derivative liability	9,431	-	(9,431) 5(f)	-	-
Earn-out liability	-	-	92,880 5(j)	-	92,880
Common stock warrant liability	737	-	(737) 5(i)	-	-
Warrant liability	-	24,165	(14,317) 5(c)	-	9,848
Deferred underwriting compensation	-	6,038	(6,038) 5(a)	-	-
Long-term debt, noncurrent	18,424	-	(4,389) 5(f)	-	14,035
Total liabilities	 67,856	31,679	33,453	-	132,988

UNAUDITED PRO FORMA CONDENSED COMBINED BALANCE SHEET AS OF MARCH 31, 2021 — (Continued)

(in thousands, except share and per share data)

		Historical (A) Evolv		istorical (B) NHIC	Transaction Accounting Adjustments		PIPE Transaction Adjustments		Pro Forma Combined
NewHold common stock subject to possibl redemption	e	_		136,691	(136,691)	5(m)	_		_
Evolv convertible preferred stock; (Series A,A-1, B, and B-1) Commitments and contingencies		75,877		-	(75,877)	• •	-		_
Stockholders' equity (deficit):									
NewHold convertible preferred stock NewHold Class A common stock		-		-	100	5(1)	30	5(n)	139
rewriting Class A common stock		_		_	-	5(d)	30	3(11)	137
					9	5(m)			
NewHold Class B common stock		-		-	-	, ,	-		-
Evolv common stock		30		-	217	5(e)	-		-
					4	5(f)			
					3	5(g)			
					(258)	5(i)			
Additional paid-in capital		10,703		20,994	(12,257)		280,084	5(n)	371,779
Tudinonal para in capital		10,702		20,>>.	14,317		200,00.	C (11)	3,1,,,,
						5(d)			
					75,660				
					44,086				
						5(g)			
					737				
					(92,880)	5(i)			
					(2,940)				
					(15,836)				
					49,118				
Accumulated deficit		(107,636))	(15,994)	(5,755)	5(f)			(113,391)
					15,994	5(l)			
	Stockholders' deficit	(96,903)		5,000	70,316		280,114		258,527
	Total liabilities, convertible preferred stock and stockholders' deficit	\$ 46,830	\$	173,370	\$ (108,799)		\$ 280,114		\$ 391,515

See accompanying notes to the unaudited pro forma condensed combined financial information.

_

UNAUDITED PRO FORMA CONDENSED COMBINED STATEMENT OF OPERATIONS FOR THE THREE MONTHS ENDED MARCH 31, 2021 (in thousands, except share and per share amounts)

	Historical					
	(A) Evolv	(B) NHIO	C	Pro Forma Adjustment		o Forma ombined
Revenue:						_
Product revenue	\$ 2,502	\$	-	\$	-	\$ 2,502
Subscription revenue	1,300		-		-	1,300
Service revenue	197		-		-	197
Total revenue	3,999		_		_	3,999

Cost of revenues:					
Cost of product revenue	2,229	-	-		2,229
Cost of subscription revenue	595	-	-		595
Cost of service revenue	127	-	-		127
Total cost of revenue	 2,951	-	-		2,951
Gross profit	1,048	-			1,048
Operating expenses:					
Research and development	3,612	-	-		3,612
Sales and marketing	3,684	-	-		3,684
General and administrative	 2,899	1,455			4,354
Total operating expenses	 10,195	1,455			11,650
Loss from operations	(9,147)	(1,455)	-		(10,602)
Other income (expense):					
Interest expense	(2,447)	-	635	6(a)	(1,812)
Change in fair value of derivative liability	(1,425)	-	-		(1,425)
Change in fair value of warrants	(736)	(2,646)	/	6(c)	(1,180)
			736	. ,	
Interest income	 	 19	(19) 6(b)	-
Total other income (expense)	 (4,608)	(2,627)	2,818		(4,417)
Net loss before provision for income taxes	(13,755)	(4,082)	2,818		(15,019)
Provision for income taxes	 <u>-</u>	=			-
Net loss attributable to common stockholders – basic and diluted	\$ (13,755)	\$ (4,082)	\$ 2,818	\$	(15,019)
Net loss per share attributable to common stockholders – basic and diluted	\$ (0.52)	\$ (0.95)		\$	(0.10)
Weighted average common shares outstanding – basic and diluted	26,433,739	4,312,500	138,845,104		143,157,604
	 		· · · · · · · · · · · · · · · · · · ·		

See accompanying notes to the unaudited pro forma condensed combined financial information.

5

UNAUDITED PRO FORMA CONDENSED COMBINED STATEMENT OF OPERATIONS FOR THE YEAR ENDED DECEMBER 31, 2020

(in thousands, except share and per share amounts)

	Historical						
		(A) Evolv		(B) NHIC	Pro Forma Adjustments		Pro Forma Combined
Revenue:							
Product revenue	\$	1,279	\$	-	\$ -		\$ 1,279
Subscription revenue		2,637		-	-		2,637
Service revenue		869			 		 869
Total revenue		4,785		-	-		4,785
Cost of revenues:							
Cost of product revenue		1,177		-	-		1,177
Cost of subscription revenue		1,824		-	-		1,824
Cost of service revenue		495		<u>-</u>			495
Total cost of revenue		3,496			_		3,496
Gross profit		1,289		-	-		1,289
Operating expenses:							
Research and development		15,710		-	-		15,710
Sales and marketing		7,365		-	-		7,365
General and administrative		5,110		1,030	 <u>-</u>		 6,140
Total operating expenses		28,185		1,030	-		29,215
Loss from operations		(26,896)		(1,030)	-		(27,926)
Other income (expense):							
Interest expense		(430)		-	40	7(a)	(390)
Loss on extinguishment of debt		(66)		-	(5,566)	7(b)	(5,632)
Cost of issuance of public and private warrants		-		(390)	-		(390)
Change in fair value of warrants		-		(10,571)	6,267	7(d)	(4,304)
Interest income		<u>-</u>		79	(79)	7(c)	-
Total other income (expense)		(496)		(10,882)	 662		 (10,716)
Net loss attributable to common stockholders – basic and diluted	\$	(27,392)	\$	(11,912)	\$ 662		\$ (38,642)
Net loss per share attributable to common stockholders – basic and diluted	\$	(1.16)	\$	(2.97)	 		\$ (0.27)
Weighted average common shares outstanding – basic and diluted		23,625,483		4,312,500	138,845,104		143,157,604

See accompanying notes to the unaudited pro forma condensed combined financial information.

NOTES TO UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION

1. Description of the Transactions

The Business Combination

The board of directors of NewHold Investment Corp., a Delaware corporation ("NewHold"), unanimously approved the Agreement and Plan of Merger (the "Merger Agreement") by and among NewHold, NHIC Sub Inc., a Delaware corporation and a wholly owned subsidiary of NewHold ("Merger Sub"), and Evolv Technologies, Inc. dba Evolv Technology, Inc., a Delaware corporation ("Evolv"). Pursuant to the terms of the Merger Agreement, a business combination between NewHold and Evolv will be effected through the merger of Merger Sub with and into Evolv, with Evolv surviving the merger as a wholly owned subsidiary of NewHold (the "Merger"). We refer to the Merger and the other transactions contemplated by the Merger Agreement as the "Business Combination" and the "PIPE Transaction".

Treatment of Evolv Securities

Preferred Stock. Immediately prior to the effective time of the Merger (the "Effective Time") and subject to the consent of the holders of Evolv's preferred stock, par value \$0.001 per share (the "Evolv Preferred Stock"), each issued and outstanding share of Evolv Preferred Stock converted into shares of the common stock, par value \$0.001 per share, of Evolv (the "Evolv Common Stock") at the then-applicable conversion rates.

Convertible Notes. Immediately prior to the Effective Time, each issued and outstanding convertible promissory note of Evolv (the "Evolv Convertible Notes") automatically converted into shares of Evolv Common Stock in accordance with the then-applicable conversion rates. We have assumed for pro forma purposes the conversion of all outstanding Evolv Convertible Notes, including the Evolv Convertible Notes issued in January 2021, and the extinguishment of the related derivative liability.

Warrants. With the exception of a warrant to purchase 6,756,653 shares of Evolv Common Stock (the "Finback Warrant"), immediately prior to the Effective Time, Evolv shall cause each outstanding warrant to purchase shares of Evolv capital stock to be exercised in full on a cash or cashless basis or terminated without exercise. With respect to the Finback Warrant, the portion that is vested immediately prior to the Effective Time was either exercised in full on a cash or cashless basis or terminated as of the Effective Time, while the portion that is unvested as of immediately prior to the Effective Time automatically converted into a warrant to purchase shares of the Class A common stock, par value \$0.0001 per share, of NewHold (the "NHIC common stock"), proportionately adjusted for the Exchange Ratio (as defined below). We have assumed for pro forma purposes that 348,598 of the Finback warrants are vested as of the Effective Time. All Evolv warrants that are converted into shares of Evolv Common Stock are hereafter referred to as the "Evolv Warrants." We have assumed for pro forma purposes that all of the warrants (excluding the unvested Finback Warrant) are exercised on a cashless basis using an implied fair value of Evolv Common Stock of \$3.78 per share and include a warrant exercisable for 2,686,246 shares of Series A-1 Preferred Stock and warrants exercisable for 2,665,678 shares of Evolv Common Stock. The warrant exercisable for Series A-1 Preferred Stock with 711 shares of Series A-1 Preferred Stock being forfeited to cover the exercise price using the assumed Evolv Common Stock are converted into 2,686,390 shares of Evolv Common Stock with 103,199 shares of Evolv Common Stock are converted into 333,836 shares of Evolv Common Stock fair value of \$3.78 per share. The vested Finback warrants exercisable for Evolv Common Stock are converted into 333,836 shares of Evolv Common Stock with 14,762 shares of Evolv Common Stock being forfeited to cover the exercise price using the assumed Evolv Common Stock fair v

Common Stock. At the Effective Time, each share of Evolv Common Stock (including shares outstanding as a result of the conversion of the Evolv Preferred Stock, the Evolv Convertible Notes and the Evolv Warrants converted into the right to receive such number of shares of NHIC common stock equal to the Exchange Ratio and a number of Earn-Out Shares (as defined below). The Exchange Ratio is defined in the Merger Agreement to be 125,000,000 divided by the number of outstanding shares of Evolv Common Stock and options to purchase shares of Evolv Common Stock as of immediately prior to the Effective Time, after giving effect to the conversion of the Evolv Preferred Stock, Evolv Convertible Notes and Evolv Warrants and as further adjusted pursuant to the Merger Agreement. The Exchange Ratio is 0.378 at the effective time of the Merger.

Stock Options. At the Effective Time, each outstanding option to purchase shares of Evolv Common Stock converted into an option to purchase shares of NHIC common stock equal to the number of shares subject to such option prior to the Effective Time multiplied by the Exchange Ratio, with the per share exercise price equal to the exercise price prior to the Effective Time divided by the Exchange Ratio.

Restricted Stock Units. At the Effective Time, each outstanding award of Evolv restricted stock units whether vested or unvested, converted into a restricted stock unit covering a number of shares of NewHold Common Stock equal to the number of shares subject to such restricted stock units prior to the Effective Time multiplied by the Exchange Ratio.

7

Earn-Out Shares Following the closing of the Merger, former holders of shares of Evolv Common Stock (including shares received as a result of the Evolv Preferred Stock conversion, Evolv Convertible Notes conversion and Evolv Warrants conversion), the former holder of the unvested portion of the Finback Warrant, former holders of Evolv RSUs and former holders of Evolv stock options and restricted stock units shall be entitled to receive their pro rata share of up to 15,000,000 additional shares of NHIC common stock (the "Earn-Out Shares") if, within a five-year period following the signing date of the Merger Agreement, the closing share price of the NHIC common stock equals or exceeds any of three thresholds over any 20 trading days within a 30-day trading period (each, a "Triggering Event") and, in respect of a former holder of Evolv stock options, the former holder of the unvested portion of the Finback Warrant and former holders of Evolv RSUs, the holder continues to provide services to NewHold or one of its subsidiaries at the time of such Triggering Event. The Earn-Out Shares to be issued upon the occurrence of the Triggering Events is deemed to be a contingent consideration arrangement and should be accounted for as a liability ("Earn-Out Liability"). The Earn-Out Liability should be accounted for as a liability and remeasured to fair value each reporting period. The Earn-Out Liability fair value of \$92.3 million was determined using the Monte Carlo Simulation for the Earn-Out shares to be issued to Evolv's selling equity holders upon the achievement of the Triggering Events, assuming no Earn-Out Forfeitures by Earn-Out Service Provider and estimated share price at closing of \$10.00 per share. The key inputs into this formula included (i) an assumed share price at closing (see 5(i) for a sensitivity analysis on a fluctuation of the estimated closing share price; (ii) the constant risk-free rate over the Earnout Period, which was based on the US Treasury Yield Curve and (iii) a constant equity volatility as

The PIPE Investment

On March 5, 2021, concurrently with the execution of the Merger Agreement, NewHold entered into subscription agreements (collectively, the "Subscription Agreements") with certain investors (collectively, the "PIPE Investors" which include certain existing equity holders of Evolv), pursuant to which the PIPE Investors have collectively subscribed for 30,000,000 shares of NHIC common stock for an aggregate purchase price equal to \$300.0 million (the "PIPE Investment") less \$19.9 million of equity issuance costs associated with the PIPE Investment accounted for as a reduction to additional paid-in capital. The PIPE Investment consummated immediately prior to the closing of the Business Combination.

2. Basis of Presentation

The unaudited pro forma condensed combined financial information were prepared in accordance with Article 11 of SEC Regulation S-X as amended by the final rule, Release No. 33-10786 "Amendments to Financial Disclosures about Acquired and Disposed Businesses." Release No. 33-10786 replaces the existing pro forma adjustment criteria with simplified requirements to depict the accounting for the transaction ("Transaction Accounting Adjustments") and present the reasonably estimable synergies and other transaction effects that have occurred or reasonably expected to occur ("Management's Adjustments"). Evolv has elected not to present Management's Adjustments and will only be presenting Transaction Accounting Adjustments in the unaudited pro forma condensed combined financial information. The adjustments presented in the unaudited pro forma condensed combined financial information have been identified and presented to provide relevant information necessary for an understanding of the combined company upon consummation of the Business Combination and the PIPE Investment.

The unaudited pro forma condensed combined balance sheet as of March 31, 2021 gives effect to the Business Combination, and the PIPE Investment as if they occurred on March 31, 2021. The unaudited pro forma condensed combined statement of operations for the three months ended March 31, 2021 and the year ended December 31, 2020 gives effect to the Business Combination and the PIPE Investment as if they occurred January 1, 2020, the beginning of the earliest period presented.

The pro forma adjustments reflecting the consummation of the Business Combination, and the PIPE Investment are based on certain currently available information and certain assumptions and methodologies that Evolv believes are reasonable under the circumstances. The pro forma adjustments, which are described in the accompanying notes, may be revised as additional information becomes available and is evaluated. Therefore, it is likely that the actual adjustments will differ from the pro forma adjustments, and it is possible that the difference may be material. Evolv believes that its assumptions and methodologies provide a reasonable basis for presenting all of the significant effects of the Business Combination, and the PIPE Investment based on information available to management at this time and that the pro forma adjustments give appropriate effect to those assumptions and are properly applied in the unaudited pro forma condensed combined financial information.

8

The unaudited pro forma condensed combined financial information does not give effect to any anticipated synergies, operating efficiencies, tax savings, or cost savings that may be associated with the Business Combination, and the PIPE Investment. NewHold and Evolv have not had any historical relationship prior to the Business Combination. Accordingly, no pro forma adjustments were required to eliminate activities between the companies.

On January 21, 2021, Evolv issued convertible promissory notes (the "2021 Notes") to certain existing investors for gross proceeds of \$30.0 million with a stated interest rate of 8.0% per annum. The 2021 Convertible Notes provided a conversion option whereby upon the closing of a Qualified Financing event, in which the aggregate gross proceeds totaled at least \$100.0 million, the notes would automatically convert into shares of the same class and series of capital stock issued to other investors in the financing at a conversion price equal to 80% of the price per share paid by the other investors. Separately, Evolv issued the Finback Warrant to a service provider for the purchase of 6,756,653 shares of Evolv Common Stock at an exercise price of \$0.16 per share. The warrants are not exercisable upon grant and become exercisable upon meeting certain performance-based milestone vesting conditions. The warrants will be accounted for as a performance based equity award to non-employees for remuneration of services using the Black-Scholes model to measure the value of the warrants upon grant date and will assess the probability of the warrants vesting at each reporting date. As of the Effective Date, there is a total of 348,598 warrant shares vested and \$1.0 million adjustment for the Finback Warrant. Accordingly, the pro forma adjustments are preliminary, subject to further revision as additional information becomes available and additional analyses are performed, and have been made solely for the purpose of providing unaudited pro forma condensed combined financial information. Upon consummation of the Business Combination, final valuations and studies will be performed. Differences between these preliminary estimates and the final accounting may occur and these differences could have a material impact on the accompanying unaudited pro forma condensed combined financial information and the combined company's future financial position and results of operations.

Management has made significant estimates and assumptions in its determination of the pro forma adjustments. As the unaudited pro forma condensed combined financial information has been prepared based on these preliminary estimates, the final amounts recorded may differ materially from the information presented.

The unaudited pro forma condensed combined financial information reflects 8,755,987 shares of outstanding NewHold Class A common stock that were redeemed, resulting in an aggregate payment of \$87.6 million out of the trust account, at a redemption price of \$10.00 per share.

This unaudited pro forma condensed combined financial information and related notes have been derived from and should be read in conjunction with:

- the unaudited historical condensed financial statements of NewHold as of and for the three months ended March 31, 2021 and the audited historical financial statements for the period from January 24, 2020 (inception) through December 31, 2020 (as restated), and the related notes thereto, included in the proxy statement/prospectus and incorporated by referee;
- the unaudited historical consolidated financial statements of Evolv as of and for the three months ended March 31, 2021 and the audited historical financial statements for the year ended December 31, 2020, and the related notes thereto, included in the proxy statement/prospectus and incorporated by reference; and
- the sections entitled "Management's Discussion and Analysis of Financial Condition and Results of Operations of NewHold," "Management's Discussion and Analysis of Financial Condition and Results of Operation of Evolv," and other financial information relating to NewHold and Evolv included in the proxy statement/prospectus and incorporated by reference.

The unaudited pro forma condensed combined financial information is for illustrative purposes only and is not necessarily indicative of what the actual results of operations and financial position would have been had the Business Combination, and the PIPE Investment taken place on the dates indicated, nor are they indicative of the future consolidated results of operations or financial position of the combined company.

9

3. Accounting for the Business Combination

The Business Combination will be accounted for as a reverse recapitalization, in accordance with GAAP. Under this method of accounting, although NewHold will issue shares for outstanding equity interests of Evolv in the Business Combination, NewHold will be treated as the "acquired" company for financial reporting purposes. Accordingly, the Business Combination will be treated as the equivalent of Evolv issuing stock for the net assets of NewHold, accompanied by a recapitalization. The net assets of NewHold will be stated at historical cost, with no goodwill or other intangible assets recorded. Operations prior to the Business Combination will be those of Evolv.

4. Capitalization

The following summarizes the pro forma ownership of common stock of Evolv following the Business Combination, and the PIPE Investment:

Equity Capitalization Summary	Shares	%
Evolv Equity Holders	100,351,091	70.1%
NHIC Public Stockholders	8,494,013	5.9%
NHIC Sponsor(1)	4,312,500	3.0%
PIPE Investors(2)	30,000,000	21.0%
Total Class A common stock	143,157,604	100.0%

- (1) The 4,312,500 NHIC Sponsor Shares ("Founder Shares") outstanding are subject to certain share-performance-based vesting provisions whereas 50% of the Founder Shares shall vest at the closing of the Merger, 25% of the Founder Shares shall vest on or before the fifth anniversary of the Closing if the closing share price of the common stock equals or exceeds \$12.50 over any 20 trading days within a 30-day trading period and the remaining 25% will vest on or before the fifth anniversary of the Closing if the closing share price of the common stock equals or exceeds \$15.00 over any 20 trading days within any 30-day trading period.
- 2) The PIPE Investment consummated in accordance with its terms for aggregate proceeds of \$300.0 million in connection with the issuance of 30,000,000 shares of NHIC common stock issued to the PIPE Investors.

5. Adjustments to Unaudited Pro Forma Condensed Combined Balance Sheet as of March 31, 2021

The pro forma notes and adjustments, based on preliminary estimates that could change materially as additional information is obtained, are as follows:

Pro forma notes

- (A) Derived from the unaudited consolidated balance sheet of Evolv as of March 31, 2021.
- (B) Derived from the unaudited condensed balance sheet of NewHold as of March 31, 2021.

Pro forma transaction adjustments

- (a) To reflect the settlement of \$6.0 million of deferred underwriting compensation incurred during NHIC's IPO that are contractually due upon completion of the Business Combination
- (b) To reflect the payment of NHIC's total estimated advisory, legal, accounting and other professional fees of \$12.3 million that are deemed to be direct and incremental costs of the Business Combination as a reduction to additional paid-in capital.
- (c) To reflect the reclassification of NHIC's warrant liability related to the public warrants as the NHIC public warrants are expected to qualify for equity classification upon the consummation of the Business Combination.
- (d) To reflect (i) to reflect the release of \$166.6 million of cash from the Trust Account, and (ii) to reflect the conversion of 4,312,500 shares of NewHold Class B common stock for NewHold Class A common stock, on a one-for-one basis. The 4,312,500 NewHold Class B common stock as converted to NewHold Class A common stock ("Founder Shares") are subject to certain share-performance-based vesting provisions whereas 50% of the Founder Shares shall vest at the closing of the Merger, 25% of the Founder Shares shall vest on or before the fifth anniversary of the Closing if the closing share price of the common stock equals or exceeds \$12.50 over any 20 trading days within a 30-day trading period and the remaining 25% will vest on or before the fifth anniversary of the Closing if the closing share price of the common stock equals or exceeds \$15.00 over any 20 trading days within any 30-day trading period.

10

- (e) To reflect the issuance of 216,441,907 shares of Evolv Common Stock upon the automatic conversion of all outstanding shares of Evolv Preferred Stock immediately prior to the Effective Time of the Merger. The adjustment reflects the derecognition of the carrying value of the Evolv Preferred Stock of \$85.4 million, including 2,685,535 shares of Evolv Series A-1 Preferred Stock issued upon the cashless exercise of a warrant immediately prior to the Effective Time of the Merger.
- (f) To reflect (i) the automatic conversion of Evolv's convertible notes (including the 2021 Notes) into 3,409,694 shares of Evolv Common Stock and the right to receive 1,000,000 shares of NHIC Common Stock at the Effective Time as further consideration for the conversion of the 2021 Notes consistent with terms thereof, (ii) the extinguishment of the related derivative liability and (iii) the resulting loss on extinguishment. Upon the conversion, the carrying value of the debt of \$28.9 million, including unamortized debt discount of \$6.3 million, and the related derivative liability of \$9.4 million were derecognized. The 3,409,694 shares of Evolv Common Stock issued upon conversion of the debt were recorded at the implied fair value of Evolv Common Stock of \$3.78 per share as negotiated between NHIC and Evolv, or \$44.1 million, and with the resulting difference being accounted for as a loss on extinguishment of \$5.8 million in earnings (see note 6(b) below).
- (g) To reflect the issuance of 2,686,390 shares of Evolv Common Stock upon the cashless exercise of common stock warrants outstanding immediately prior to the Effective Time of the Merger.
- (h) To reflect the elimination of the loss on Evolv's common stock warrant reality as the warrants are expected to convert to common stock and qualify for equity classification upon the consummation of the Business Combination.
- (i) To reflect the release of restriction on 3,888,889 shares of Evolv Common Stock held by an employee which secure a promissory note. The shares were released upon the settlement of the promissory note through the surrender 115,565 shares of Evolv Common Stock by the employee.
- (j) To record the Earn-Out Liability of \$92.9 million for the estimated fair value of the Earn-Out shares to be issued to Evolv's selling equity holders upon the achievement of the Triggering Events, assuming no Earn-Out Forfeitures by Earn-Out Service Providers and an assumed share price at closing of \$10.00 per share. An increase or decrease of \$1.00 in the underlying fair value of the common stock used in the valuation would increase or decrease the estimated fair value of the Earn-Out Liability by \$21.7 million.
- (k) To reflect the payment of Evolv's total estimated advisory, legal, accounting and other professional fees of \$2.9 million that are deemed to be direct and incremental costs of the Business Combination as a reduction to additional paid-in capital.
- (1) To reflect the recapitalization of Evolv through the contribution of all outstanding common stock of Evolv to NewHold and the issuance of 100,351,091 shares of NHIC common stock and the elimination of the accumulated deficit of NewHold, the accounting acquiree.
- (m) To reflect 8,755,987 shares of outstanding NewHold Class A common stock that were redeemed, resulting in an aggregate payment of \$87.6 million out of the trust account, at a redemption price of \$10.00 per share

Pro forma transaction directly attributable to the PIPE Investment

(n) To reflect the issuance of an aggregate of 30,000,000 shares of NHIC common stock at a price of \$10.00 per share, for an aggregate purchase price of \$300.0 million and to record the fee associated with the PIPE Transaction in the amount of \$19.9 million.

11

6. Adjustments to Unaudited Pro Forma Condensed Combined Statement of Operations for the Three Months Ended March 31, 2021

The pro forma basic and diluted earnings per share amounts presented in the unaudited pro forma condensed combined statements of operations are based upon the number of shares of NHIC common stock outstanding at the closing of the Business Combination and the PIPE Transaction, assuming the Business Combination.

The pro forma notes and adjustments, based on preliminary estimates that could change materially as additional information is obtained, are as follows: *Pro forma notes*

- (A) Derived from the unaudited consolidated statement of operations of Evolv for the three months ended March 31, 2021.
- (B) Derived from the unaudited statement of operations of NewHold for the three months ended March 31, 2021.

Pro forma adjustments

- (a) To reflect an adjustment to eliminate interest expense and amortization of discount on debt upon the automatic conversion of Evolv's convertible promissory notes as it is assumed that the convertible notes would have been converted to Evolv Common Stock and then to NHIC common stock as if the Business Combination had occurred on January 1, 2020.
- (b) To eliminate interest income earned on the Trust Account which will be released upon closing of the Business Combination.
- (c) To reflect the elimination of the loss on NHIC's derivative warrant liability related to the public warrants as the NHIC public warrants are expected to qualify for equity classification upon the consummation of the Business Combination.
- (d) To eliminate the loss on Evolv's common stock warrant as the warrants will convert to common stock upon the consummation of the Business Combination.
- (e) The pro forma basic and diluted net loss per share amounts presented in the unaudited pro forma condensed combined statement of operations are based upon the number of shares of NHIC common stock outstanding at the closing of the Business Combination and the PIPE Investment, assuming the Business Combination and PIPE Investment occurred on January 1, 2020. The unaudited pro forma condensed combined financial information has been prepared to reflect 8,755,987 shares of outstanding NewHold Class A common stock that were redeemed, resulting in an aggregate payment of \$87.6 million out of the trust account, at a redemption price of \$10.00 per share.

12

	Three M Ended M 202 (in thou except sh per shar	
Pro forma net loss	\$	(15,019)
Weighted average shares outstanding – basic and diluted	Ψ	143,157,604
Net loss per share – basic and diluted	\$	(0.10)
Weighted average shares calculation - basic and diluted		
NewHold weighted average public shares outstanding		4,312,500
NewHold common stock subject to redemption reclassified to equity		8,494,013
Issuance of NewHold common stock in connection with closing of the PIPE Transaction		30,000,000
Issuance of NewHold common stock to Evolv shareholders in connection with Business Combination		100,351,091
Weighted average shares outstanding		143,157,604

As the unaudited pro forma condensed combined statement of operations is in a loss position, the following anti-dilutive instruments were not included in the calculation of diluted weighted average number of common shares outstanding:

	Three Months
	Ended March
	31, 2021
Options to purchase NewHold Common Stock	22,231,178
Warrants to purchase NewHold Common Stock	16,793,554
	39,024,732

Following the Closing, the former holders of shares of Evolv Common Stock (including shares received as a result of the Evolv Preferred Stock conversion, Evolv Convertible Notes conversion and Evolv Warrants conversion), the former holder of the unvested portion of the Finback Warrant, former holders of Evolv RSUs and former holders of Evolv stock options shall be entitled to receive their pro rata share of Earn-Out Shares, issuable in three equal tranches upon the occurrence of each Earnout Triggering Event during the Earnout Period. Because the Earnout Shares are contingently issuable based upon the share price of NHIC common stock specified thresholds that have not been achieved, the Earnout Shares have been excluded from basic and diluted pro forma net loss per share.

7. Adjustments to Unaudited Pro Forma Condensed Combined Statement of Operations for the Year Ended December 31, 2020

NewHold and Evolv did not have any historical relationship prior to the Business Combination. Accordingly, no pro forma adjustments were required to eliminate activities between the companies.

The pro forma basic and diluted earnings per share amounts presented in the unaudited pro forma condensed combined statements of operations are based upon the number of shares of NHIC common stock outstanding at the closing of the Business Combination and the PIPE Transaction, assuming the Business Combination.

The pro forma notes and adjustments, based on preliminary estimates that could change materially as additional information is obtained, are as follows:

- (A) Derived from the audited consolidated statements of operations for the year ended December 31, 2020.
- (B) Derived from the audited consolidated statements of operations for the period from January 24, 2020 (inception) through December 31, 2020, as restated.

Pro forma adjustments

- (a) To reflect an adjustment to eliminate interest expense and amortization of discount on debt upon the automatic conversion of Evolv's convertible promissory notes as it is assumed that the convertible notes would have been converted to Evolv Common Stock and then to NHIC common stock as if the Business Combination had occurred on January 1, 2020.
- (b) To reflect an adjustment to record a loss of \$5.6 million on conversion of Evolv's convertible promissory notes as if the Business Combination had occurred on January 1, 2020. It should be noted that the loss on conversion of \$5.6 million was calculated based on the carrying amounts of the convertible notes and derivative liability as of December 31, 2020, which represents the best available information.
- (c) To eliminate interest income earned on the Trust Account which will be released upon closing of the Business Combination.
- (d) To reflect the elimination of the loss on NHIC's derivative warrant liability related to the public warrants as the NHIC public warrants are expected to qualify for equity classification upon the consummation of the Business Combination.
- (e) The proforma basic and diluted net loss per share amounts presented in the unaudited proforma condensed combined statement of operations are based upon the number of shares of NHIC common stock outstanding at the closing of the Business Combination and the PIPE Investment, assuming the Business Combination and PIPE Investment occurred on January 1, 2020. The unaudited proforma condensed combined financial information has been prepared to reflect 8,755,987 shares of outstanding NewHold Class A common stock that were redeemed, resulting in an aggregate payment of \$87.6 million out of the trust account, at a redemption price of \$10.00 per share.

Vear Ended

	December 31, 2020	
	(in thousands, except share and per share data)	
Pro forma net loss	\$	(38,642)
Weighted average shares outstanding – basic and diluted		143,157,604
Net loss per share – basic and diluted	\$	(0.27)
Weighted average shares calculation - basic and diluted		
NewHold weighted average public shares outstanding		4,312,500
NewHold common stock subject to redemption reclassified to equity		8,494,013
Issuance of NewHold common stock in connection with closing of the PIPE Transaction		30,000,000
Issuance of NewHold common stock to Evolv shareholders in connection with Business Combination		100,351,091
Weighted average shares outstanding		143,157,604

14

As the unaudited pro forma condensed combined statement of operations is in a loss position, the following anti-dilutive instruments were not included in the calculation of diluted weighted average number of common shares outstanding:

	Year Ended
	December 31,
	2020
Options to purchase NewHold Common Stock	16,964,452
Warrants to purchase NewHold Common Stock	16,821,525
	33,785,977